COLLECTIVE AGREEMENT

BETWEEN
VALE CANADA LIMITED
AND
UNITED STEELWORKERS, LOCAL 6500

JULY 8, 2010 - MAY 31, 2015

ONTARIO OPERATIONS

03/30/2019(09)

INDEX

AGREEMENT	1
ARTICLE 1- SCOPE	1
ARTICLE 2- RECOGNITION	
Contracting Out	3
ARTICLE 3- NO DISCRIMINATION	3
ARTICLE 4- MANAGEMENT	4
ARTICLE 5- UNION SECURITY	4
ARTICLE 6- NO CESSATION OF WORK	6
ARTICLE 7- ADJUSTMENT OF GRIEVANCES	6
Stage One	6
Stage Two	
Differences Between Company and UnionGeneral Grievance Committee, Grievance Stewards and Plant	
Committeemen	
General	8
ARTICLE 8- THIRD PARTY DETERMINATION	
Grievance Commissioner	
Arbitration	12
ARTICLE 9- DISCHARGE AND DISCIPLINE	13
Special Discharge Grievance and Arbitration Procedure	14
ARTICLE 10 - EMPLOYEES IN ARMED FORCES	15
ARTICLE 11 - SENIORITY	15
Application of Seniority on Layoff	17
Application of Seniority in a Department	
Application of Seniority - Curtailment in Operations	
Temporary Promotions, Demotions and Transfers	
Job Posting Transfers General	
Leaves of Absence	
General	
ARTICLE 12 -ESTABLISHMENT OF NEW DEPARTMENTS, TRANSFERS AND SPECIAL SENIORITY PROVISIONS	33

ARTICLE 13 - HOURS OF WORK - OVERTIME	35
Applicable Hourly Rate	36
Overtime	36
Daily Overtime	36
Weekly Overtime	36
Recognized Holidays	37
Sunday Premium	39
Shift Premium	39
Collar to Collar	39
4-Hour Minimum	39
Call-Out	
Transportation	
Lunches	40
Work Day - Work Week	41
ARTICLE 44 WAGES	44
ARTICLE 14 - WAGES	
Technological Change	
Jury Duty and Crown Witness	
Bereavement Leave	
Pregnancy Leave Supplemental Pay	44
ARTICLE 15 - VACATIONS	45
ARTICLE 16 - SPECIAL VACATIONS	47
ARTICLE 17 - HEALTH, SAFETY AND ENVIRONMENT	49
ARTICLE 18 - BULLETIN BOARDS	55
ARTICLE 19 - MISCELLANEOUS	56
ARTICLE CO. TRAREGAMB CTATIONARY ENGINEERS	50
ARTICLE 20 - TRADES AND STATIONARY ENGINEERS	
Trades	
Apprentices	
Stationary Engineers, Refrigeration Operators and Compressor Operators	
Incentive for Study Courses	60
ARTICLE 21 - GENERAL TRAINING	60
ARTICLE 22 - INCENTIVES	62
ARTICLE 23 - PENSIONS	64
ARTICLE 24 - INSURANCES, HEALTH AND HOSPITAL SERVICES	

ARTIC	CLE 25 - COST OF LIVING ALLOWANCE67
ARTIC	CLE 26 - SUPPLEMENTARY UNEMPLOYMENT BENEFITS PLAN (SUB)68
ARTIC	CLE 27 - CO-OPERATIVE WAGE STUDY68
ARTIC	CLE 28 - SPECIAL SHORT TERM SHUTDOWN PROTECTION PLAN71
ARTIC	CLE 29 - AUTHORITY71
ARTIC	CLE 30 - FURTHER CONFERENCES72
ARTIC	CLE 31 - TERMINATION72
SCHE	DULES
Α	Classifications of Employees Excluded from the Bargaining Unit73
В	Rules Governing Proceedings of Grievance Commissioner79
С	Rules of Arbitration80
D	Departments81
Е	Classifications Entitled to Preferential Recall82
F	Wage Rates83
G	Schedule of Apprenticeship Training / CWS Wage Class Payable84 for Apprenticeship Periods
Н	Plant Committeeman Areas85
J	Health, Safety & Environment Committee Areas/ Work Locations for
N	Employee Bonus Plan88
0	Special Short Term Shutdown Protection Plan91
Р	Plants for Section 11.09 (2)
Q	Twelve (12) Hour Shift Language Amendment93
U	The Contracting Out Information and Review System Components95

EMPLOYEE BENEFIT PLANS

Group Term Life Insurance Plan	98
Group Non-Occupational Accidental Death and Dismemberment Insurance Plan	99
Prescription Drug Plan (Formulary - 2)	. 100
Non-Occupational Sickness and Accident Insurance Plan	. 101
Dental Plan	. 102
Group Optical Insurance Plan	. 102
LETTERS OF AGREEMENT	
CONTRACTING OUT LETTERS OF AGREEMENT:	
Mining Skills & Contracting Out	. 104
Advance Notification of Small Contracts	. 105
All Mines Construction Group	. 106
Contracting Out Committee	.108
Contracting Out Steering Committee	. 109
Construction Group	. 111
Central Field Services Workgroup	. 113
Contractors and Mobile Equipment	. 115
Contracting Out Initiative Review	. 116
DISABILITY MANAGEMENT LETTERS OF AGREEMENT:	
Pension (YEWPs and Entitlement to Non-Occupational Sickness and Accident Insurance Payments)	117
Increases in Group Non-Occupational Sickness and Accident Insurance	

Payments	118
Joint Committee (Placement of Permanent Partially Disabled Employees)	119
Absent Employees Subject to Layoff	120
Disability Retirement Applications for Individuals with Recall Rights	121
Recovered Disability Pensioners Eligible to Apply for Full Time Employment	122
Vocational Rehabilitation Committee	123
Rehabilitation Group	124
No Fault Auto	125
DISPUTE RESOLUTION LETTERS OF AGREEMENT:	
Grievance Commissioners	126
Grievance Process	127
Education Regarding the Grievance Process	128
Mediated Arbitration	129
SAFETY SYSTEM LETTERS OF AGREEMENT:	
Joint Occupational Health Study Program	130
Training in Use of Monitoring Equipment	133
Prescription Safety Glasses	135
Environmental Awareness Committee	136
Changes to <i>OHSE</i>	138
TRADES LETTERS OF AGREEMENT:	
Mileage Compensation - Construction Group Employees	140
Trades Committee	141
Force Adjusted <i>Tradespersons</i>	143

Apprentice Posting Travel Compensation	144
Trades Apprenticeship Opportunities	145
WORKFORCE TRANSITION LETTERS OF AGREEMENT:	
Timing of Job Posting Procedure	146
Trial Process for Awarding Unlimited Jobs	148
Reciprocal Recall Rights & Rehiring	149
Special Short Term Shutdown Protection Plan	150
Travel Compensation	. 151
OTHER LETTERS OF AGREEMENT:	
'Will' and 'Shall'	152
Credit Union Deductions	153
Plant Committeeman Facilities	154
Plant Committeemen - Upper and Lower Yard	155
Union Elections or Votes	156
Vacations	157
Lists of Overtime Worked	158
Employee Involvement in Training	159
Distribution of Overtime	160
Orientation of New Employees	161
Coleman and Lower Coleman	162
Education, Humanities, SOAR & Community Funds	163
Warehouse Travel Crew	164
Vacation Earnings for Employees in Receipt of Benefits under the Group Non-	

Occupational Sickness and Accident Insurance Plan and <i>Pregnancy or Parental</i> Leaves, in the Preceding Year	
Whistle Mine Development Work	.166
Team Leader	.167
Health Care Benefits Protection Program	. 169
Summer Student Category	. 171
Return to Production Bonus	. 172

THIS AGREEMENT made as the eighth of July 2010

BETWEEN

VALE CANADA LIMITED, hereinafter called the 'Company',

OF THE FIRST PART

- and -

UNITED STEEL, PAPER AND FORESTRY, RUBBER, MANUFACTURING, ENERGY, ALLIED INDUSTRIAL AND SERVICE WORKERS INTERNATIONAL UNION (UNITED STEELWORKERS) hereinafter called the 'Union',

OF THE SECOND PART

WHEREAS the Ontario Labour Relations Board has, by its certificate dated the fifteenth day of October, 1962, certified the Union as Bargaining Agent for all employees of the Company in the Sudbury District save and except office staff, officers and officials of the Company and individuals employed by the Company in the classifications set forth in Schedule 'A' attached hereto and made a part hereof;

AND WHEREAS it is the intent and purpose of the Union and the Company to further harmonious industrial relationships between the Company and its employees;

NOW THEREFORE THIS AGREEMENT WITNESSETH that the parties hereto hereby agree as follows:

ARTICLE 1 - SCOPE

- 1.01 This Agreement is entered into by the Union on its own behalf and on behalf of all employees.
- 1.02 The word 'employee' or 'employees' wherever used in this Agreement shall mean the employees of the Company in the Sudbury District defined as the appropriate collective bargaining unit by the above mentioned certificate of the Ontario Labour Relations Board dated the fifteenth day of

October, 1962.

- 1.03 If the Company establishes a new nickel-copper mining, milling, smelting or refining operation in the Sudbury District, the provisions of this Agreement will apply to bargaining unit employees working at such operations.
- 1.04 The words 'Local Union' wherever used in this Agreement shall mean Local 6500 of the *United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)*.
- 1.05 Wherever the masculine gender appears in this Agreement, it shall also mean the feminine gender, unless the context requires otherwise.

ARTICLE 2 - RECOGNITION

- 2.01 The Company recognizes the Union as the sole collective bargaining agency for all employees with respect to rates of pay, hours of work, and other conditions of employment.
- 2.02 Foremen, supervisors and other persons employed by the Company, but not employees as defined in this Agreement, shall not perform work which is normally done by employees (except cases of an emergency nature) where qualified employees are reasonably available to do such work. Nothing in this paragraph shall interfere with the normal performance of duties such as the carrying out of research programs, the necessary instruction of employees, the inspection of equipment, product or process by such non-bargaining unit employees. If a grievance is filed alleging violation of the foregoing, the foreman, supervisor, or other person referred to in the grievance and the person on whose evidence the grievance is substantially based shall be required, on the request of either party, to attend any grievance meetings established under Article 7 of this Agreement.
- 2.03 It is recognized that changing technology or changes in methods of operation may result in the creation of new production and maintenance jobs. Such new jobs shall as soon as practicable be performed by bargaining unit employees if such jobs comprise the significant functions which have been part of jobs performed by bargaining unit employees. If such new jobs do not comprise the significant functions, but do comprise similar functions which have been part of jobs performed by bargaining unit employees, such new jobs will on the request of the Local Union be reviewed by the Company with the Local Union as to whether such new jobs are to be performed by bargaining unit employees.

The Company will meet on request with the Local Union to discuss those new occupational classifications for which training will be given under Articles 20 or 21. If, in the opinion of the Union, alleged breaches of this Section cannot be satisfactorily dealt with under the grievance procedure, the Company undertakes to discuss with the Union or the Local Union under the provisions of Section 7.21 hereof what appropriate action should be taken to correct specific or repeated instances of such alleged violations.

2.04 Contracting Out

- a) The Company agrees that the Union has an understandable concern over 'contracting out' by the Company because of its effect upon such matters as job opportunity for the employees.
- b) The Company will, therefore, having due regard to the availability of equipment, engineering, skills, manpower, supervision and services and to operating efficiency, and to the time to do the work, make efforts to limit the amount of the future production or maintenance work to be 'contracted out' during this Agreement.
- c) No employee will be demoted or laid off as a direct result of work being contracted out by the Company.
- d) Persons employed by contractors shall not, except in cases of emergency, use or operate Company owned equipment or machinery. This shall not apply to such equipment or machinery which is installed and/or in a fixed location.
- e) Contractors will not perform work outside the scope of their contract with the Company.
- f) The provisions of Schedule 'U' relating to notice, information, trend analyses, base load, operated mobile equipment and annual review form part of this Agreement.

ARTICLE 3 - NO DISCRIMINATION

- 3.01 No person shall be required, as a condition of employment, to become or remain a member of any Union or other organization, and no statements or representations to the contrary shall be made.
- 3.02 There shall be no discrimination or favouritism by the Company or the Union or its members against any employee because of membership or

non-membership in any lawful union, or because of the employee's sex, race, colour, religious creed, political affiliation, or because the employee has exercised or failed to exercise any right specifically provided under this Agreement.

- 3.03 There shall be no solicitation of membership in any Union organization or collection of Union dues or any Union activity on Company time or Plant or Mine property. This clause shall not prevent employees from engaging in casual conversation relating to Union affairs.
- 3.04 The parties endorse a workplace environment which is free from harassment that denies individual respect and dignity on grounds of race, national or ethnic origin, colour, religion, age, sex, marital status or sexual orientation. For the purpose of this clause, harassment means:
 - (a) unwanted attention, remarks or behaviour of a persistent or abusive nature made by a person who knows or ought reasonably to know is unwarranted and can include comments, slurs, racist or sexists jokes, pictures or posters, bullying or intimidation, graffiti, unnecessary physical contact, remarks about a person's appearance or personal life, unwelcome sexual advances or demands;
 - (b) implied or expressed promise of reward for complying with a sexually oriented request;
 - (c) implied or expressed threat or reprisal, in the form of either actual reprisal or the denial of opportunity for refusal to comply with a sexually oriented request;
 - (d) harassment does not include properly exercised employee or supervisory responsibilities or conduct.
- 3.05 The Company recognizes the Local Union's Human Rights officer in providing assistance in the investigation and resolution of human rights complaints arising in the workplace. Leaves of absence shall be granted to this Local Union representative to the extent required. Time lost by the Local Union representative in this function shall be paid for by the Company at the equivalent of the wage rate for Maintenance Electrician, plus any applicable shift premium and Cost of Living Allowance and any applicable **Schedule** 'N' Bonus. The Company may at any time discontinue such payments if, in the opinion of the Company, there is abuse of the privilege.

ARTICLE 4 - MANAGEMENT

4.01 The Union agrees that the Company has the exclusive right and power to manage the Plants and Mines, to direct the working forces and to hire, promote, transfer, demote or lay off employees, and to suspend, discharge or otherwise discipline employees for just cause. Provided, however, that the Company agrees that any exercise of these rights and powers in conflict with any of the provisions of this Agreement shall be subject to the provisions of the grievance procedure.

ARTICLE 5 - UNION SECURITY

- 5.01 The Company shall during the life of this Agreement deduct as a condition of each employee's continued employment, from the pay cheque due in each *pay period* to each such employee, a sum equal to Union weekly dues in the amount certified by the Union to the Company to be currently in effect according to the Union Constitution and remit the same prior to the 10th day of the month following the month in which such deductions are made to the International Treasurer of the Union. The Company will, at the time of making each such payment to the International Treasurer of the Union, provide the names and current addresses according to the Company records of the employees from whose pay such payment has been deducted.
- 5.02 The said deductions shall commence, in the case of each employee entering the employment of the Company, with the calendar month in which his first pay cheque from the Company is received by him.
- 5.03 The remittance and the R-115 form shall be accompanied by a statement containing the following information:
 - a) A list of the names of all employees from whom dues were deducted and the amount of dues deducted;
 - b) A list of the names of all employees from whom no deductions have been made.
- 5.04 The Union shall indemnify and save the Company harmless against all claims or other forms of liability that may arise out of any actions taken by the Company in compliance with this Article.
- 5.05 The Company, when preparing T4 slips for the employees, will enter the amount of Union dues paid by the employee during the taxation year for which the T4 slip is issued.

ARTICLE 6 - NO CESSATION OF WORK

6.01 Neither the Union nor any employee shall take part in or call or encourage any strike, sit-down, slow-down, or any suspension of work against the Company which shall in any way affect the operations of the Company; nor shall the Company engage in any lockout at its Plants or Mines in the Sudbury District.

ARTICLE 7 - ADJUSTMENT OF GRIEVANCES

Differences Between the Company and Employees

7.01 Should any difference arise between the Company and any of the employees from the interpretation, application, administration or alleged violation of the provisions of this Agreement, an earnest effort shall be made to settle such differences without undue delay in the following manner.

7.02 Stage One

An aggrieved employee may, provided it is done with reasonable promptness, make his representations orally or in writing to the Management Representative designated by the Company from time to time to handle such matters at this stage. In making such representations he may, if he desires, be accompanied and represented by a Grievance Steward in his Department or a Grievance Steward from his Plant, if there is no Grievance Steward available in his Department, and such meeting will normally occur during the grievor's working hours. Any settlement of the grievance reached between the aggrieved employee and the Company at this stage will be considered applicable to the case in question and not as establishing a precedent for future or other cases. If within ten (10) days from the time when such representations were presented to such Management Representative a decision satisfactory to the employee is not given, then the grievance may be referred to Stage Two.

7.03 Stage Two

Failing resolution, the grievance may, within *ten (10)* days after the decision of the Management Representative for Stage One has been or should have been given, be referred in writing to the Management Representative designated by the Company to handle such meetings at Stage Two.

The Company will notify the Plant Committeeman and the Chair of the

General Grievance Committee of the date, time and place for the Stage Two meeting within five (5) days after receiving such referral. **The Stage Two meeting will be held in the Plant where the grievance arose.** Such meeting shall be held within **twenty-eight (28)** days following such notification and will normally be scheduled during the grievor's working hours **between 6 a.m. and 6 p.m., Monday to Friday, unless otherwise agreed.**

The employee shall be accompanied at such meeting by the Plant Committeeman or Grievance Steward and up to three (3) Chief Stewards. The Management Representative hearing the grievance may be accompanied by such other personnel as the Company considers necessary.

The Management Representative hearing the grievance will, within seven (7) days from the date of the meeting, give a written decision to the grievor, with a copy for the Plant Committeeman or Grievance Steward and the Chair of the General Grievance Committee. If the grievance is denied, the decision shall contain brief and concise reasons.

- 7.**04** Regular days off of both the grievor and the Management Representative involved at each stage, vacation time off of the grievor and recognized holidays shall not be counted in determining the time within which any action is to be taken in each of the foregoing Stages.
- 7.05 The Company will indicate by notice posted for each Department the Management Representatives designated by it from time to time to handle matters in such Department at Stages One and Two of the Grievance Procedure and a copy of each notice posted in the Departments in the Plant shall be given to the Plant Committeeman.
- 7.06 If no Management Representative shall have been designated by the Company for any of the stages through which a matter must proceed such matter may proceed directly to the next higher stage for which a Management Representative shall have been so designated.

7.07 <u>Differences Between Company and Union</u>

Any differences arising between the Union and the Company from the interpretation, application, administration or alleged violation of the provisions of this Agreement, instead of following the procedure hereinbefore set out, may be submitted in writing by either of such parties to the other, with opportunity for oral discussion between the officers of the Local Union and a senior official of the Company or the designated representative. At such discussion either party may require the attendance of a representative of the Union and/or not more than the three

- (3) Chief Stewards. The parties shall within ten (10) days from the giving of notice as aforesaid sign a memorandum of settlement, or, if the matter has not been satisfactorily adjusted, the party to whom the matter was submitted shall deliver to the other party a reply in writing to such submission.
- 7.08 Time limits fixed by this Article may be extended by written agreement of the Company and the Union or the Local Union.

General Grievance Committee, Grievance Stewards and Plant Committeemen

7.09 The General Grievance Committee and Grievance Stewards, as designated by the Local Union, shall be employees and shall consist of not more than one (1) for each thirty-five (35) employees.

There shall be one Chief Grievance Steward from each of the following sections:

Mines Section
Operating
Maintenance and Electrical.

- 7.10 From each Plant listed in Schedule 'H' the Local Union shall designate to the Company from time to time those Grievance Stewards in numbers as designated in Schedule 'H' who shall be Plant Committeemen. The Plant Committeeman shall be the responsible Local Union representative for the preparation and processing of grievances in respect of Stage Two of the Grievance Procedure in their respective Plants.
- 7.11 The General Grievance Committee shall consist of three (3) Chief Stewards who shall be continuing members together with two (2) Grievance Stewards who may be designated by the Local Union to act for the purpose and chosen from time to time from among the other Grievance Stewards.
- 7.12 The Local Union will designate to the Company from time to time the Grievance Stewards acting as Chief Grievance Stewards and as *Chair* of the General Grievance Committee.
- 7.13 Any Grievance Steward, Plant Committeeman and officer of the Local Union shall be allowed such time off as shall be reasonably required to attend such necessary meetings with supervisory personnel as provided in this Article, subject, however, to obtaining permission (which shall not be unreasonably withheld) from the Management Representative designated by the Company. When it is necessary for Grievance Stewards, Plant Committeemen or officers of the Local Union to leave their regular place of

employment to attend such necessary meetings with supervisory personnel at other than their regular place of employment, and they are required to return to their regular employment after the completion of such meetings, the Local Union agrees to pay the transportation to such meetings and the Company agrees to pay the return transportation.

- 7.14 Each Plant Committeeman shall be allowed such time off as shall be reasonably required up to two and one-half (2 1/2) hours per week at a time to be arranged with his Plant Manager, or his designate in order to consult with employees or Grievance Stewards in connection with Stage Two *grievance matters*. In the case of surface operations the reasonable travelling time involved in any visits to work sites, which the Plant Committeeman and the *Plant Manager* or his designate agree are necessary, shall not be included in the Committeeman's two and one-half In the case of underground operations, (2 1/2) hour allowance. arrangements will be made at each Mine to permit a Plant Committeeman to come up early or delay his descent time on a prearranged schedule so that he may talk to employees or Grievance Stewards arriving at or leaving from work in connection with Stage Two matters. circumstances warrant, as agreed by the Plant Manager, or his designate, and Plant Committeeman, the Committeeman in connection with Stage Two preparation will be permitted to visit a particular location in a Mine but the necessary time involved in travelling to and from the particular location in the Mine in connection with such visits shall not be included in the Committeeman's two and one-half (2 1/2) hour allowance.
- 7.15 Each Plant Committeeman will be provided the facilities as determined under the Letter of Intent dated the 8th day of July, 2010.
- 7.16 The Company shall pay the Grievance Stewards, Plant Committeeman and officers of the Local Union (other than full-time officers) at their applicable hourly rate, plus any applicable shift premium, any applicable Cost of Living Allowance, and any applicable Schedule 'N' Bonus, for any time lost during their regular shifts in connection with any of the above mentioned duties, but the Company may discontinue such payments if the privilege of requesting time off is being abused.
- 7.17 The Company will grant the three (3) Chief Grievance Stewards up to a maximum of five (5) days leave of absence with pay in each week at the equivalent of the wage rate for Maintenance Electrician plus any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus**.

<u>General</u>

7.18 All grievances being processed under the provisions of the 2006-2009

Agreements will be continued under the provisions of Article 7 of this Agreement. Any arbitration board established to deal with such grievances under this Agreement will have the right to determine the employees' rights under the previous Agreement.

- 7.19 Should any difference arise between the Company and the Union or any employee from the interpretation, application, administration or alleged violation of the provisions of this Agreement, or should any employee believe that his discharge is in violation of the provisions of this Agreement and should any party desire to take advantage of the procedure provided for in this Article, each step in such procedure (up to and including the appointment of a Board of Arbitration) required to effect a satisfactory disposition of the matter shall be taken by such party within the time limits above set forth or the grievance shall be deemed to have been abandoned.
- 7.20 In the event that it should be decided through the Grievance Procedure that any employee has failed to receive the rate of pay to which the employee is properly entitled under the provisions of this Agreement respecting the application of seniority or rates of pay, the Company shall, if the employee's grievance shall have been presented in writing within forty five (45) days after the pay day in respect of the pay period in which the grievance arose, pay to such employee the difference between the rate of pay which such employee should have received and the pay actually received by such employee.
- 7.21 Upon the request of the Union or the Local Union President, the Company will meet with five (5) employees of one Department, a representative of the Union and the Local Union President, or his designate, to discuss problems of a general nature arising out of this Agreement and relating to that Department. If a meeting is requested, the Local Union President must notify the Manager Human Resources prior to the meeting of the subjects to be discussed thereat. In the case of up to two (2) such meetings for each Department and five (5) additional meetings in each year of the Agreement, the Company will pay such employees attending the meetings at their applicable hourly rate plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable Schedule 'N' Bonus for any time lost during their regular shifts. At any such meeting, the Local Union President may require the attendance of up to two (2) additional employees, not paid for by the Company, having special technical knowledge relating to the subjects to be discussed thereat.
- 7.22 Where two (2) or more employees in the same Department have differences with the Company concerning the application, interpretation or an alleged violation of the provisions of this Agreement which are

sufficiently common in nature to be dealt with together, they shall constitute a group grievance and it shall be heard at Stage Two. For the purposes of Stage Two, one employee shall be appointed by the Union as representative of all aggrieved employees.

ARTICLE 8 - THIRD PARTY DETERMINATION

8.01 In the event that any difference arising from the interpretation, application, administration or alleged violation of the provisions of this Agreement shall not have been satisfactorily settled under the foregoing provisions, the matter may then by notice in writing given to the other party within thirty (30) days from the giving of the decision of the Management Representative at Stage *Two* (or in the event of a difference between the Union and the Company, within thirty (30) days from the date when the written reply to the submission was or should have been delivered) be referred either by the Union or the Company to Third Party Determination as hereinafter provided.

Within fifteen (15) days after a grievance has been referred to Third Party Determination, the designated representatives of the *Human Resources* Department and the three (3) Chief Stewards, an officer of the Local Union and a representative of the Union may meet and agree in writing to submit the grievance to a Grievance Commissioner as provided in paragraph 8.02 hereof. Failing such agreement, the grievance will proceed to an arbitration board as provided hereunder.

Grievance Commissioner

8.02 The Company and Union may agree in writing to the appointment of a person or persons as a single arbitrator to be known as a Grievance Commissioner hereunder. All cases referred to arbitration where an employee has a grievance concerning either discipline received by him or overtime not paid to him will be referred to a Grievance Commissioner if the Company and the Local Union agree in writing on all the facts to be placed before the Grievance Commissioner. With respect to overtime cases, either party may elect to have the matter heard at arbitration if it considers the issues involved to be of precedential importance. Grievance Commissioner (where more than one, acting in rotation) will set aside each month such time as may be requested by the Company and the Union to consider and determine grievances referred to him hereunder for final and binding arbitration. A Grievance Commissioner shall have the same powers and be subject to the same limitations as a Board of Arbitration hereunder, save and except as expressly provided in paragraphs 8.02 to 8.05 hereof.

- 8.03 Through the Grievance Commissioner the parties desire the expeditious means for the effective disposition of grievances which the parties have agreed may be handled in a summary manner. The rules governing the summary proceedings of the Grievance Commissioner are set out in Schedule 'B' hereto.
- 8.04 The decision of the Grievance Commissioner shall only be applicable to the case in question and shall not constitute a precedent nor be used by either party as a precedent in future cases. Notwithstanding anything contained in this Agreement, the decision of the Grievance Commissioner shall:
 - a) be consistent with the provisions of this Agreement.
 - b) be confined to the grievance referred to him.
- 8.05 The Local Union and the Company shall each be responsible for one-half (1/2) of the expenses of and fees payable to the Grievance Commissioner.

Arbitration

8.06 Any matter referred to arbitration shall be heard by an arbitrator selected in rotation from the following panel:

Janice G. Johnston Richard O. MacDowell W. B. Rayner C. Gordon Simmons Kevin Burkett

The Company and the Local Union may upon agreement add further names to the panel.

- 8.07 The party referring the matter to arbitration shall within fifteen (15) days take all steps necessary for the selection of an arbitrator from the panel in the manner herein set out.
- 8.08 If after making all reasonable efforts to select an arbitrator within the time herein set out, the party responsible for the selection is unable to find an arbitrator able or willing to act, the time limit will be extended to the length of time required to obtain the services of an arbitrator.
- 8.09 Any member of the panel who, having been requested in turn to act as an arbitrator, is unable or unwilling to act shall not again be requested to act as arbitrator on any arbitration until his/her name comes up again on the regular rotation of the panel. If a member of the panel has been unable or

- unwilling to act as arbitrator on six (6) consecutive occasions when requested during the term of this Agreement, he/she shall be removed from the panel and a replacement shall be selected by agreement between the Company and the Union within sixty (60) days of the removal.
- 8.10 No matter shall be considered by the arbitrator nor shall the arbitrator render any decision in connection with the matter unless and until he/she first decides that the matter constitutes a difference between the Company and the Union, or the employee initiating the matter, arising from the interpretation, application, administration or alleged violation of this Agreement.
- 8.11 The Rules of Arbitration annexed hereto as Schedule 'C' shall govern the conduct of any arbitration proceedings. A finding by the arbitrator as to the facts and as to the interpretation, application, administration or alleged violation of this Agreement shall be conclusive and binding upon all parties concerned but in no event shall the arbitrator alter, modify or amend any part of this Agreement.
- 8.12 The Local Union and the Company shall each be responsible for one-half (½) of the expenses and fees of the arbitrator or mediator.
- 8.13 Any and all times fixed by this Article for the taking of action by either party may at any time be extended by written agreement of the Company and the Union or Local Union.
- 8.14 When a grievance has neither been settled under the grievance procedure nor referred to a Grievance Commissioner and if the Company and the Local Union agree, or the grievance concerns the discipline or discharge of an employee, the grievance shall be referred to a single arbitrator selected in rotation from the panel set out in Section 8.06 of the Collective Bargaining Agreement for final and binding arbitration. The provisions of the Collective Bargaining Agreement, including Article 8 and Schedule 'C', shall apply in cases where a single arbitrator is used, with the necessary changes being made.

ARTICLE 9 - DISCHARGE AND DISCIPLINE

9.01 A discharged employee may contest his discharge as being in violation of this Agreement provided the written grievance is filed within fourteen (14) days of the date of discharge subject to extension to twenty-one (21) days, if it is impossible for an employee to file within the fourteen (14) days, or such further period as the parties may agree. Such grievance will be treated as a special grievance and presented at a Stage *Two* meeting convened for the purpose of dealing with the grievance.

9.02 If a discharged employee is reinstated pursuant to the Grievance or Arbitration Procedure, any compensation for earnings lost shall be on the basis of the employee's regular basic rate and normal hours of work (not to exceed forty (40) hours per week) plus any off shift premium or Sunday premium applicable to such normal hours of work and any applicable Cost of Living Allowance, any applicable *Schedule 'N' Bonus* and any holiday pay he would have otherwise been entitled to less amounts earned by the employee during the time so lost. Upon reinstatement there shall be deemed to have been no break in the employee's continuous service.

Special Discharge Grievance and Arbitration Procedure

- 9.03 The Stage **Two** meeting will be held within three (3) days after the grievance is received and the Management Representative shall give his decision in writing to the Local Union within three (3) days from the date of the meeting. If the decision is unsatisfactory, the Local Union may, within seven (7) days after the decision is given, refer the grievance to Arbitration and it shall be heard pursuant to the provisions of Article 8 within thirty (30) days of the reference to arbitration given by the Local Union.
- 9.04 Any Counselling or Discipline Reminder, penalty warning or penalty suspension will contain reasons for its issuance and will be given in writing to the employee with a copy to the Grievance Steward designated by the employee. If more than one representative from Management is present during the issuance of any Step 1 or Step 2 Counselling or Discipline Reminder, it shall be given to such employee, if he so requests, in the presence of a Grievance Steward. Any Counselling or Discipline Reminder given as a Step 3 or Step 4 Reminder will, where practicable, be given to an employee, if he so requests, in the presence of a Grievance Steward. Where practicable, any notice of discharge shall be given to an employee in the presence of his Plant Committeeman or Grievance Steward if the Plant Committeeman is not available. Such notice of discharge shall specify the reason for discharge.
- 9.05 **Penalty warnings (but not a suspension) or any** Counselling or Discipline Reminder given as a Step 1 Reminder shall be removed from an employee's record and shall be disregarded in determining discipline if in the six (6) months following such warning or Counselling or Discipline Reminder there is no further misconduct. In any event a penalty warning (but not a suspension) or a Counselling or Discipline Reminder given as a Step 1 Reminder shall be removed from an employee's record and shall be disregarded in determining discipline twelve (12) months after the issuance of such warning or Counselling or Discipline Reminder.

A suspension notice or a Counselling or Discipline Reminder given as a

Step 2 Reminder shall be removed and disregarded in determining discipline if in the twelve (12) months following such Counselling or Discipline there is no further misconduct. In any event a Step 2 Reminder shall be removed and disregarded in determining discipline two (2) years after the issuance of such Counselling or Discipline Reminder. A suspension notice or a Counselling or Discipline Reminder given as a Step 3 or Step 4 Reminder shall be removed and disregarded in determining discipline two (2) years after the issuance of such suspension or Counselling or Discipline Reminder.

An employee who has no other Counselling or Discipline Reminder on his record at the time of receiving a Step 4 Reminder shall have such Step 4 Reminder reduced to a Step 3 Reminder one (1) year after its issuance provided he has not received any other Counselling or Discipline Reminder during that year. Thereafter, such Step 3 Reminder shall be removed and disregarded in determining discipline if in the twelve (12) months following there is no further misconduct.

ARTICLE 10 - EMPLOYEES IN ARMED FORCES

10.01 An application for leave of absence from an employee required to be absent from work for the treatment of injuries sustained in Canadian military service will be carefully considered by the Company and discussed with the Local Union on request.

ARTICLE 11 - SENIORITY

11.01 Definitions

'Seniority' - is the principle of granting preference to employees in promotions, transfers, demotions due to a change or reduction in operations, layoffs, and rehiring after layoffs in accordance with length of employment.

'Seniority Ranking' - means the ranking of employees within a Department or the District as determined by their length of continuous service with the Company. Any employee transferred from one Department to another within the District shall carry his seniority from the old Department into the new Department.

'Department' - means those Departments as set out in Schedule 'D' attached hereto.

'Limited Job' - means

- a) A job not exceeding six (6) months' duration, which
 - i) is seasonal only, or
 - ii) is open due to a leave of absence other than one granted to a Union Officer for Union purposes; in the case of such leave of absence for Union purposes, the job will be posted limited for the first six (6) months of the leave and, if the leave continues, it will then be posted as an unlimited job;
- b) Any job which is open by reason of sickness or accident;
- c) Any job which the Company believes will not exceed three (3) months' duration;
- d) Any trainee job which the Company believes will not exceed twelve (12) months' duration;
 NOTE: This section is suspended, in accordance with Letter of Agreement, "Trial Process for Awarding Unlimited Jobs", page 148 CBA.
- e) Any job which is open by reason of the assignment of an employee to a job outside the bargaining unit. Any such job which remains 'limited' for a period of (1) year will be posted 'unlimited' at the expiration of the one (1) year period or sooner if the Company advises the Union that the assignment is permanent.

'Continuous Service' - means the length of unbroken service since the employee last entered the service of the Company according to the Company's records, less the following:

- a) Any leave of absence in excess of three (3) weeks for reasons other than disabling accident or illness, compassionate grounds or pregnancy **or parental leave**;
- b) A layoff for lack of work for a continuous period of three (3) full payroll periods or more;
- c) Any period of unlawful strike.

For the purpose of the seniority provisions of this Agreement all leaves of absence under Section 11.26, and any leaves of absence after July 10, 1969, in excess of three (3) weeks under Sections 11.27 and 11.28 shall

not be deducted from an employee's continuous service.

For the purpose of the seniority provisions of this Agreement an employee with recall rights who was laid off and rehired after his recall rights expired will be credited with his prior continuous service.

'Break in Service' - shall occur if an individual employed by the Company:

- a) quits (absence without leave for a period of over fourteen (14) days shall constitute a quit);
- b) is discharged;
- c) is laid off by the Company for lack of work, except that no break shall be deemed to have occurred if such a laid-off person is rehired in accordance with the recall provisions of this Article.

Application of Seniority on Layoff

- 11.02 a) Where employees are laid off from the Company for lack of work (other than temporary layoffs not to exceed two (2) weeks resulting from accident, breakdown or other emergency) the Company shall lay off the employees affected in the inverse order of their seniority ranking with the Company except in those cases where the requirements and efficiency of operations and ability, knowledge, training and skill of the individual to fill the normal requirements of the job require otherwise.
 - b) In the case where an employee will be permanently laid off out of seniority ranking under Section 11.02 (a), he will be provided with an appropriate one-time training period of up to three (3) consecutive months duration for the purpose of training in an occupational classification occupied by a junior employee in his Department or, if such an occupational classification is not available in his Department, in the Sudbury District.
- 11.03 In the event of a reduction in force, other than a temporary layoff, the Company shall, before notifying the employees to be laid off, have prior discussion with the Local Union executive and the Union.
- 11.04 An individual whose employment has been terminated due to a layoff for lack of work shall have preferential rights for rehiring (herein referred to as 'recall' rights) for the following periods subject to other provisions in this Article:

Length of Continuous
Service at Date of Layoff

Duration of Recall Rights

Three months or more Eighteen months from date of layoff

One year or more Twenty-four months from date of layoff

Three years or more Thirty-six months from date of layoff

Four years or more Forty-eight months from date of layoff

- 11.05 The Company shall maintain a recall list for the District. When hiring for jobs covered by this Agreement (other than for jobs of a temporary or emergency nature), the Company shall determine which individuals on the recall list are qualified to fill the normal requirements of the jobs available and shall give preference in rehiring to such persons in inverse order to that in which they were laid off subject to the following provisions:
 - a) Notice of recall shall be sent to the individual by registered mail and shall be sufficient if sent to the last address the individual has communicated in writing to the *Human Resources* Office of the Company at Copper Cliff. Upon layoff the individual will receive from the Company written instructions as to his rights and obligations for recall under this Section 11.05.
 - b) The notice shall stipulate the job to which the individual is being recalled and the proposed time and place of rehiring. The individual shall indicate his acceptance within a period not to exceed fourteen (14) days from the date of mailing of such notification and shall report for duty within twenty-eight (28) days from such date. If the individual reports for work within such twenty-eight (28) day period, his continuous service with the Company will be conclusively deemed to have recommenced from the date he was requested to report for duty.
 - c) Individuals who accept the recall and report for duty within the specified time set out in paragraph (b) hereof and at the specified place will be rehired in inverse order to that in which they were laid off.
 - d) No individual shall be denied rehiring solely because of a medical or physical condition which existed at the time he was laid off.
 - e) The Company shall be entitled to fill any jobs available on a temporary basis pending the rehiring of individuals with recall rights.
 - f) An individual who is unable to report for work as specified in his notice of

recall because of sickness or accident and so advises the Company within the fourteen (14) day period specified in paragraph (b), and provides satisfactory medical evidence on or before the specified date of rehiring, shall not lose his recall rights solely because of his failure to so report.

- g) On being rehired on recall there shall be deemed to have been no break in such employee's continuous service.
- h) An individual who fails to indicate his acceptance of recall or to report for work as specified in paragraph (b) shall lose all recall rights, except as otherwise specifically provided in this Article 11.
- i) An individual may decline a recall to a job of an expected duration of less than thirty (30) days without loss of recall rights.
- 11.06 An individual with recall rights may present a grievance at Stage Two of the Grievance Procedure and process such grievance through the Grievance and Third Party Determination Procedures if he believes that his layoff or the Company's failure to recall or rehire him was in contravention of this Agreement. Should such grievance succeed and the individual is not otherwise disqualified, he shall be reinstated and any compensation shall be on the basis of the regular basic rate of the job and its normal hours of work (not to exceed forty (40) hours per week) plus any off shift premium or Sunday premium applicable to such normal hours of work, any applicable Cost of Living Allowance, any applicable **Schedule 'N' Bonus** and any holiday pay he would have otherwise been entitled to for the time lost after the receipt of the written grievance by the Company, and during a period of up to thirty (30) days prior to such receipt, less any earnings during such time lost.
- 11.07 An employee who is employed by the Company in any of the classifications mentioned in Schedule 'E' (including an apprentice in the case of a trade classification or a helper who has previously held an unlimited job with the Company as a *tradesperson*) may, within four (4) working days of transfer or demotion to another classification due to lack of work, elect to be laid off and shall be governed by the recall provisions set out in this Article 11. Any other employee who is demoted as a direct result of a layoff due to lack of work may, within four (4) working days and subject to the requirements and efficiency of operations, elect to be laid off and shall be governed by the recall provisions set out in this Article 11. In addition, any such employee laid off either by his own election or for lack of work shall not lose his recall rights by reason only of his failure to report for rehiring in any job other than one in his former classification.

Application of Seniority in a Department

- 11.08 Promotions Any promotion within a Department (other than promotions of a temporary nature for a period not in excess of seven (7) days or promotions to or in supervisory or confidential positions) will be filled with regard to the job posting provisions hereinafter set forth.
- 11.09 (1) Demotions In all cases of demotions within a Department (other than demotions of a temporary nature for a period not in excess of seven (7) days) resulting from a change or reduction in operations in such Department, the following two factors shall be considered in determining which employees are to be demoted:
 - (a) seniority ranking in the Department of the employee affected;
 - (b) the requirements and efficiency of operations and the ability, knowledge, training and skill of the individual to fill the normal requirements of the job. Where factor (b) is to all intents and purposes equal as between two or more employees, their relative seniority ranking shall govern except as hereinafter provided.
 - (2) Inter-Plant Work Assignments In making a work assignment or a temporary work assignment in excess of seven (7) days from a Plant listed in Schedule 'P' of this Agreement to another such Plant within a Multi-Plant Department, the Company will, to the extent permitted by the requirements of operations, give preference according to the seniority ranking of employees in the same occupational classification as the available job.

An employee on such a temporary work assignment will be returned to the Plant from which he was assigned within sixty (60) days. In the event that the assignment extends beyond sixty (60) days, a job will be posted in accordance with the process outlined in **Section** 11.12(3).

(3) It is recognized that it may be necessary from time to time to assign an employee in any occupational classification within a working group to another working group within the same Plant for a period in excess of seven (7) days. In making such assignment, the Company will, to the extent permitted by the requirements of operations, give preference according to the seniority ranking of the employees in the same occupational classification within the working group.

This subsection (3) will not apply to the Power and Construction Plant.

11.10 Application of Seniority - Curtailment in Operations

- a) Transfers If, due to a change or reduction in operations within a Department it becomes necessary to transfer employees out of the Department for a period in excess of seven (7) days the employees affected will be transferred by giving preference in accordance with their seniority ranking in the Department, unless the requirements and efficiency of operations and the ability, knowledge, training and skill of the employee or employees to fill the normal requirements of the job require otherwise.
- b) If the Company decides to discontinue a Department or any substantial portion of a Department it will notify the Union and the local Union in writing and will discuss with it the employees who are to be transferred from the Department or Plant. When additional employees are required in other Departments, such employees will be given the opportunity to transfer in accordance with their seniority ranking in the Department into one of the other Departments, providing the efficiency of operations in the curtailed Department shall not suffer.
- c) In any transfer under sub-paragraphs (a) or (b) the Company will give preference in accordance with their seniority ranking to the requests of employees to be transferred in determining the Plant and Department to which they will be transferred from among the jobs which the Company determines are available.
- d) An employee to be transferred to a new Department under subparagraph (a) or (b) who is retained in the employee's current Department for a period in excess of seven (7) days, will be eligible to apply for a posted job in the employee's new Department and, if the successful applicant, fill that job when his transfer is effective.
- e) Commencing with the date of the employee's transfer to a new Department due to a curtailment in operations, the employee's rate of pay shall not be reduced for a period of three (3) months below the rate of the full-time occupational classification that the employee previously held.
- f) An employee awarded a job under Section 11.14 will receive the higher of the rate of pay of the employee's current occupational classification and the rate of pay of the new occupational classification for the duration of the rate protection provided under Subsection 11.10 e).

11.11 Temporary Promotions, Demotions and Transfers

a) In making temporary promotions, demotions and transfers not in excess

- of seven (7) days, the Company will, to the extent permitted by the requirements of operations, give preference according to the seniority ranking of the employees reasonably available in the Department. With respect to temporary promotion of employees in all Departments except Mines, such preference shall be given to employees who are working on their regularly scheduled shifts.
- b) If the Company transfers an employee from one Department to another and signifies to the employee that such transfer is temporary only, the employee will be returned to the Department from which he was transferred within ninety (90) days. In making such transfers the Company will, to the extent permitted by the requirements and efficiency of operations, give preference according to the seniority ranking of employees within the Department. During the period of temporary transfer he will not acquire any seniority ranking in the Department to which he was transferred but will be able to use his seniority for the purpose of temporary promotions only. In the event an employee is transferred contrary to his preference and seniority ranking to work in an occupational classification carrying a lower rate of pay than the occupational classification from which he was transferred, he shall continue to be paid at the rate applicable to the occupational classification from which he was transferred.

Job Posting

- 11.12 (1) Where a vacancy (other than a vacancy resulting from absence on vacation or of a temporary nature) occurs in any job, limited or unlimited, in wage step three (3) or higher in the case of surface jobs, or in wage step five (5) or higher in the case of underground jobs, notice of such vacancy shall be posted for seven (7) calendar days on the bulletin board or boards in the Department concerned. If such vacancy occurs as the result of a job being vacated or an additional job being added to an existing occupational classification, the job posting notice will apply to the vacated or additional job. The notice shall include:
 - a) the shift (day, afternoon or night) on which the vacancy exists on the date of posting;
 - b) in the case of an underground job in the mines, the level or levels (and, if the vacancy is in a stope or pillar, the number of such stope(s) or pillar(s) and areas of such stopes or pillars) to which the successful applicant will be assigned;
 - c) in the case of jobs in the Exploration Diamond Drilling Department and surface Departments, the plant and the operating Department

- or Departments, building or shop to which the successful applicant will be assigned; and
- d) the schedule number and position of the vacancy on the date of posting;
- e) in the case of the Electrical or Maintenance Department, the working group to which the successful applicant will be assigned.
- (2) In the case of a vacancy in a trade's classification, the Company will post the initial vacancy and up to two (2) vacancies resulting from the filling of that initial vacancy, the last of which shall be called the last vacancy. It may fill the position vacated by the successful applicant for the last vacancy by placing a new hired tradesperson or intermediate tradesperson directly into such vacancy, notwithstanding anything to the contrary contained in the Agreement and subject to Section 20.10 of the Agreement. The successful applicant will not be eligible to apply for another posted job in a different Plant for a period of two (2) years from the posting of his name as the successful applicant; it is understood that this shall not restrict employees' eligibility to apply for another posted job within his current Plant. The Company may grant an exception to this in cases where it determines there are special needs, hardship situations or where operations require otherwise.
- (3) In the event that the Company declares a surplus within a trades classification in a Plant listed under Schedule 'P', and simultaneously identifies a vacancy within that same trades classification at another Plant, the Company will:
 - i.) post the vacancy;
 - ii.) assign the successful applicant to this vacancy;
 - iii.) if a *tradesperson* from a Plant other than the Plant where the surplus exists is the successful applicant on the above job posting, the vacated job will be filled by the assignment of an employee from the initial Plant which declared the surplus, by granting preference in accordance to the seniority ranking of the employees within the trades classification at such Plant.
- 11.13 Any employee in such Department may apply in writing for such job within the seven (7) day posting period:
 - a) whose wage rate is greater or less than the rate of the posted job;
 - b) whose wage rate is the same but who is in a different occupational classification;

- c) who is in the same occupational classification but is assigned to shift work while the posted job is for daytime work or vice versa;
- d) who is in the same occupational classification but is assigned to steady night shift while the posted job is for steady afternoon, night or rotating shift or vice versa:
- e) who is in the same occupational classification but is assigned to swingshift while the posted job is for steady afternoon, night or rotating shift or vice versa:
- f) who is in the same occupational classification but is assigned to a two (2) shift rotation while the posted job is for a three (3) shift rotation or vice versa;
- g) who is in the same occupational classification but is assigned to a limited job while the posted job is for an unlimited job.

In addition to the foregoing, an employee may apply in writing for a posted job in his Department within the seven (7) day posting periods if he is in the same occupational classification and assigned to the same shift, including daytime work, as the posted job. If the employee is the successful applicant for such job, he will not be eligible to make a new application under the terms of this section for a period of six (6) months from the posting of his name as the successful applicant.

Any application under this Section shall be made in duplicate, one (1) copy to be initialled by the Company official receiving it and returned to the applicant.

- 11.14 In deciding which applicant shall fill the vacancy, the Company shall consider the following two (2) factors:
 - a) the seniority ranking in the Department of the applicants affected;
 - b) the requirements and efficiency of operations and the ability, knowledge, training and skill of the applicant to fill the normal requirements of the job. When factor (b) is to all intents and purposes equal as between two (2) or more employees, their relative seniority ranking shall govern, except where otherwise specifically provided in this Article.
 NOTE: Section 11.14 (b) will only apply to limited jobs, in accordance with Letter of Agreement, "Trial Process for Awarding Unlimited Jobs", page 148 CBA.
- 11.15 Part-Time Relief Jobs The Company will determine the maximum number

of employees required to fill regularly recurring part-time relief jobs in each classification in each Department and shall post and award each such job in accordance with the foregoing job posting provisions subject to the following additional conditions:

- a) where a work schedule includes a relief job which consists of working two (2) or more shifts in the same classification within the work week with one (1) or more scheduled changes of shift, the whole sequence of such shifts shall be deemed to constitute one (1) regularly recurring part-time relief job and the complete job must be accepted by the successful applicant.
- b) where a regularly recurring part-time relief job involves steady day work, steady afternoon work, or steady night work or work on a particular rotation of shifts, this fact will be indicated and no employee will be permitted to hold such job unless his regular work is scheduled to coincide with the work involved in such regularly recurring part-time relief job.
- 11.16 As relief are required on each shift for a particular classification, employees who have been awarded regularly recurring part-time relief jobs on that shift will be assigned in order of seniority, provided that:
 - a) where as the result of working on a regularly recurring part-time relief job an employee is required to work on more than one (1) shift in any week, the shift changes involved shall be deemed to be regularly scheduled changes of shift;
 - b) An employee shall not be entitled to be assigned to relief work in a regularly recurring part-time relief job if such assignment would require him to work on either of his regularly scheduled days off.
- 11.17 Limited Jobs Notices posted for vacancies in limited jobs shall state that the job is limited and shall indicate the estimated duration of the job. Upon termination of a limited job, the employee filling the vacancy shall be returned to the classification and job location (if it still exists) in which he last worked which was not a limited job.
- 11.18 The Company shall, without delay, post on the same bulletin board or boards for at least three (3) working days, the name, employment number and length of continuous service of the successful applicant for the job. The successful applicant shall be conclusively deemed to have accepted the job unless he withdrew his application in writing prior to the conclusion of the seven (7) day posting period under paragraph 11.13. If the successful applicant on a limited (other than a trainee job) or unlimited vacancy cannot be assigned to that job because of the Company's inability to select a

replacement for him, he will receive the rate of the posted job not later than fourteen (14) days after the posting of his name as the successful applicant. If there is no applicant or no successful applicant, this fact shall be posted in like manner and the Company may, within (4) weeks of such posting, fill the vacancy by selecting from within the Department the employee with the lowest seniority ranking who is eligible to be appointed under this Section and who, in its judgement (which shall not be exercised in an arbitrary or discriminatory manner), has the ability, knowledge, training and skill to fill the normal requirements of the job. If the Company appoints the eligible qualified employee with the lowest seniority ranking in the Department, or appoints some other person, the name, employment number and length of continuous service of the employee appointed to fill the vacancy shall be posted without delay, for three (3) working days. The junior employee appointed to fill a vacancy shall not, so long as he remains in that job, be appointed under this Section to another vacancy for a period of six (6) months.

- 11.19 The Company will promptly forward to the Local Union copies of notices of vacancies posted showing the disposition including the name, employment number and length of continuous service of the successful applicant or the employee appointed, as the case may be.
- 11.20a) It is recognized that it may be necessary from time to time, in the interest of efficiency of operations, to assign an employee in any occupational classification within a working group from day time work to shift work, or vice versa, or from one shift rotation to another in the same occupational classification, or to another schedule providing for different regular days off in the same occupational classification. In making such an assignment, the Company will, to the extent permitted by the requirements of operations, give preference according to the seniority ranking of the employees in the affected occupational classification. Where the Company determines that any such assignment will be for a period in excess of twenty (20) working days and it occurs within a Plant listed in Schedule 'P' of the Agreement, the Company will give preference according to the seniority ranking of the employees in the affected occupational classification in the Plant.
 - b) In those situations where an employee is assigned to a vacancy under the apprentice posting provisions as outlined in **Section** 20.10 or the trades job posting provisions as outlined in **Section** 11.12(3), or is a new hire, hereinafter referred to as the 'initial vacancy':
 - (i) Employees within the same trade in the same Plant as the assigned employee, working on a different shift schedule and with more seniority than the assigned employee, will be allowed to

- make application for the initial vacancy;
- (ii) If a senior employee is moved to the initial vacancy, the assigned employee will fill this vacated job.

Transfers General

- 11.21 a) An employee who has at least one (1) year of continuous service may, if wishing to transfer from one Department to another, file a written application for such transfer with the Company. A duplicate of the application will be given to the employee. When filling vacancies in the Department (other than a student hired for a fixed term), preference will be given to an employee's application subject to Section 11.22 and the employee's application will be dealt as set out in subsection 11.21 (b).
 - b) As long as there are applications to transfer to the Department, the vacancy(s) will be filled, on a one-for-one basis, by providing opportunities to transfer into the Department equal to the number of new hires into the Department, taking into consideration the following factors:
 - i) the length of continuous service of the employee;
 - ii) the wishes of the employee and the reasons for requesting a transfer:
 - iii) the requirements and efficiency of operations and the ability, knowledge, training and skill of the employee to progress in the Department to which he has made application.
 - c) Any openings created by the transfer of an employee may also be filled by the Company with a newly hired employee.
 - d) In each phase of request transfers, the process of filling vacancies in a Department set out above will continue.
 - e) Successful applicants for transfer will be eligible to receive and apply for job postings in the department to which they will be transferring. An employee awarded such a job will, while awaiting the placement in the new job, receive the higher of the rate of pay of the employee's current occupational classification and the rate of pay of the new occupational classification.
 - f) If a junior successful transfer applicant in a department is transferred before more senior transfer applicants in the same department, the most senior successful transfer applicant awaiting transfer will be entitled to be paid mileage in accordance with the applicable company policy.

- g) Any successful applicant for a transfer under this Section will not be eligible to make a new transfer application for a period of two (2) years from the posting of his name as the successful applicant. The Company may grant an exception to this in cases where it determines there are special needs, hardship situations or where operations require otherwise.
- h) As soon as practicable after his name is posted as the successful applicant, the employee will be advised of the date, subject to the requirements of operations, on which he will be released to his new department.
- 11.22 (a) An employee who is transferred out of a Department due to a general layoff for lack of work or a reduction of force within the Department or for other Company requirements will be deemed to have made written application to retransfer to the Department. When a vacancy occurs in the Department and after making any transfers pursuant to Section 11.10 other than in the circumstances described in subsection (b) of this Section, but before making any transfers pursuant to Section 11.21, preference will be given to the employee's application for retransfer considering the employee's length of continuous service and the employee's ability, knowledge, training and skill to fill the normal requirements of the vacancy.
 - (b) Before making any transfers pursuant to Section 11.10, preference in filling the initial vacancies which the Company has declared for the purpose of effecting Section 11.10 transfers will be given to employees who have an existing application to retransfer to the Department(s) where such vacancies exist and who are senior to the employees being transferred under Section 11.10. Any subsequent vacancies created by these retransfers will be filled under Section 11.10 without regard to any retransfer applications.
 - (c) An employee who is transferred out of a Department in the Mines Section for a reason described in subsection (a) of this Section, may within ninety (90) days of the employee's transfer out make special written application for one time retransfer right to the next job in a Department in the Mines Section available after the other requirements of Section 11.22 are fulfilled. Should the employee refuse such a retransfer when offered, the employee's special application shall be deemed to be withdrawn.
- 11.23 (a) If an employee is temporarily or permanently unable to perform his or her regular work due to age, disease, injury or pregnancy, the Company will endeavour to find employment in the employee's own Department or failing the employee's Department another Department, which in the

opinion of a duly qualified medical practitioner is suitable for the employee. In making such an assignment, the Company may disregard the seniority provisions of this Agreement except that such incapacitated employee shall not be assigned to take over a job occupied at the time by another employee. If employment is found for such an employee in another Department and a duly qualified medical practitioner later determines that the employee is able to return to regular work, the employee shall be deemed to have applied under Section 11.22 for retransfer to their original Department. Each employee who obtains employment under this Section shall, upon request, provide proof satisfactory to the Company that the employee is unable to perform his regular job.

(b) If an employee who obtains employment under this Section is subsequently transferred pursuant to Section 11.10 and is not capable of performing any available job in the Department to which the employee is transferred, the employee may displace the junior employee in the District who occupies a job obtained under this Section` whose job the employee can perform.

Leaves of Absence

- 11.24 The Company will upon the presentation of satisfactory medical evidence of sickness grant sick leave as follows:
 - a) an employee with seniority rights and less than one (1) year of continuous service -- up to one (1) year's leave of absence without pay;
 - b) an employee with one (1) or more years' continuous service -- up to two(2) years' leave of absence without pay.

The Company may extend such leaves for such further period as it may decide.

- 11.25 Upon written request from the Local Union given at least one (1) week in advance, the Company will during each year of the term of this Agreement grant leave of absence, without pay, to the employees named in the request for the purpose of attending:
 - a) one (1) National Policy Conference of the Union;
 - b) one (1) Ontario Federation of Labour Convention;
 - c) one (1) International Convention of the Union;
 - d) one (1) Canadian Labour Congress Convention;
 - e) one (1) Nickel Conference or in lieu thereof one other convention, conference, school or seminar of the Union, Ontario Federation of Labour or Canadian Labour Congress;

- f) four (4) Local Union Schools;
- g) health and safety training, schools or conferences

Such leaves of absence will be limited for each conference, convention, school or seminar to not more than thirty-five (35) employees in the aggregate, of which not more than a reasonable number shall be from any one Department. Not more than twenty-one (21) calendar days' leave in the case of each convention and not more than seven (7) days' leave in the case of each school shall be granted to any employee. The granting of such leave may be withheld when it would be unreasonable having regard to the requirements of operations.

- 11.26 Upon written request by the Local Union given at least one (1) week in advance, the Company will, during each year of the term of this Agreement, grant leave of absence, without pay, for a maximum period of two (2) months, to each of the three (3) employees named in the request, for the purpose of attending the recognized Labour College.
- 11.27 Upon request by the Local Union to the *Human Resources* Office at Copper Cliff, at least forty-eight (48) hours in advance, the Company will, subject to the requirements of operations, grant casual leave of absence to employees for Local Union business for not less than one (1) day or more than fourteen (14) consecutive days at any one time.

The total of all such casual leaves granted during each year of the term of this Agreement shall not exceed twelve hundred (1,200) man days.

However, time spent by employees on Local Union Negotiating Committee during negotiations for renewal of this Collective Agreement and in preparation for such negotiations during the ninety (90) days preceding the commencement of such negotiations or on Local Union Committee at In-Term Review Meetings scheduled by agreement between the Company and the Local Union shall not be included in such twelve hundred (1,200) days. No more than five (5) employees shall be absent from any Department on such casual leave at any one time.

11.28 Upon application by the Union, the Company will grant leaves of absence, without pay, aggregating not more than seven (7) man years during each year of the term of this Agreement. In addition, upon application by the Local Union, the Company will grant, without pay, leaves of absence aggregating not more than six (6) man years during each year of the term of this Agreement to Local Union officers for Local Union purposes. No leave granted under the terms of this paragraph shall be for a period in excess of one (1) year or less than two (2) weeks. However, on request of the Union or Local Union, as the case may be, the period of leave may be extended for not more than one (1) year. No more than thirteen (13) employees will be

- absent on leaves of absence granted or extended as aforesaid under the provisions of this or any previous Collective Agreement.
- 11.29 (a) In the event that an employee is unable to perform or return to the normal duties of her job owing to either a pregnancy or delivery related illness, and this is confirmed by a medical report satisfactory to the Company's Medical Director, the provisions of Section 11.23 (a) will apply. If the employee does not receive a new work assignment, she will be entitled to apply for and, if eligible, receive Sickness and Accident Insurance benefits. These benefits are payable up to and following her delivery date.
 - (b) Employees are entitled to receive pregnancy *or parental* leave in accordance with the provisions of the Employment Standards Act. Employees who exercise leave under these provisions will continue to accrue their continuous service during these leave periods.
- 11.30 Upon written request by the individual concerned, the Company will grant leave of absence without pay to any employee elected to, campaigning for his election to or acting as the campaign manager for an individual's election to any municipal, provincial or federal government office. Such leave shall be for a maximum period of two (2) months in the case of his campaigning or for the term of such office in the case of his election. Leave of absence for an employee acting as the campaign manager shall be for a maximum of two (2) months and shall be limited to one employee for each candidate.
- 11.31 Upon written application to his immediate supervisor by an employee with ten (10) or more years of continuous service, the Company may grant such employee a one time special personal leave of absence without pay of not less than two (2) months and not more than six (6) months. Any such leave of absence shall be taken outside the prime vacation period as determined by the parties and shall not be used in whole or in part to take other employment. An employee who takes such a one time special personal leave of absence will be credited with continuous service for the length of the leave provided that the employee returns to work for a period of time equalled to the length of the leave.

General

11.32 An employee will be considered on probation until completion of **six** (6) months of continuous service with the company from the date of last hire. Upon completion of the probationary period, his seniority ranking shall be determined and shall include such **six** (6) month period. The discharge of a probationary employee may be for any reason at the discretion of the company.

- 11.33 The Company retains the right to transfer to the category of employee any person employed by the Company in any other category. Any person so transferred will be credited on the date of transfer with seniority ranking equivalent to his or her length of Company service on that date.
 - The Company will advise the Vice-President of the Local Union on a monthly basis, of employees who have taken temporary or full-time assignments outside of the bargaining unit.
- 11.34 Notwithstanding any other provisions of this Article, the Company may from time to time designate to the Local Union, individuals (including technical or scientific students in part-time employment) who are to be given special experience or training to prepare them or test their capabilities for other or broader assignments or for future service other than to the Company. The Company may promote, demote and direct their efforts, free from any limitations imposed by this Article, until such special experience or training is completed, provided that no such designated individual shall be employed in any one occupational classification for more than six (6) months. The total number of individuals designated at any time shall not exceed the lesser of two hundred and fifty (250) or five per cent (5%) of the employees. No such designated individual will be assigned to take over the job occupied at the time by an employee nor will he be eligible for temporary promotions of seven (7) days or less unless he has greater seniority ranking than other employees who are reasonably available. Any overtime worked by a designated individual will be recorded for the purpose of overtime distribution in the classification and working group to which he was assigned prior to being designated. The Company will advise the Local Union of the nature and location of the work to be performed by the individual.
- 11.35 The Company may hire students as employees for a fixed term to perform work during the period of April 1 to September 15 in any year. The Company may also hire students on co-operative training programs to perform work provided that there cannot be more than one (1) such student for every hundred (100) employees at any time during the year. The termination of any such student's employment at or about the end of his fixed term shall not be subject to grievance or arbitration. The Company will notify each student in writing of the period of time for which he is being employed, with a copy to the Local Union. Individuals with recall rights shall be given notice of recall prior to the hiring of such students.
- 11.36 At least once every three months the Company shall prepare the Company seniority ranking and seniority ranking lists, showing employee classifications, for each Department. Departmental lists shall be available at the Company's General Office at Copper Cliff and a copy applicable to that Department in an office of each Plant in each Department for reference by employees at reasonable times. Access to the lists at the General Office

shall be available at reasonable times to the Staff Representative of the Union regularly assigned to the Local Union or to the President or Secretary or Plant Committeemen of the Local Union. A copy of the seniority lists shall be sent to the Local Union and one copy of the appropriate Departmental Seniority list will be sent to each Plant and given to the Plant Committeeman or Plant Committeemen, as the case may be, for use at the Plant.

ARTICLE 12 - ESTABLISHMENT OF NEW DEPARTMENTS, TRANSFERS AND SPECIAL SENIORITY PROVISIONS

- 12.01 It is understood that during the currency of this Agreement the Company may establish a new Department or Departments. Such new Department or Departments may include employees employed in other existing Departments.
- 12.02 It is recognized that the putting into operation of a new Department or Departments may require the transfer of employees from other Departments and that in these circumstances it is recognized that the application of the provisions of Article 11 would not be in the interests of either the Company or the employees involved and that it is accordingly desirable to declare the provisions with respect to transfer of employees and their seniority which should be applied during the period of putting into operation a new Department or Departments.
- 12.03 It is also recognized that it is not possible at this time to determine when a new Department may be established or to fix the date of commencement of first operation of any new Department or Departments. Consequently, when the Company establishes any new Department it shall notify the Union of the establishment and of the affected Departments as herein defined. The Company will subsequently fix a date for the commencement of first operation of such new Department (hereinafter called the "starting date") as soon as it becomes possible to do so and shall notify the Union thereof not less than ten (10) days before such starting date of such new Department.
- 12.04 During the period commencing with the starting date of any new Department and continuing until such date as the Company shall have declared such new Department to be in full scale operation (the date of full scale operation being hereafter defined), the following provisions shall apply notwithstanding the provisions of Article 11 of this Agreement:
 - a) For the purposes of this Article 12, the expression 'affected Department' shall mean such existing Departments as may be named by the Company, and the expression 'new Department' shall mean the new Department which has been determined by the Company and of which notification has been given to the Union as aforesaid.

- b) The date of full scale operation shall in the case of a new Mine Department be not more than six (6) months after the starting date, and in the case of any other new Department be not more than ten (10) months after the starting date, provided, however, that in the event of technological or major mechanical difficulties in a new Department other than a Mine Department the Company may extend the date of full scale operation for a further period not exceeding five (5) months.
- c) As soon as practicable after the giving of notice of the establishment of a new Department the Company will post a notice in all affected Departments which will list the occupational classifications of employees (other than Surface Labourers, Process Labourers, and Underground Labourers) and the number in each classification which is required at the time. Such notice will also fix a date up to which applications for transfer to the new Department and to specific occupational classifications will be received by the Plant Personnel Officer or other designated Company representative. Any employee in an affected Department will be eligible to make application. The Company will on the starting date transfer the applicants to the number required in the order of what would be their seniority ranking at that time in the new Department provided that the requirements and efficiency of operations in the affected Department and the new Department do not suffer and that the applicants have the ability, knowledge, training and skill to fill the normal requirements of the job for which they have applied. Whenever following the starting date the Company requires additional employees in the new Department it will post a notice or notices in the new and affected Departments similar in form and content to the notice hereinbefore referred to and any employee in the new and affected Departments will be eligible to apply. Subject to the same conditions set out with respect to what would be their seniority ranking at that time in the new Department if they are transferred to the new Department and to the requirements and efficiency of operations and the ability, knowledge, training and skill of the applicants the Company will transfer to the new Department or promote the number of employees required.
- d) At any time during the said period an employee who has been transferred from an affected Department to a new Department may be retransferred at his own request or by the Company to his original Department, in which event, however, he shall be returned to the classification of the unlimited job most recently held by him in such affected Department.
- e) During the said period the expression 'limited job' as defined in Article 11 hereof shall be enlarged to include any job in an affected Department which may be limited in duration by reason of the possible retransfer to

- that affected Department of any employee who has been transferred to such new Department.
- f) If at any time during the said period a vacancy occurs in any job in any of the affected Departments other than a Mine Department, such vacancy shall be posted in the new Department in addition to such affected Departments and any employee who is then employed in the new Department and has been transferred from such affected Department under the provisions of sub-paragraph (c) hereof shall be entitled to apply for such job in accordance with and subject to the provisions contained in Article 11.
- g) In making temporary promotions and demotions in the new Department during the said period the Company will, to the extent permitted by the requirements of operations, give preference according to the seniority ranking of the employees reasonably available. With respect to temporary promotions of employees in other than a Mine Department the Company will give such preference to employees who are working on their regularly scheduled shifts.
- h) In the event of an employee wishing to transfer from any Department to the new Department, he may file with the Company a written application in accordance with Article 11.
- 12.05 At the end of the said period the seniority ranking of the employees in the new Department and affected Departments shall be determined in accordance with the foregoing provisions as of the date on which the Company shall have declared the new Department to be in full scale operation. Thereafter the provisions of Article 11 shall govern all questions of seniority in the said Departments.

ARTICLE 13 - HOURS OF WORK - OVERTIME

- 13.01 The work day for each employee is as follows:
 - a) Day Workers The day for each day worker commences with the start of his regular working hours in that day. Day workers shall normally commence work at eight (8) a.m. but in the interest of efficiency of operations certain day workers may be required to commence work earlier or later than eight (8) a.m. Day workers shall be given a continuous lunch period of one-half (1/2) hour which shall commence not later than five (5) hours after the start of their work but such lunch period shall not be considered as time worked.
 - b) Shift Workers The day for each shift worker commences with the start

of his day shift hours. Shift workers will be allowed thirty (30) minutes per shift to eat a meal and such time will be considered as time worked. A day worker who receives a change of shift instruction before the end of his shift on Friday of the preceding week which requires him to work two or more afternoon and/or night shifts during a week will be considered a shift worker for that week.

- c) Time worked will be calculated in units of one-half (1/2) hour.
- 13.02 The work week for each employee commences with the start of his day on Monday.
- 13.03 Applicable Hourly Rate means the straight time rate per hour provided under Schedules 'F' and 'G' and applicable to the work an employee is performing at any time, but does not include holiday pay, Sunday premium, shift premium or overtime premium.

13.04 Overtime

The overtime rate will be one and one-half (1 1/2) times the applicable hourly rate.

13.05 <u>Daily Overtime</u> - An employee who has worked eight (8) hours at his applicable hourly rate in any twenty-four (24) hour period will be paid at the overtime rate for any additional time worked in such period, except where the additional time worked is due to his regularly scheduled change of shift.

Weekly Overtime

- a) An employee who has worked forty (40) hours at his applicable hourly rate in his work week will be paid at the overtime rate for any additional time worked in such week less amounts paid to him pursuant to the immediately preceding paragraph of this Article.
 - b) Hours paid to an employee at the overtime rate because of an unscheduled short change of shift will be considered time worked at his applicable hourly rate in calculating weekly overtime, provided he has not worked more than eight (8) hours in either day involved in the change of shift.
- 13.07 Should an employee be instructed by the Company to change his regular day or days off in a work week to another day or days in that week, and receive such instruction after the end of his shift on Friday of the preceding week, he shall be considered as having worked his normal working hours on those days rescheduled as his days off, but for the purpose only of determining entitlement to the overtime rate for any other time actually worked by him.

- 13.08 Should an employee be instructed by the Company to change his shift and he receives such instruction after the end of his shift on Friday of the preceding week, such employee shall, except in the case of an unscheduled short change in shift, be considered as having worked an additional eight (8) hours in his work week but for the purpose only of determining entitlement to the overtime rate for any other time actually worked by him.
- 13.09 When overtime work, other than that performed on a statutory holiday, is scheduled by the Company it shall distribute such work as evenly as practicable among the employees in the working group and for this purpose it shall take into consideration the availability of the employees in the same classification who can do the work and the wishes of the employees. Upon request to his immediate supervisor an employee or his Grievance Steward (or if there is no Grievance Steward in his working area, his Plant Committeeman) may examine the record of his immediate supervisor pertaining to overtime distribution within the employee's working group.
- 13.10 Notwithstanding anything in this Article, an employee shall not be entitled to more than two and one-half (2 1/2) times his applicable hourly rate for time worked on plant holidays or to more than one and one-half (1 1/2) times his applicable hourly rate for time worked on other days, unless such additional payment is made in respect to premiums for shift work or Sunday work or the four (4) hours minimum for a 'call out'.

Recognized Holidays

13.11 The Plant holidays recognized under this Agreement shall be:

New Year's Day Labour Day

Family DayThanksgiving DayGood FridayRemembrance DayVictoria DayChristmas DayCanada DayBoxing Day

Civic Holiday

(the first Monday in August)

In addition, each employee will be entitled to one floating holiday with pay in each year of the Agreement. Such holiday will be scheduled on a day agreed upon between the employee and his supervisor subject to the requirements and efficiency of operations and the wishes of the employee.

Each employee shall take his floating holiday by May 31st for each year of the Agreement or shall forfeit the holiday. Any employee who forfeits the holiday shall be paid eight (8) hours at his applicable hourly rate.

13.12 Should an employee be required to work on a recognized holiday he shall be

paid for work performed at the rate of two and one-half (2 1/2) times his applicable hourly rate. If the employee did not volunteer to work on the recognized holiday and is assigned to a shift other than the shift he would otherwise have worked on that day, he shall also be paid eight (8) hours at his applicable hourly rate.

- 13.13 Where work, including overtime work, is scheduled by the Company on a recognized holiday it shall assign such work among the employees in the working group (excepting those employees for whom the recognized holiday is their regular scheduled day off) and for this purpose it shall take into consideration the availability of the employees in the same classification who can do the work, the wishes of the employees and their seniority ranking. Overtime work performed by an employee on a recognized holiday will be recorded for the purpose of overtime distribution under Section 13.09 of the Agreement. In the case of non-continuous operations, work will be assigned on a voluntary basis.
- 13.14 An employee who does not work on a recognized holiday will be paid eight (8) hours at what would have been his applicable hourly rate had he worked on that day, provided he will not be so paid:
 - a) if he has been absent from work on either of the qualifying days which are his last regular work day or shift before and his first regular work day or shift after such holiday; or
 - b) if he has been instructed to report for his regular work on such holiday and has failed to do so; or
 - c) if he has been granted leave of absence or casual time off for that holiday.

An employee will be granted holiday pay if his absence on days referred to in (a) or (b) above is due to leave of absence for Jury Duty, Bereavement or Local Union business, or due to being subpoenaed as a crown witness in any court of law or due to illness or accident which requires the employee to be absent from work for five (5) consecutive days, including the holiday in the case of an employee instructed to work on such holiday, or including any qualifying days on which he was absent in the case of an employee not instructed to work on such holiday, and a doctor's certificate to that effect is submitted by the employee. An employee who has not worked in the thirty (30) day period immediately preceding such holiday *for reasons other than vacation* shall not be entitled to compensation hereunder.

13.15 Where by proclamation or otherwise another day is established as a holiday in lieu of Canada Day, the provisions of this Article shall apply to the holiday so established and not to July 1st.

- 13.16 Except where the employee's regular day off occurs on a recognized holiday, when an employee is entitled to be paid for a recognized holiday not worked, the time so paid shall be considered time worked at his applicable hourly rate but for the purpose only of determining entitlement to the overtime rate for any other time actually worked by him.
- 13.17 When an employee works on a recognized holiday the time so worked shall be considered time worked at his applicable hourly rate but for the purpose only of determining entitlement to the overtime rate for any other time actually worked by him.

13.18 Sunday Premium

A Sunday premium of one dollar (\$1.00) will be paid for all time worked by an employee between the start of the day shift on Sunday and the start of the day shift on Monday.

13.19 Shift Premium

An employee shall be paid an off shift premium of fifty (50) cents per hour for time worked by him during his afternoon shift hours and an off shift premium of seventy (70) cents per hour for time worked by him during his night shift hours.

13.20 Collar to Collar

Hoisting and lowering schedules at the Company's Mines will be arranged to provide approximately eight (8) hours from collar to collar for each shift, but variations from such schedules, not exceeding ten (10) minutes, which do not occur consistently, shall be disregarded. Delays in such schedules in excess of ten (10) minutes shall be considered time worked but not for the purpose of applying the overtime provisions of this Article.

13.21 4-Hour Minimum

When an employee, because of failure of the Company to inform him by notice or otherwise that no work will be available, reports for work on schedule, in good faith, and is advised there is no work available he shall receive four (4) hours' pay at what would have been his applicable hourly rate but such four (4) hours shall not be considered time worked for the purpose of applying the overtime provisions of this Article. This paragraph shall not apply when an employee has been absent from his regular work and has failed before reporting for work to inform his foreman or such other supervisor designated by the Company for this purpose of his intention to return.

13.22 Call Out

An employee reporting for work on a call out shall be paid for work performed at one and one-half (1 1/2) times his applicable hourly rate, or at least four (4) hours at his applicable hourly rate, whichever is the greater.

To qualify for call out allowance, an employee shall have:

- a) been requested to work outside of his regularly scheduled hours; and
- b) been notified to commence work at a time other than time immediately following a shift worked by the employee or at a time which is more than one (1) hour immediately preceding a shift worked by the employee; and
- c) received notice from the Company less than sixteen (16) hours before the time he was to commence such work.

An employee who reports for work on a call out on a recognized holiday and who qualifies for recognized holiday pay under Section 13.14 of this Agreement shall not have any reduction made in that holiday pay because of the call out.

13.23 Transportation

Transportation to and/or from call out work will be provided by the Company when public bus transportation is not reasonably available or when in the opinion of the Company the urgency of the call out work requires it.

13.24 Lunches

Where an employee is held at work for a period of more than two (2) hours immediately following (8) hours of work (or ten (10) hours of work if working on his or another employee's ten (10) hour shift; or twelve (12) hours of work if working on his or another employee's twelve (12) hour shift), the Company will provide him with a lunch. Such employee will be provided with an additional lunch if such period of being held at work is more than five (5) hours.

If an employee who reports for work on a call out performs more than four (4) hours of work outside their regularly scheduled hours, they will be provided with a lunch. The Company will use its best efforts to supply a hot lunch to any such employees where practicable.

13.25 Work Day - Work Week

- The Company does not guarantee to provide work for any employee, nor to maintain the work week or working hours presently in force.
- 13.26 The Company and the Union may enter into agreements to give effect to shift schedules which alter or exceed the daily and/or weekly hours of work contained in this Agreement.
- 13.27 In situations where the parties have agreed to implement shift schedules other than regular eight (8) hour shift schedules, the provisions of Schedule 'Q' shall apply to those employees working on those shift schedules.

ARTICLE 14 - WAGES

- 14.01 The Company and the Union agree that, during the term of the Agreement and except as otherwise provided, the wage rates shall be as set out in Schedules 'F' and 'G' attached, with increases effective as of the dates set out in such Schedules.
- 14.02 The wage rate for *tradespersons*, including leaders and intermediates, shall be two job class increments above their wage rates as set out in Schedule 'F'.
- 14.03 An employee who is temporarily assigned to work in an occupational classification carrying a higher rate of pay than the occupational classification from which he was assigned shall receive such higher rate of pay immediately.
- 14.04 An employee who is temporarily transferred or assigned to work in an occupational classification carrying a lower rate of pay than the occupational classification from which he was transferred or assigned shall be paid in accordance with the following provisions:
 - a) If there is no work reasonably available for him in the occupational classification from which he was transferred or assigned he shall be paid the rate of the occupational classification to which he is transferred or assigned.
 - b) If there is work reasonably available for him in the occupational classification from which he was transferred or assigned he shall continue to be paid at the rate applicable to the occupational classification from which he was transferred or assigned. Provided that an employee will not be paid for work performed during a shift or day at a rate less than that applicable to the occupational category in which he

commences work on such shift or day.

- 14.05 a) Technological Change For the purpose of this Section, 'technological change' means the introduction of an innovation in equipment, material or process used by the Company in its Departments and Plants in the Sudbury District as described in Schedules 'D' and 'P' of this Agreement or at the Port Colborne Plant which directly results in a change in the manner in which the Company carries out its work. If the Company decides to effect a technological change that will affect the security of employment of a significant number of employees, it will give the Union and Local Union written notice of the change at least sixty (60) days prior to technological change coming into effect.
 - b) A joint Technological Change Committee will be established consisting of six (6) members, three (3) to be appointed by the Local Union and three (3) to be appointed by the Company. The Committee will meet at least once every three (3) months and at more frequent times if agreed upon for the purpose of discussing matters in connection with technological change.

The Company shall pay employee members of the Committee at their applicable hourly rate plus any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus** for any time lost during their regular working hours while attending Committee meetings.

- c) The following provisions shall apply to any employee who has acquired seniority rights under the provisions of this Agreement, and who is either transferred from his Department as a result of technological change, or who remains in his Department and is directly affected by such change by reasons of the fact that his occupational classification is discontinued, or he is placed in a different occupational classification. As of the date of his occupational classification being altered, the employee's rate of pay shall not be reduced for a period of three (3) years below the rate of occupational classification which he previously held and shall not subsequently be reduced by more than one (1) step in rate in each succeeding year up to a maximum of eight (8) years from the date of his occupational classification being altered. It is a condition of this Section that any such employee:
 - must accept any training in any occupational classification which is offered to him by the Company otherwise;
 - ii) will be deemed to be an applicant for any job vacancy which is posted in his Department for a job the rate for which is higher than the rate of his actual occupational classification at the time of posting and that if he is named as the successful applicant for any such job he cannot

refuse to accept the promotion otherwise;

- iii) if he applies under the provision of Section 11.21 of this Agreement for transfer from the Department in which he is located subsequent to the technological change and he is transferred as a result of his application, he will cease to be entitled to the benefits of this Section 14.05.
- d) If an employee who has one (1) year or more of seniority is laid off for a period exceeding one (1) year as a direct result of a technological change, he shall be paid in weekly amounts a sum equal to forty (40) times the applicable hourly rate of his occupational classification at the time of layoff multiplied by his years of Continuous Service, to a maximum of twenty-six (26), as of the date of such layoff. An employee shall not be entitled to continue to receive weekly payments if he returns to work.

14.06 Jury Duty and Crown Witness

Should an employee be required to report on their regular work day for jury duty or as a witness due to being subpoenaed as a crown witness in any court of law and produces a satisfactory statement that he/she did so report, the employee will be paid the difference between pay for the regularly scheduled shift on that date at the employee's applicable hourly rate plus any applicable shift premium and any applicable Cost of Living Allowance and **any applicable Schedule 'N' Bonus** whichever is greater and the jury duty pay or witness pay and/or conduct money received by the employee for that day.

14.07 Bereavement Leave

A maximum bereavement leave of absence of three (3) consecutive days, including the day of the funeral will be granted to an employee upon a death in the employee's immediate family. Where any such day occurs on a regularly scheduled working day for the employee, the employee shall be paid on the basis of the standard number of hours which otherwise would have been worked at the employee's applicable hourly rate of pay plus any applicable shift premium and any applicable Cost of Living Allowance and any applicable Schedule 'N' Bonus. To qualify for bereavement leave, the employee shall notify his or her immediate supervisor as soon as possible following the bereavement. `Immediate family' shall mean mother-in-law, father-in-law, daughter-in-law, son-inlaw, spouse's brother, spouse's sister, employee's brother's spouse, employee's sister's spouse, grandparent, spouse's grandparent or grandchild. This maximum bereavement leave of absence will be increased to five (5) consecutive days, including the day of the funeral, in the case of the death of the spouse, son, daughter, mother, father, sister or brother of the employee.

If the death in the immediate family occurs and the bereavement leave begins while the employee is on a regular or special vacation, the employee will be allowed to extend his vacation in an amount equal to the number of bereavement days during such vacation.

14.08 Pregnancy Leave Supplemental Pay

- (1) Pregnancy leave will be granted in accordance with the provisions of The Employment Standards Act (Ontario).
- (2) An employee who is granted pregnancy leave will be provided Supplemental Pay for a maximum period of seventeen (17) weeks provided that she:
 - (a) has completed at least one (1) year of continuous service with the Company immediately prior to the commencement of her pregnancy leave;
 - (b) provides to the Company satisfactory proof that she has applied for and is in receipt of Employment Insurance maternity benefits; and
 - (c) signs an agreement with the Company which indicates that she will return to work for a period of at least six (6) months on the expiry of her pregnancy leave or, if she takes parental leave immediately following her pregnancy leave, on the expiry of her parental leave and that she will immediately repay to the Company the amount of Supplemental Pay provided to her in the event that she fails to return to work for a period of at least six (6) months for reasons other than disability, death or layoff on the expiry of her pregnancy leave or, if she takes parental leave immediately following her pregnancy leave, on the expiry of her parental leave.
- (3) The amount of Supplemental Pay provided to an eligible employee will be as follows:
 - (a) for the two (2) week waiting period for Employment Insurance maternity benefits, ninety percent (90%) of the employee's basic weekly wage in effect immediately prior to the commencement of her pregnancy leave; and
 - (b) for each week of pregnancy leave in respect of which the employee receives Employment Insurance maternity benefits,

to a maximum of fifteen (15) weeks, ninety percent (90%) of the employee's basic weekly wage in effect immediately prior to the commencement of her pregnancy leave, less the gross amount of Employment Insurance maternity benefits paid to the employee.

- (c) For the purpose of this Section 14.08, "basic weekly wage" shall mean the employee's gross basic weekly wage (before income tax and other deductions) in effect immediately prior to the commencement of the pregnancy leave irrespective of any increases to basic weekly wages that may be granted during the leave of absence.
- (4) The Company may at any time require an employee on pregnancy leave or entitled to pregnancy leave pursuant to this Article, to provide certification from a qualified medical practitioner of her condition including the expected and actual date of her delivery. In addition such an employee may be required by the Company prior to the employee returning to work from pregnancy leave or, if she takes parental leave immediately following her pregnancy leave, parental leave to present the Company with the written opinion of a qualified medical practitioner that she is able to perform the normal duties of her job.
- (5) Supplemental Pay will be reduced by income tax, Canada Pension Plan contributions, and any other deductions required by law or applicable collective agreement.
- (6) Supplemental Pay will be reduced by any other earnings that are allocated to the period by Human Resources and Social Development Canada.
- (7) The combined weekly rate of El maternity benefits and weekly Supplemental Pay will not exceed ninety percent (90%) of the employee's basic weekly wage. The Company does not compensate employees for any income tax claw-back of maternity El benefits due to the employee's total income.

ARTICLE 15 - VACATIONS

15.01 During the vacation year commencing January 1st, **2010**, each employee whose first (1st) or any subsequent anniversary of employment falls within the said vacation year shall receive two (2) weeks' vacation with pay which shall not commence until after such anniversary. Each employee whose third (3rd) or any subsequent anniversary of employment falls within the said

vacation year shall receive an additional one (1) week's vacation with pay which shall not commence until after such anniversary. Each employee whose fifteenth (15th) or any subsequent anniversary of employment falls within the said vacation year shall receive an additional one (1) week's vacation with pay which shall not commence until after such anniversary. Each employee whose twenty-fifth (25th) or any subsequent anniversary of employment falls within the said vacation year shall receive an additional one (1) week's vacation with pay which shall not commence until after such anniversary of employment falls within the said vacation year shall receive an additional one (1) week's vacation with pay which shall not commence until after such anniversary.

- 15.02 During the vacation years commencing January 1st, 2010, January 1st, 2011, January 1st, 2012, January 1st, 2013, January 1st, 2014 and January 1st, 2015, the provisions of Section 15.01 shall be equally applicable in respect of an employee's vacation during each of such vacation years except that the period within which an employee's anniversary of employment falls shall be the vacation years commencing January 1st, 2010, January 1st, 2011, January 1st, 2012, January 1st, 2013, January 1st, 2014 and January 1st, 2015 as the case may be instead of January 1st, 2015.
- 15.03 A week of vacation means seven (7) consecutive days including Saturdays, Sundays and holidays (normally eight (8) a.m. Monday to eight (8) a.m. the following Monday). Where an employee is entitled to two (2) weeks' vacation such weeks shall be consecutive. An employee entitled to three (3) or more weeks may split his vacation into not more than *three* (3) parts, each of which must consist of a period of one (1) or more complete weeks and only one (1) of such parts shall commence between May 30th and the following September 1st. In the event that the Company schedules an employee to work during a vacation shutdown, regular vacation scheduled by such employee immediately preceding and following this vacation shutdown period will be considered as one (1) continuous period of vacation. Notwithstanding the foregoing in the event of a vacation shutdown employees will be expected to take their vacation during the period of shutdown, subject to the requirements of operations.
- 15.04 An employee will be granted and shall take his vacation for each vacation year within twelve (12) months of the anniversary of his employment at such times as the Company finds most suitable, considering in each Department the employee's seniority, his wishes and the efficient operation of the Plant. Vacation dates may be changed by the Company when it considers it necessary for efficient operations. A vacation list will be posted in each Department as soon as practicable after vacations for the employees of the Department have been scheduled, the list to show the name, number and

- seniority of each employee and date of his scheduled vacation.
- 15.05 An employee will be paid, for each week of vacation to which he is entitled in a vacation year, an amount equal to 2% of his earnings from the Company in the twelve (12) months immediately preceding his anniversary of employment falling within that vacation year. An employee will receive an additional \$325 of vacation pay for each week of vacation entitlement under this Article.
- 15.06 In cases where an employee's employment with the Company has been interrupted the Company may grant to such employee, in whole or in part, the vacation pay to which such employee would have been entitled had no such interruption occurred.
- 15.07 Wherever in this Article the expression 'date of employment' is used it shall mean the date of last hiring and the expression 'anniversary of employment' shall have a comparable meaning; provided that in the case of each employee whose date of employment is more than one (1) year prior to the second January 2nd which occurs during his employment, the second anniversary of his employment and all subsequent anniversaries thereof shall for the purposes of this Article be deemed to be January 2nd in each year.
- 15.08 On cessation of employment, and in lieu of vacation with pay, the Company will pay to an employee an amount equal to the sum of:
 - a) the vacation pay for any vacation to which he is entitled at the time of cessation but which he has not received, and
 - b) a percent of his earnings from the Company in the period since the date of his employment or of the last preceding anniversary of his employment, whichever shall be later, up to the date of such cessation in accordance with the following:
 - before 3rd anniversary of employment 4%
 - after 3rd and before 15th anniversary of employment 6%
 - after 15th and before 25th anniversary of employment 8%
 - after 25th and before 30th anniversary of employment 10%
 - after 30th anniversary of employment 12%

ARTICLE 16 - SPECIAL VACATIONS

16.01 The Company will during the currency of this Agreement provide Special Vacations with pay in accordance with the following provisions:

- a) Each employee who completes his first five (5) years on the anniversary date of his employment shall become entitled to five (5) weeks of Special Vacation with pay in addition to vacation entitlements under the provisions of Article 15 of this Agreement.
- b) Each employee upon completion of each additional five (5) years on the anniversary date of his employment since last becoming entitled to a Special Vacation with pay shall thereupon similarly become entitled to five (5) weeks of Special Vacation with pay.
- c) The expression 'anniversary date of his employment' in this Article shall be determined on the same basis as set out in Section 15.07.
- d) An employee can have access to his next Special Vacation entitlement to the extent of one (1) week of Special Vacation for each completed year of employment during the five (5) year period either before becoming entitled to or between entitlements to Special Vacation, to a maximum of four (4) weeks of Special Vacation. The employee cannot have any outstanding Special Vacation at the time he accesses Special Vacation under this Section. Any Special Vacation taken under this Section will be deducted from the employee's next Special Vacation entitlement.
- e) The pay for each week of Special Vacation which is taken shall be equal to two per cent (2%) of the employee's earnings from the Company in the calendar year immediately preceding the year in which such week of Special Vacation is taken.
- f) Should an employee after becoming entitled to Special Vacation with pay fail to take it within five (5) years of the date of entitlement for any reason including death or retirement, the Company will pay the employee or his estate in lieu of the vacation time. The amount of payment will be calculated as though the employee had taken the vacation immediately prior to the fifth (5th) anniversary of his entitlement or to the date of his termination of employment or his death as the case may be.
- g) In determining the length of a Special Vacation a week shall mean seven
 (7) consecutive days including Saturdays, Sundays and holidays falling within the period.
- h) The allocation of Vacations with Pay under the provisions of Article 15 shall have priority over the allocation of Special Vacations granted under this Article.

- i) In order to minimize interference with the normal operations of the Company, Special Vacations will be granted only at such times and in such amounts as the Company, in its discretion, may determine but subject thereto due consideration will be given to the wishes of the individual employee. It is anticipated that, in most cases, an employee's Special Vacation will be taken within the five (5) year period following the date on which he becomes entitled to it and it is hoped early in that period.
- j) In the case of any employee who retires or otherwise ceases to be employed by the Company or dies, he or his estate shall be entitled, in addition to the payment provided for in subparagraph (e) of this Article, to an amount equal to two per cent (2%) of his earnings in the calendar year preceding his retirement, cessation of employment or death for each full year from the date of his last entitlement to a Special Vacation.
- k) An employee who is entitled to Special Vacation with pay may take in any calendar year up to forty (40) hours of that Special Vacation in individual days. Each day taken will be equal in length to the number of hours the employee would have worked in his regular shift on that day and pay for that day will be equal to the pay he is entitled to under subsection (e) above divided by the number of hours in his regular shift on that day. If at the end of the calendar year the total number of hours taken is less than forty (40), the employee will be paid the difference between the amount of Special Vacation pay received under this subsection and a week of Special Vacation under subsection (d) unless the employee elects to carry this remaining time into the next calendar year in accordance with Subsection (l). An employee working on a twelve (12) hour shift schedule, will have the option to schedule up to forty (40) hours of Special Vacation, made up of three (3) twelve (12) hour shifts and one (1) four (4) hour period of such Special Vacation.
- I) An employee who has taken any special vacation in individual days pursuant to subsection (k) may elect to carry over any remaining hours into the next calendar year by notifying his/her supervisor in writing, prior to December 1. This remaining time must be used before commencing a new entitlement of individual days of special vacation under subsection (k). Any unused remaining time carried over from a previous year will be paid out at the end of the calendar year.

ARTICLE 17 - HEALTH, SAFETY AND ENVIRONMENT

17.01 The Company, Union and Local Union acknowledge their common concern for maintaining a healthy and safe working environment. In order to effect a thoroughly understood and accepted health, safety and environment

program for employees at work it is agreed that joint and co-operative methods shall be encouraged.

To this end, joint Health, Safety and Environment Committees will be established for the Operations as set out in Schedule 'J' attached to this Agreement. In addition, the Union, the Executive of the Local Union and, at the request of the Executive of the Local Union, the General Health, Safety and Environment Committee of the Local Union shall have the right at any time to discuss matters dealing with health, safety and environment conditions affecting employees while at work.

17.02 The Company shall continue to make provisions for the health and safety of its employees and shall continue its existing safety devices and practices for the purpose of protecting employees from injury, accident and unhealthful conditions subject to such improvements or changes as may be introduced by the Company from time to time. In making any changes affecting employees of an Operation with respect to either the Health and Safety Program, the Workplace Environment Program or a Personnel Protective Program, such changes will be discussed prior to their introduction at a meeting of the appropriate Operation Health, Safety and Environment Committee in order to give the Committee an opportunity to review the changes and make suggestions regarding possible alternative arrangements or programs.

The Union and the Local Union agree to co-operate with the Company in developing and maintaining a strong sense of safety awareness among employees. In this connection, it is recognized that every employee has the right to report unsafe conditions and practices to his immediate supervisor.

The Company undertakes to provide each Operation Health, Safety and Environment Committee once a month with a daily and monthly summary of injuries for review and discussion by the Committee. The Company will also provide the Local Union on request with such summaries and with a list of employees who have applied for Workers' Compensation benefits.

17.03 The General Health, Safety and Environment Committee of five (5) employees appointed by the Local Union and consisting of the General *Chairperson*, one permanent Mine Co-Chair, one permanent Operations Co-Chair, one permanent Maintenance and Electrical Co-Chair and one Recording Secretary, is hereby recognized.

The Company will, notwithstanding anything to the contrary contained in this Article, grant the *Chairperson* of the General Health, Safety and Environment Committee or his designate up to a maximum of five (5) days

leave of absence with pay in each week at the equivalent of the wage rate for Maintenance Electrician, plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus**, for the purpose of attending all Safety Executive Committee meetings, dealing with union business pertaining to the internal administration and organization of the General Health, Safety and Environment Committee and for the purpose of performing any other functions, including the attending of meetings, under the provisions of this Article.

The Company shall endeavour to release the Committee members from their plants to attend all scheduled Committee meetings. The Company shall pay members of the Committee at their applicable hourly rate plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus** for any time lost during their regular working hours while attending Committee meetings.

17.04 An Executive Safety Committee shall be established consisting of the General Health, Safety and Environment Committee, three (3) senior Management Representatives, including a Manager of a Surface Operation, a Mine Manager and the Manager of Health, Safety and Environment or their designates. The Local Union and the Company shall each designate a member as **Co-Chair**.

The Committee shall meet monthly, and at such other times as are agreed to, to review and monitor the safety performance of the Ontario Operation, review and make recommendations with respect to new health, safety and environment initiatives, ensure that relevant information is forwarded to other committees, request input from appropriate management and worker safety representatives and generally to act as a resource group for the Operation. The *Co-Chairs* shall meet and prepare the agenda for each meeting at least fourteen (14) days in advance of the meeting. The Company will supply minutes of such meetings to the members of the Committee, the Union and the Local Union.

17.05 The Operation Health, Safety and Environment Committee established for each Operation shall consist of up to ten (10) members, up to five (5) to be appointed by the Local Union from among the employees in the Operation at least one (1) of whom shall be from either the Maintenance or Electrical Department and up to five (5) to be appointed by the Company from among management representatives familiar with the Operation, one (1) of whom shall be the *Manager* of the Operation, or his designate. In addition, the Local Union may appoint up to five (5) alternate members, any one of whom may replace a regular member of the Committee. The Committee shall meet at least once each month with additional meetings approved by both *Co-Chairs*, for the purposes of reviewing the on-going Health, Safety and

Environment Program at the Operation, exchanging and discussing health and safety information and considering specific health and safety problems at work locations within the Operation. The *Co-Chairs* shall meet at least five (5) days prior to each scheduled meeting in order to discuss the matters to be dealt with by the Committee at that meeting. The Company shall supply minutes of each such meeting to the members of the Committee and the Local Union.

Each *Co-Chair* of the Committee shall designate one (1) Committee member for the purpose of forming an inspection team. This inspection team shall conduct a monthly inspection within the Operation and report its findings at the next Committee meeting. The Company shall pay employee members of the Operation Health, Safety and Environment Committee at their applicable hourly rate plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable *Schedule 'N' Bonus*, for any time lost during their regular working hours, while attending Committee meetings or making inspections as part of the committee's Inspection Team and preparing for or attending any Committee function, as approved by the *Co-Chairs*.

All regular members of the Committee will receive basic certification and hazard-specific training as soon as practicable following appointment to the Committee, it being understood that this training is not a prerequisite for full participation in the Committee.

- 17.06 The Company will, upon request, arrange a change of shift for a regular member, or an alternate, of an Operation Health, Safety and Environment Committee in order to enable such regular member, or alternate, to attend a scheduled meeting of that Committee.
- 17.07 Where an employee, after he has commenced work in any day or shift, suffers either an industrial accident or a recurrence of the effects of a previous industrial accident, if, in the opinion of a duly qualified medical practitioner, such accident or recurrence prevents him from continuing at work he shall be paid at his applicable hourly rate plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable Schedule 'N' Bonus for the balance of the time he would have worked in such day or shift had such accident or recurrence not occurred.
- 17.08 In any case where subsequent to his last date of hiring the Company requires that an employee undergo a medical examination or obtain a medical certificate as a condition of his employment with respect to a job, the Company will, if any such examination takes place outside the employee's regular working hours, pay such employee at his applicable hourly rate for any time spent on such examination and, if applicable, mileage in accordance with current company policy.

- 17.09 An employee member of the appropriate Operation Health, Safety and Environment Committee and the Worker Safety Representative shall be notified of the following events as soon as possible in order that he may be accompanied to the site of this incident/accident: a fatality, an employee death at work, a critical injury, an accident that has resulted in the injured employee being admitted to hospital, or other serious condition or incident.
- 17.10 If an employee suffers a fatal accident while at work, or if the Company calls the police as a result of an employee having had a serious accident while at work, the Company shall notify the *Chairperson* of the Local Union Health, Safety and Environment Committee in order that he may designate two (2) employee members of the Local Union Inquest Committee who shall, as soon as possible, be accompanied on an inspection of the accident site and, at the same time, provided with all available information concerning the incident/accident. Each such employee shall be given a leave of absence of up to two (2) consecutive days, for the purpose of making such inspection and investigating the incident. Additional leave may be approved by the Manager of Health, Safety & Environment. If the investigation time occurs on a regularly scheduled working day for the employee, the employee shall be paid for that day on the basis of the standard number of hours which the employee otherwise would have worked at the employee's applicable hourly rate of pay, plus any applicable shift premium, any applicable Cost of Living Allowance, and any applicable Schedule 'N' Bonus and any applicable incentive earnings.

If a Coroner's inquest is held with respect to such fatality an employee who is subpoenaed by the Coroner to testify at the inquest will be given a leave of absence of one day for each day the employee is required to attend at the inquest. Where any such day occurs on a regularly scheduled working day for the employee, the employee shall be paid on the basis of the standard number of hours which the employee otherwise would have worked at the employee's applicable hourly rate of pay plus any applicable shift premium, any applicable Cost of Living Allowance, and any applicable **Schedule 'N' Bonus**, and any applicable incentive earnings. If the Coroner's inquest results in recommendations being issued, the Company will meet within a reasonable time following their issuance with the two (2) designated employees who investigated the fatality for the purpose of discussing such recommendations. Where such meeting occurs on a regularly scheduled working day for the employee, the employee shall be paid on the same basis as employees subpoenaed by the Coroner to testify at the inquest.

17.11 If an Inspector or Engineer of the Mining Health and Safety Branch of the Ministry of Labour is making an inspection in a work location, the Local Union may designate a member of the appropriate Operation Health, Safety and Environment Committee for the purpose of accompanying the Inspector

or Engineer on his inspection. Such member shall be paid for the time spent while on the inspection on the basis of his applicable hourly rate plus any applicable shift premium and any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus**.

17.12 The Company and the Union agree to establish a system of worker safety representatives having the duties and responsibilities set out in Schedule 'J' of this Agreement. Each worker safety representative will be appointed as the Certified Member and *Co-Chair* for each Operation Health, Safety and Environment Committee in his Work Location as set out in Schedule 'J'. Worker safety representatives will, notwithstanding Article 11 of the Collective Bargaining Agreement, be appointed, removed and replaced by the Local Union from among the employees working in the Work Locations set out in Schedule 'J'. Upon completion of his appointment he will return to his regular job which will have been posted as a limited job for the period of his absence.

Worker safety representatives will receive such training as is required for the purpose of certification under the Occupational Health and Safety Act and training and on-the-job experience in relation to internal responsibility, the Company Loss Control Program, the Neil George Safety System, the Standard St. John Ambulance certificate, the Health, Safety and Environment Committees, the Occupational Health and Safety Act, the Occupational Exposure Monitoring Program and such other training as may be agreed upon.

Worker safety representatives will, notwithstanding Article 27 of the Collective Bargaining agreement, be paid at the equivalent of the wage rate for the Maintenance Electrician or at the rate of his regular occupational classification, whichever is greater, plus any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus**, and will be supervised by the Local Union and the manager, or his designated superintendent, responsible for part or all of his Work Location.

If during normal operations a worker safety representative is unable to act and the Company and the Local Union have reasonable notice that he will be absent for at least three (3) consecutive working days, he will be replaced by another worker safety representative on the first day of absence. Failing such notice, replacement will occur after the worker safety representative has been absent for three (3) consecutive working days.

17.13 The parties recognize April 28th as the annual day of remembrance for workers who died on the job. The Company agrees to lower to half-mast all flags flown at the workplace on that day. Allowance will be made for those working on that day to have a minute of silence in memory of those who died on the job.

The Company shall release all regular members of the OHSE Committee to attend the ceremonies on this date and will pay for such employees who were regularly scheduled to work on this date at their applicable hourly rate plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable Schedule 'N' Bonus, for any time lost during their regular working hours while attending the ceremonies.

ARTICLE 18 - BULLETIN BOARDS

- 18.01 No bills, bulletins, newspapers, handbills or other documents shall be posted or distributed in or about the premises of the Company by the Union, Local Union or by an employee except as hereinafter provided.
- 18.02 The Local Union may provide dispensing boxes for the Local Union publication 'The Searcher' and the Union publication 'Steel Labor' located at such places agreed to with the Plant or Department *Manager*.

The Company at the request of the Local Union shall post on the bulletin boards hereinafter provided for notices of:

- a) the time, place and type of meetings of the Local Union, and of Union conventions, including if desired the names of speakers at such meetings and conventions;
- b) the names of candidates for nomination or election as regular officers of the Union, the Local Union or as delegates to Union conventions, and the names of the offices for which they are candidates and other matters relevant to proceedings at Union or Local Union elections and conventions including the results of such nominations or elections and the names of the officers and delegates;
- the names of General Grievance Committeemen, Grievance Stewards, Plant Committeemen, and other Committeemen provided for under this Agreement;
- d) recreational and social events sponsored by the Local Union.
- 18.03 The Company on receipt of the necessary information from the Local Union shall from time to time post on the bulletin board space for each Plant the names of the Grievance Stewards designated by the Local Union to handle matters under the Grievance Procedure in such Plant, and shall also post on the same bulletin board space lists indicating the supervisory personnel designated by the Company to handle matters under the Grievance

Procedure in such Plant at each of the first two stages of the grievance procedure.

18.04 For the purpose of the above described notices, the Company will provide general bulletin boards or bulletin board space conveniently located for the purpose of bringing such notices to the attention of employees in each Department.

ARTICLE 19 - MISCELLANEOUS

19.01 Nothing in this Agreement contained shall be construed to bind the Company to comply with any of the provisions of this Agreement when such provisions may be impractical having regard to any law which shall be binding upon the Company.

ARTICLE 20 - TRADES AND STATIONARY ENGINEERS

A. Trades

20.01 The trades in the Company's operations in the Sudbury District are at the date of this Agreement:

Carpenter Machinist

Combustion Serviceman Maintenance Electrician

Heavy Duty Equipment Mason
-Technician Plateworker

MillwrightPower Protection &InstrumentmanControl Serviceman

Power Line Technician Welder Fitter

Power Utilityman Winder

and in the said trades there are, or may be, Leaders, *Tradespersons*, Intermediate *Tradespersons* and *Tradespersons* Helpers.

- 20.02 It is the Company's intention to provide opportunity for intermediate *tradespersons* to improve their qualifications in order that they may have the opportunity to advance in their respective trades. It is also the Company's intention to provide additional opportunity for helpers in each trade for which the Company has not established an apprenticeship program to attain the necessary qualifications to become *tradespersons*.
- 20.03 Therefore the Company will provide for helpers in non-apprenticed trades and intermediate *tradespersons* up to the number which are required to be trained in any trade at any time training programs consisting of 'on the job'

training combined with appropriate study courses.

- 20.04 Any intermediate *tradesperson* or any helper in a non-apprenticed trade who desires to participate in such a training program may make application to register as a trainee with the Company's Training Supervisor. Such applicants as may be selected will be informed as to the appropriate study course or courses that they must take, the nature of any such course or courses being dependent upon any prior training which may have been taken by the successful applicant. In selecting applicants the Company will give preference to their lengths of continuous service provided they have sufficient aptitude, knowledge or education to take the training.
- 20.05 The Company will upon the successful completion of any such approved study course by an employee reimburse such employee for the cost of such course for all courses in his training program, provided that at the time of completion of the course, the employee shall be in the employ of the Company, unless he shall at such time have been laid off by the Company for lack of work.

20.06 B. Apprentices

Apprentices now or hereafter in the Company's employ will be trained and upgraded in the manner heretofore customary in the Company's service but as may be modified by the provisions of this Article. They will be given an opportunity for training in as many branches of their trade as the Company's facilities will reasonably permit. At no time will the ratio of apprentices to *tradespersons* in any trade exceed the ratio of one (1) to three (3). It is understood that age alone will not prevent entrance into the apprenticeship training program.

- 20.07 No apprentice, during his apprenticeship, may apply in writing under the job posting provisions of this Agreement to fill any job vacancy notwithstanding the provisions of Article 11 hereof.
- 20.08 Apprentices will be required to take and successfully complete such study courses during their apprenticeship as are appropriate to assist them to qualify in their respective trades, and the Company agrees that upon the successful completion of each such course it will reimburse the apprentice for the cost of such course provided:
 - that the taking of each course by an apprentice shall, prior to the commencement of such course and during the currency of this Agreement, have been approved in writing by the Supervisor of Apprentices; and
 - 2) that such apprentice shall be in the employ of the Company at the time of

successful completion of such courses unless he shall at such time have been laid off by the Company for lack of work.

20.09 The trades in which apprenticeship training may be provided are as follows:

Carpenter

Heavy Duty Equipment Technician

Millwright
Instrument and Control Technician
Power Line Technician

Machinist
Maintenance Electrician

Mason
Power Protection & Control Serviceman

Welder Fitter

Winder

and such other trades as may be designated from time to time by the Company. In the case of any additional trade being so designated, the Company shall inform the Local Union in writing and discuss it with the Trades Committee prior to implementation.

20.10 The length of apprenticeship shall be up to four (4) years depending upon the particular trade and as may be determined by the Company. The Company may reduce the regular period of apprenticeship training where it is satisfied that an apprentice has received equivalent training or experience in his trade outside the Company's apprenticeship program.

Upon the successful completion of the period of apprenticeship training an apprentice will, subject to the requirements of operations, be employed as a tradesperson in the trade for which the employee has apprenticed and for this purpose the job posting provisions of this Agreement shall be applied in the following manner. Within one month of the commencement of the final period of his apprenticeship training, the Company will post a tradesperson job in the trade in which the apprentice will graduate with the name of the apprentice on the posting. Any applicant for the job must be qualified as a tradesperson in such job and employed as a tradesperson in the trade line for such job and cannot be employed in the plant under Schedule 'P' for which the vacancy is posted and assigned to the same shift, including daytime work, as the posted job. Notwithstanding the provisions of article 20.07, the apprentice who will graduate, may apply to the vacancy posted for the purpose of this Section, 20.10. The graduating apprentice will be assigned to the job vacated by the successful applicant if they are both in the same classification.

If not in the same classification, the vacated job will be posted and the

graduate apprentice will be assigned to the new vacated job if he and the successful applicant for the first vacated job are in the same classification. This process will continue until the graduate apprentice can be assigned to a vacated job.

- 20.11 If the Company establishes an apprenticeship program for trades not currently apprenticed the employees in the helper classification of the trade concerned will, if they make application for apprenticeship training in such trade, at the time of establishment of the program and up to the number which are required to be trained in such trade, be given preference in the order of their departmental seniority ranking to enter such apprenticeship program provided they possess sufficient knowledge, ability and skill to justify their being accepted for training. In the event of a force reduction apprentices will be subject to the provisions of Section 11.02 to the same extent as other employees.
- 20.12 An employee who is transferred to an apprenticeship shall receive the rate of the occupational classification from which he was transferred, up to a maximum of Job Class 13, or the rate payable under Schedule 'G', whichever is greater.
 - C. Stationary Engineers, Refrigeration Operators and Compressor Operators
- 20.13 It is the Company's intention to provide opportunity for employees in the classifications of Stationary Engineer, Refrigeration Operator and Compressor Operator and others who are required by law to have Government certificates, to improve their qualifications in order that they may have the opportunity to advance in their respective classifications.
- 20.14 Therefore the Company will provide training programs consisting of 'on the job' training combined with appropriate study courses for up to the number of such employees which it requires to be trained at any time.
- 20.15 Any employee in the classifications referred to in Section 20.13 hereof who desires to participate in such a training program may make application to register as a trainee with the Company's Training Supervisor. Such applicants as may be selected will be informed as to the appropriate study course or courses that they must take, the nature of any such course or courses being dependent upon any prior training which may have been taken by the successful applicant. In selecting applicants the Company will give preference to their lengths of continuous service provided they have sufficient aptitude, knowledge or education to take the training.
- 20.16 The Company will upon the successful completion of any Company approved study course by an employee reimburse such employee for the

cost of such course provided that at the time of completion of the course the employee shall be in the employ of the Company, unless he shall at such time have been laid off by the Company for lack of work.

D. Incentive for Study Courses

20.17 It is the Company's intention to further assist those employees referred to in this Article in the completion of any study courses (including any course the satisfactory completion of which would be required to qualify an employee for a Government certificate) approved under the provisions of this Article, and with this in view the Company will pay to each employee who successfully completes such a study course an amount equal to the employee's applicable hourly rate for one-half (1/2) of the standard hours specified for the completion of such study course. The determination of the standard hours specified for the completion of such study course shall be made by the Company and shall be final following consultation by the Company with the school or other agency supplying such course.

ARTICLE 21 - GENERAL TRAINING

21.01 A joint training committee will be established consisting of eight (8) members, four (4) to be appointed by the Local Union from among the employees of the Company, one (1) of whom shall be from the Mining and Milling Section, one (1) from the Smelting and Refining Section and one (1) from either the Maintenance or Electrical Department, and four (4) to be appointed by the Company from among Management Representatives, one (1) of whom shall be from the Mining and Milling Section, one (1) from the Smelting and Refining Section and one (1) from the Maintenance Department. The Local Union and the Company shall each designate a member of the committee as Co-Chair. The committee shall meet at least once, but not more than twice each month for the purposes of exchanging and discussing information pertaining to training, discussing the occupational classifications for which the Company has created trainee jobs, reviewing Company training programs, and considering specific problems arising out of these programs. The committee may make recommendations to the Company concerning matters discussed. The Co-Chairs shall meet at least five (5) days prior to each scheduled meeting in order to discuss the matters to be dealt with by the committee at that meeting. The Company will supply minutes of each such meeting to the members of the committee. committee member who is unable to attend a meeting owing to sickness, vacation or leave of absence may be replaced for that meeting by an alternate from the Section or Department of the regular member. The Company shall pay employee members of the committee at their applicable hourly rate plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable Schedule 'N' Bonus for any time lost during their regular working hours while attending meetings of the committee.

NOTE: Sections 21.02, 21.03 and 21.04 are suspended, in accordance with Letter of Agreement, "Trial Process for Awarding Unlimited Jobs".

- 21.02 The Company desires to provide an opportunity for employees to qualify for certain occupational classifications by providing training during working hours. The jobs in which employees may be employed for these purposes are hereinafter called 'trainee jobs'.
- 21.03 Whenever the Company decides that additional employees are required to be trained in such occupational classifications it shall post vacancies for trainee jobs as limited jobs as hereinbefore in this Agreement defined. If the successful applicant for a posted trainee job is from outside the working group where temporary relief will be required, the Company may assign an additional employee from within such working group to such trainee job without posting such job. In making such assignment, the Company will, to the extent permitted by the requirements of operations, give preference according to the seniority ranking of the employees in such working group who, in its judgement, possess sufficient knowledge, ability and skill to justify acceptance for training.

No employee shall be eligible to apply for a trainee job: a) who in the judgement of the Company is qualified in such occupational classification: b) who has been successfully trained as a result of a previous training posting within the previous six (6) month period unless waived by the Company; c) who within the six (6) month period prior to his application has failed to qualify in the same trainee job for which he is currently applying: d) who does not possess sufficient knowledge, ability and skill to justify his being accepted for training: e) who within the six (6) month period prior to his application has voluntarily withdrawn from a trainee job before its completion, unless waived by the Company. The employees required to fill the vacancies for trainee jobs will be determined in accordance with their respective seniority ranking. In the case where the vacancy is a Surface Department vacancy

resulting from an employee leaving the work force and the Company intends to add an employee to the Department, the trainee jobs resulting from the filling of that vacancy will be posted as a

Trainee/Full-time Unlimited job and filled in accordance with the provisions of this Section. The successful applicant for the Trainee/Full-time Unlimited job will, upon successful completion of the training required for the specific full-time unlimited job, be assigned to the job. For the purpose of this Section, "Surface Department" means the following Departments: Concentrators, Mills and Crushing, Copper Cliff Smelter, Nickel Refinery, Canadian Alloys/Oxide, Copper Refinery, and Transportation.

- 21.04 A successful applicant for a trainee job shall be paid during the training period at the rate of his regular occupational classification.
- In the case of Double Drum Hoistmen, Haulage Truck Driver, Electric Loco Engineer and General Brakeman a successful applicant for a trainee job shall be paid at the rate of the occupational classification for which he is being trained or his regular occupational classification whichever is greater.
- 21.05 Any employee who desires to take a home study course shall prior to commencing such course obtain written approval from the Company's Training Supervisor who shall advise the applicant whether or not the proposed course is a work related course and is acceptable to the Company for that individual under the provisions of this Article.
- 21.06 The Company will upon the successful completion of a work related study course, accepted under the provision of Section 21.05, by an employee reimburse such employee for the cost of such course provided that at the time of completion of the course the employee shall be in the employ of the Company, unless he shall at such time have been laid off by the Company for lack of work.

ARTICLE 22 - INCENTIVES

22.01 A joint incentive committee will be established consisting of ten (10) members, *five* (5) to be appointed by the Local Union from among the employees of the Company with one (1) representative from each of the *five* (5) mine Departments active on the date of this Agreement, and *five* (5) to be appointed by the Company from among Management Representatives. If the Company declares additional mines to be active during the life of this agreement, the Local Union and the Company shall each appoint one (1) more member in the same manner as above for each additional mine. The Local Union and the Company shall each designate a member of the committee as Co-Chair. The Committee will be provided with written notice of new incentive standards, changes to established incentive standards or extensions of the present program, prior to their

implementation. The committee shall meet once each month for the purpose of exchanging and discussing information pertaining to incentives and reviewing the Company incentive program including methods employed to develop work standards, factors affecting the application of the program, alterations in work standards, the introduction of new standards or the extension of the present program. The committee may make recommendations to the Company concerning matters discussed. Committee members and the designated alternates will be provided with instruction in the calculation and application of incentive rates. The Co-Chairs shall at least five (5) days prior to each scheduled meeting discuss the matters to be dealt with by the committee at that meeting. The Company will supply minutes of each such meeting to the members of the committee. Each party will designate five (5) alternate members to the Committee and may appoint such an alternate to replace a regular Committee member who is unable to attend a meeting owing to sickness, vacation or leave of absence. The Company shall pay employee members of the committee and designated alternates at their applicable hourly rate, plus any applicable shift premium, any applicable Cost of Living Allowance and any applicable Schedule 'N' Bonus, for any time lost during their regular working hours while attending meetings of the committee.

- 22.02 The Local Union may establish for each of the Plants in which an incentive system has been introduced and such other Plants in which an incentive system may be introduced a Plant Incentive Committee to consist in each case of not more than five (5) employees in the Plant in question. Committee members will be provided with instruction in the calculation and application of incentive rates.
- 22.03 Each such Plant Incentive Committee shall be entitled at reasonable times but not more often than once per month, to present to the *Manager* of its Department and/or such other officials of the Company as the said *Manager* may desire to be present, any complaints or objections with regard to any incentive applied in its Plant during the preceding month and any matters pertaining to a particular contract on which an individual or crew may be working currently or during the preceding month. The Plant Incentive Committee shall notify the *Manager* of the Department concerned at least one week prior to any such meeting of the particular contracts to be considered there at. The Company will on request furnish the Local Union and the Plant Incentive Committee with all reasonable information as to the factors involved in the calculation of any incentive. The Company will, within seven (7) days, give the Local Union and the Plant Incentive Committee its written response to the complaints or objections considered at the meeting.
- 22.04 If any such complaint or objection is not disposed of to the satisfaction of the Plant Incentive Committee, it shall be entitled, on application in writing to the Manager Human Resources, specifying in reasonable detail the nature of

the complaint or objection, to present any such complaint or objection at a meeting with a Company Committee. Such application in writing shall be made at least three (3) working days prior to the meeting with the Company Committee and shall be made within a reasonable time after the meeting with the *Manager* of the Department referred to in Section 22.03. The Company Committee will meet each Plant Incentive Committee as may be required for the dispatch of business but not more often than once in each month to consider complaints or objections of which notice has been given as aforesaid by the Plant Incentive Committee prior to such meeting. The date, time and place of such meeting shall be fixed by the *Manager Human Resources* and notice thereof shall be given to such Plant Incentive Committee. At each such meeting the Plant Incentive Committee may be accompanied, if it so desires, by a representative of the Union and/or an officer of the Local Union or a Local Union member of the Joint Incentive Committee.

- 22.05 Employees working on a particular contract to be considered may be represented at any meeting provided for in this Article by not more than one (1) of their number from each shift working on the contract. The provisions of Article 7 of this Agreement with respect to time off for Grievance Stewards and payment for time lost by them shall apply equally to the members of the Plant Incentive Committee and the representatives of the employees herein mentioned.
- 22.06 An employee may initiate a grievance under Article 7 where he claims that his pay has been incorrectly calculated or the number of units of work has been wrongly determined.
- 22.07 The Company's incentive system will be based on the present modified standard hour plan under which bonus earnings are expressed as a percentage of the employee's applicable hourly rate so that the incentive earnings automatically increase with increases in the wage structure.

ARTICLE 23 - PENSIONS

- 23.01 The employees covered by this Agreement will continue to receive the benefits of the Pension Plan for Hourly Paid Employees of the Company (Ontario Division, Sudbury District), Schedule 'I' hereto.
- 23.02 Should any difference arise between the Company and any of the employees or between the Company and the Union covering the interpretation, application, administration or alleged violation of the provisions of the Pension Plan, such difference shall be settled in accordance with the Grievance Procedure or arbitration provisions of this Agreement.

23.03 The designated beneficiary of a pensioner who retired before June 1, 1997, will, upon the death of the pensioner, be entitled to a benefit of five hundred dollars (\$500). This benefit will be five thousand and five hundred dollars (\$5,500) in the case of the death of a pensioner who retired after June 1, 1997.

ARTICLE 24 - INSURANCES, HEALTH AND HOSPITAL SERVICES

- 24.01 The employees covered by this Agreement shall receive the benefits of a Group Term Life Insurance Plan, a Group Non-Occupational Accidental Death and Dismemberment Insurance Plan, a Group Non-Occupational Sickness and Accident Insurance Plan as reflected in Sections 24.04 and 24.07, a Group Plan for Prescription Drugs (providing coverage equivalent to the Group Health plan for prescription drugs - Formulary 2 - 35 cents deductible), a Group Plan for Semi-Private Hospitalization (providing coverage equivalent to the Group Health Supplementary (semi-private) Plan), a Group Dental Plan (providing coverage equivalent to the Group Health Dental Plan - Preventative Basic Dental Care - plus coverage for endodontics and periodontics and for orthodontics with 50% of fees to a maximum of \$1,500) and a Group Optical Insurance Plan which will provide up to two hundred and fifty dollars (\$250) coverage in each three (3) year period for each employee, the spouse of the employee and all unmarried, unemployed children up to the age of 21 or up to 23 if an unmarried full-time student. All Plans are subject to and in accordance with the terms and conditions as set out in this Article and in the Plans, all of which, except the Group Non-Occupational Sickness and Accident Insurance Plan, form part of this Agreement.
- 24.02 The Company will provide the Manulife Dental Plan #7 through a Group Dental Insurance Plan, preventative basic dental care plus coverage for endodontics, periodontics and for dental coverage of orthodontics (50% of fees, maximum \$1,500) with the 2009 Ontario Dental Association Schedule of Fees for Dental Services provided by General Practitioners. Effective January 1, 2011, the applicable fee schedule will be the 2010 Ontario Dental Association Schedule of Fees, for Dental Services provided by General Practitioners and effective January 1, 2012, the applicable fee schedule will be the 2011 Ontario Dental Association Schedule of Fees for Dental Services provided by General Practitioners and effective **January 1**. 2013, the applicable fee schedule will be the 2012 Ontario Dental Association Schedule of Fees for Dental Services provided by General Practitioner and effective January 1, 2014, the applicable fee schedule will be the 2013 Ontario Dental Association Schedule of Fees for Dental Services provided by General Practitioner. Effective January 1, 2015, through to May 31, 2015 the applicable fee schedule will be the 2014

- Ontario Dental Association Schedule of Fees, for Dental Services provided by General Practitioners. Upon completion of any necessary enrolment forms, employees will be covered by the Dental Plan on the first day of the month following the month in which the employee has acquired seniority.
- 24.03 During the term of this Agreement the Company agrees to pay the premiums payable in respect of all employees under the above plans.
- 24.04 The Group Non-Occupational Sickness and Accident Insurance Plan will provide for a weekly benefit of \$625 effective June 1, 2006.
- 24.05 The Group Term Life Insurance Plan for persons in the active employment of the Company will be \$55,000 with a maximum coverage of \$5,000 for employees who retire on a disability pension with 3 or more years of continuous service as provided in the Plan. The Group Non-Occupational Accidental Death and Dismemberment Insurance Plan for persons in the active employment of the Company will be \$40,000.
- 24.06 Under the Group Optical Plan, effective the first day of the month following the month in which the Collective Bargaining Agreement commences, the Company will pay the premiums for a two hundred and fifty dollars (\$250) coverage in each three (3) year period for each employee, the spouse of the employee and all unmarried, unemployed children up to the age of 21 or up to 23 if an unmarried full time student.
- 24.07 The long term disability protection plan for all employees who have completed one year of service will provide that employees who are disabled within the meaning of the Pension Plan for Hourly Paid Employees will be placed directly on disability pension under the Pension Plan upon completion of a maximum of 104 weeks of Group Non-Occupational Sickness and Accident coverage for any one continuous period of disability.
- 24.08 In consideration of improved employee benefits paid for by the Company, the Union for the employees releases the Company from any obligation it might hereafter have to pay to employees any Employment Insurance rebate available because of the existence of a wage loss replacement plan. (Group Non-Occupational Accident and Sickness Insurance Plan)
- 24.09 Effective July 1, 2003, the Company will pay the premiums for an emergency out- of-country insurance plan for employees in the active employment of the Company. This plan covers the fees of a physician or surgeon for emergency treatment, but not hospital charges, while travelling outside of Canada to a lifetime maximum of \$5,000 per person.

ARTICLE 25 - COST OF LIVING ALLOWANCE

- 25.01 A cost of living allowance will, if applicable, be paid to each employee as set out below. This allowance will be based on the Consumer Price Index (all items base: 1961 = 100) published by Statistics Canada (hereinafter referred to as the "CPI") and will be calculated as follows:
 - a) The CPI published for April, **2010**, shall be compared with the CPI published for July, **2010**, and effective the pay period immediately following the publication of the July, **2010** CPI, the allowance shall be one (1) cent per hour worked for each zero point three five (0.35) point increase by which the July, **2010** CPI exceeds the April, **2010** CPI.
 - b) Such allowance, if any, shall continue until the publication of the CPI for October, **2010**, at which time the October, **2010** CPI shall be compared with the CPI published for July, **2010**, and effective the pay period immediately following the publication of the October, **2010** CPI, the allowance shall be adjusted by one (1) cent per hour worked for each zero point three five (0.35) point increase by which the October, **2010** CPI exceeds the July, **2010** CPI.
 - c) A similar comparison and adjustment shall be made thereafter on the basis of the CPI published every three (3) months apart as follows:

FOLLOWING THE BASED ON THE RELEASE OF COMPARISON OF: THE CPI FOR:

January, **2011** January, **2011** with October, **2010** April **2011** April, **2011** with January, **2011** July, **2011** July, **2011** with April, **2011** October, 2011 October, 2011 with July, 2011 January, **2012** January, **2012** with October, **2011** April, **2012** April, 2012 with January, 2012 July, **2012** July, **2012** with April, **2012** October, 2012 October, 2012 with July, 2012 January, 2013 January, 2013 with October, 2012 April **2013** April, 2013 with January, 2013 July, **2013** July, **2013** with April, **2013** October, 2013 October, 2013 with July, 2013 January, **2014** January, **2014** with October, **2013** April, 2014 with January, 2014 April, **2014** July, **2014** July, 2014 with April, 2014 October, 2014 October, 2014 with July, 2014 January, 2015 January, 2015 with October, 2014 April **2015** April, 2015 with January, 2015

- d) If there is a decrease in the CPI on the basis of the quarter to quarter comparison, the allowance shall be adjusted downward by using the formula mentioned above but an employee's applicable hourly rate shall not be affected by any downward adjustment.
- e) The amount of the cost of living allowance in effect at any time shall not be part of an employee's applicable hourly rate but will be included for the calculation of vacation pay under the Collective Bargaining Agreement.
- f) No adjustment retroactive or otherwise shall be made due to any revision which may later be made in any Consumer Price Index published by Statistics Canada.
- g) The continuance of the cost of living allowance shall depend upon the availability of the CPI calculated on its present basis and in its present form. Should this occur, the parties will meet and agree upon an appropriate alternative conversion of the CPI.

ARTICLE 26 - SUPPLEMENTARY UNEMPLOYMENT BENEFITS PLAN (SUB)

26.01 The employees covered by this Agreement will receive the benefits of the Supplementary Unemployment Benefits Plan (SUB) for Hourly Paid Employees of the Company in the Sudbury District Schedule 'L' hereto.

ARTICLE 27 - CO-OPERATIVE WAGE STUDY

- 27.01 The Co-operative Wage Study Manual (herein referred to as 'the Manual'), dated November 1, 1981, shall be incorporated into this Agreement as Schedule 'M' and its provisions shall thereafter apply as if set forth in full herein.
- 27.02 Each employee's job shall be described and classified and a rate of pay applied to it in accordance with the provisions of the Manual and this Agreement.
- 27.03 The scale of rates beginning at Job Class 1, and progressing upwards by equal increments shall be known as the Standard Hourly Wage Scale and will be Schedule 'F' to this Agreement.
- 27.04 Effective June 1, 1983 all employees shall have their rates of pay adjusted as follows:

- a) If the employee is not receiving an out-of-line differential prior to that date, the rate of pay of such employee shall be adjusted to conform to the standard hourly rate for that employee's job, as provided in Schedule 'F'.
- b) If the employee is receiving an out-of-line differential prior to June 1, 1983, the rate of pay of such employee shall be increased by the amount by which the rate for Job Class 1 has been increased, as provided in Schedule 'F' and the following shall govern:
 - If the employee's new rate resulting from such increase is greater than the standard hourly rate for the job, as provided in Schedule 'F', the amount by which such employee's new rate is greater than the rate provided in Schedule 'F' shall become such employee's new outof-line differential (which shall replace the former out-of-line differential) and shall apply in accordance with the provisions of this Agreement.
 - 2. If the employee's new rate resulting from such increase is equal to or less than the standard hourly rate for the job, as provided in Schedule 'F', the rate of pay of such employee shall be adjusted to conform to the standard hourly rate for the job as provided in Schedule 'F' and the former out-of-line differential shall be terminated.
- 27.05 As of June 1, 1983 the standard hourly rate for each job class shall be the standard hourly rate for all jobs classified within such job class and shall so continue for the duration of the Standard Hourly Wage Scale and shall be applied to any employee in accordance with the provisions of this Agreement.
- 27.06 The Company and the Local Union shall agree on a list of employees who are to be paid out-of-line differentials. Such list shall contain the following information:
 - a) Name of employee to whom such out-of-line differential is to be paid,
 - b) Job title of job on which out-of-line differential is to be paid,
 - c) Job class of such job,
 - d) Wage rate of such job,
 - e) Amount of out-of-line differential and
 - f) Date such out-of-line differential became effective.
- 27.07 Except as such out-of-line differential may be changed by the means hereinafter provided, any employee included in the list referred to in Section 27.06 shall continue to be paid such out-of-line differential during such time as the employee continues to occupy the job for which the out-of-line

- differential was established.
- 27.08 If an employee with an out-of-line differential is required by the Company to either transfer or be assigned to a job having a higher hourly rate, then the differential shall be reduced by the amount of the increase in the hourly rate.
- 27.09 If, as a result of a force adjustment and/or the exercise of seniority rights, an employee with an out-of-line differential is moved to a job having a lower hourly rate, then the out-of-line differential shall be cancelled.
- 27.10 If such employee referred to in Sections 27.08 and 27.09 shall be returned to the job for which the out-of-line differential was established, the out-of-line differential shall be reinstated except as it may have been reduced or eliminated by other means.
- 27.11 When an employee would, in accordance with the terms of this Agreement, be entitled to receive his applicable hourly rate, he shall also receive any out-of-line differential to which he is entitled.
- 27.12 In addition to the means herein provided, increases in the increment between job classes shall be used to reduce or eliminate out-of-line differentials.
- 27.13 Any mathematical or clerical errors made in the preparation, establishment or application of job descriptions, classifications or standard hourly rates shall be corrected to conform to the provisions of this Agreement.
- 27.14 Except as otherwise provided in the Manual no basis shall exist for an employee to allege that a wage rate inequity exists.
- 27.15 There shall be a committee known as the 'CWS Committee' consisting of one (1) appointed by the Local Union to represent the Union and one (1)) individual appointed by the Company to represent the Company. Either party may change its representatives from time to time by written notification to the other.
- 27.16 Meetings of the CWS Committee will be held as frequently as required at mutually agreeable times.
- 27.17 Leaves of absence shall be granted to Local Union representative on the CWS Committee to the extent required for handling new or changed job descriptions and job classifications in accordance with the requirements of the Manual. Time lost by the Local Union representative on the CWS Committee shall be paid for by the Company at the equivalent of the wage rate for Maintenance Electrician, plus any applicable shift premium and Cost of Living Allowance and any applicable Schedule 'N' Bonus. The

- Company may at any time discontinue such payments if, in the opinion of the Company, there is abuse of the privilege.
- 27.18 Any arbitration under the terms of the Manual shall be conducted in accordance with the provisions of Article 8 of this Agreement.

ARTICLE 28 - SPECIAL SHORT TERM SHUTDOWN PROTECTION PLAN

- 28.01 The Company may, upon forty-five (45) days' written notice to the Union, declare a special shutdown of all or part of a Department for a period not to exceed four (4) months, which period may immediately precede or follow a vacation shutdown.
- 28.02 In the event such a shutdown is declared, the employees working in the Department or the part of the Department to be shut down will, notwithstanding Article 11 of the Collective Agreement, be placed on layoff for the period of the shutdown and covered by the provisions of the Special Short Term Shutdown Protection Plan described in Schedule 'O' hereto and hereinafter called the 'Plan'.
- 28.03 The Company may recall individuals during the period of the shutdown to perform work in the shutdown Department or shutdown part of the Department, as the case may be. For this purpose, the Company shall determine which individuals off work from the Department and covered by the provisions of the Plan are qualified to fill the normal requirements of the jobs available and shall give preference in recalling to work such individuals on the basis of their length of continuous service with the Company and the provisions of Section 11.05 (a) to (g) shall apply. An individual may decline such recall if the Company is able to recall another individual from the Department who is covered by the Plan and is qualified to fill the normal requirements of the job and has less seniority than the declining individual.
- 28.04 The Company will not declare a special shutdown of a particular part of a Department or the Department as a whole more than once during the term of the Collective Agreement.
- 28.05 The Company will meet with the President of the Local Union, a representative of the Union and one additional employee upon request and with reasonable notice for the purpose of discussing matters in connection with the Plan.

ARTICLE 29 - AUTHORITY

29.01 United Steelworkers and its officers agree that they have authority from the

members of the said Union to enter into this Agreement and agree that this Agreement shall be binding upon its members.

ARTICLE 30 - FURTHER CONFERENCES

- 30.01 Joint conferences between representatives of the Company and of the Union shall commence in Copper Cliff, Ontario, not less than sixty (60) days nor more than ninety (90) days before the termination date of this Agreement for the purpose of considering, and, if thought fit, negotiating an Agreement with regard to wages, hours and conditions of employment to take effect upon the expiration of this Agreement.
- 30.02 The In-Term Meetings between the Company, the Local Union and the Union to deal with problems arising from the administration of this Agreement and to improve communications between the parties will be continued. The In-Term Meetings will also provide an opportunity to discuss the implementation of new experimental programs designed to improve job satisfaction and productivity, and the application of this Agreement to such programs.

ARTICLE 31 - TERMINATION

31.01 This Agreement shall become effective on the 8th day of July 2010 and shall terminate at midnight on the 31st day of May 2015.

IN WITNESS WHEREOF *Vale Canada Limited* has caused this Agreement to be signed by its proper officers of *United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International <i>Union (United Steelworkers)* and the Bargaining Committee of Local 6500 have hereunto set their hands on behalf of the said Union and on their own behalf.

EXECUTED in the District of Sudbury, Ontario, this 8th day of July, 2010.

UNITED STEELWORKERS

Rick Bertrand
Pascal Boucher
Scott Duncan
John Fera
Wayne Fraser
Brent Laaksonen
Jason Patterson

Peter Saal Myles Sullivan VALE CANADA LIMITED

Harvey Beresford (Hicks Morley)

Helen Francis Heiko Leers Mitch Medina Teddy Rabski Kelly Strong Jennifer Witty

SCHEDULE 'A'

CLASSIFICATIONS OF EMPLOYEES EXCLUDED FROM THE BARGAINING UNIT

Mines Section

Mining Underground

Superintendents

Asst. Superintendents

Underground Superintendents

General Foremen

Senior Foremen

Junior Foremen

Surface Foremen

Relieving Surface Foremen

Supply Foremen

Drill Fitter Foremen

Salvage Foremen

Shift Bosses

*Rel. Shift Bosses

Drill Fitter Bosses

Matrons

Supply Bosses

Open Pits

Superintendents

Asst. Superintendents

Senior Foremen

Junior Foremen

Blasting Foremen

Asst. Blasting Foremen

Supply Foremen

Shift Bosses

*Rel. Shift Bosses

Drill Fitter Bosses

^{*} Including individuals in other classifications who automatically act for shift boss in the latter's absence.

Mechanical - All Mines

Master Mechanics of Mines

Asst. Master Mechanics of Mines

Mine Master Mechanics

Asst. Mine Master Mechanics

Garage Foremen

Asst. Garage Foremen

Rockhouse Foremen

Mechanical Shift Foremen

Steel Shop Foremen

Carpenter Shop Foremen

Garage Shift Foremen

Machine Shop Foremen

Plate Shop Foremen

Crushing Plant Foremen

Rockhouse Bosses

Steel Shop Bosses

Crushing Plant Bosses

Picker Bosses

Rigger Bosses

Reduction Section

Mill

Superintendents

Asst. Superintendents

General Foremen

Flotation Foremen

Shift Bosses

*Rel. Shift Bosses

Screening Plant Bosses

Grinding Bosses

Dewatering Bosses

Flotation Bosses

General Labour Bosses

^{*} Including individuals in other classifications who automatically act for shift boss in the latter's absence.

Bin Bosses Metallurgists Statisticians Matrons

Smelters

Supts. Reverb Supts. Convtrs. Supts. Orford

Supts. Coniston

Asst. to Supts.

General Foremen

Asst. Supts.

Shift Bosses

Smelter Foremen

Blast Furnace & Convtr. Slag Bosses (Coniston)

Transportation Foremen

*Rel. Shift Bosses

Converter Slag Bosses

Puncher Bosses

Lining St. Bosses

Sinter Pl. Bosses

Flue Dust Bosses

Cottrell Bosses

Bin Bosses

Slag Dump Bosses

Matrons

Watchmen

Copper Cliff and Coniston Mechanical

Master Mechanics

Asst. Master Mechanics

General Foremen

^{*} Including individuals in other classifications who automatically act for shift boss in the latter's absence.

Chief Engineers (Power)

Shop Foremen

Fitter Foremen

Mason Foremen

Asst. Fitter Foremen

Substation Foremen (Coniston)

Transportation

Supts.

Asst. Supts.

Yardmasters

Roadmasters

Transportation Assts.

Train Dispatchers

Truck Dispatchers

Electrical

Supts.

Asst. Supts.

Powerhouse Supts.

Chief Electricians

Asst. to Supts.

Asst. Chief Electricians

Asst. Powerhouse Supts.

Foremen

Shop Bosses (including Loco, Repair Shop and Winding Bosses)

Meter Foremen

P. Hse. & Pl. Sw. Bd. Operators

Water Regulators

Patrolmen

Copper Refining Section

Supts.

Asst. Supts.

Asst. to Supts.

Chief Engineers

Night Supervisors

General Foremen

Shift Foremen

Foremen, Tank House Aisle

Foremen, Acid Plant

Foremen, Selenium Plant

Foremen, Lead Burner

Foremen, Coal Plant

Foremen, Transportation Foremen, Watch & Patrol Shift Bosses, Electric Furnace Shift Bosses, Anode Furnace Shift Bosses, Tank House Shift Bosses, Silver Refinery Shift Bosses, Acid Plant Shops Foremen Chief Electricians Asst. Chief Electricians Chief Inspectors (Refined Shapes) Gold Room Operators Chief Sampler P.H. Switchboard Operators Matrons Watchmen

General Exclusions

All employees of the Geological Department, excepting Geological helpers. (M. & S. Division)

All employees of the Metallurgical Dept. excepting samplers, sampler helpers and labourers. (M. & S. Division)

All employees of the Safety and First Aid Dept. excepting drymen. (M. & S. Division)

All employees of the Purchasing and Warehousing Dept. excepting storemen and storemen helpers. (M. & S. Division)

All employees of the Research Dept. excepting Test Mill Operators and sampler helpers.

All employees of the Efficiency Dept. (M. & S. Division)

All employees of the Personnel and Employment Dept. (M. & S. and C.R. Divisions)

All employees of the Mines Engineering Dept. (M. & S. Division)

All employees of the Hospital and Clinical Depts.

All employees of the Accounting Dept. (M. & S. Division)

All employees of the Civil and Mechanical Engineering Dept. (M. & S. Division)

- All Real Estate Agents, Ventilation Engineers, Plant Clerks and Stenos, Agriculturists, Fire Prevention Men, Process Engineers, Watchmen (Bruce Mines), Janitors (Community Halls). (M. & S. Division)
- All employees of the Metallurgical, Chemical and Research Dept. except drillers and drill room men. (C.R. Division)
- All employees of the Accounting Dept. including plant clerks and janitors. (C.R. Division)
- All employees of the Safety and First Aid Dept. (C.R. Division)
- All employees of the Purchasing and Warehousing Dept. except stores helpers. (C.R. Division)
- Employees in any new classifications having supervisory or confidential status at least equal in degree to that of any employees herein excluded.

SCHEDULE 'B'

RULES GOVERNING PROCEEDINGS OF GRIEVANCE COMMISSIONER

- 1. The parties when referring a grievance to the Grievance Commissioner shall also provide him with *decisions of the Management Representative at Stages One and Two.*
- 2. The parties shall prepare a Statement of Facts Agreed and Not Agreed and supply it to the Grievance Commissioner. In addition, they shall supply the Grievance Commissioner and each other with brief written representations on which they intend to rely provided that such are mailed not less than ten days before the commencement of the hearings of the Grievance Commissioner.
- 3. The purpose of the hearing is to clarify the issues or facts in dispute. At the hearing the parties may make such further representations or adduce such evidence as the Grievance Commissioner may permit or require, but the Grievance Commissioner shall not be obligated to conform to the rules of evidence.
- 4. The Grievance Commissioner must render his decision in writing without reasons to both parties within seven (7) days of the conclusion of the hearings. Upon request by either party after his decision has been rendered the Grievance Commissioner shall deliver brief reasons, but such reasons shall not form part of his decision.

SCHEDULE 'C'

RULES OF ARBITRATION

- 1. Arbitrations shall be heard at Sudbury, Ontario, or at such other place as may be agreed upon by the parties from time to time.
- In any arbitration the written representations of the employee made at any stage and the decision of the Company at Stage Two of the Grievance Procedure (or in the case of a difference between the Union and the Company the written submission by the party initiating the discussion of the difference and the written reply thereto of the other party) shall be presented to the arbitrator and the award of the arbitrator shall be confined to determining the issues herein set out.
- 3. Each party to arbitration shall be entitled through counsel or otherwise to present evidence, to cross examine the witnesses of the other party, and to present oral arguments. A brief of argument may be presented by each party and each party shall be entitled to reply to the brief of argument presented by the other. Briefs of argument and replies thereto shall be filed within the times specified by the arbitrator. A party presenting a brief of argument or reply to the arbitrator shall forthwith deliver a copy thereof to the other party.
- 4. Witness fees and allowances shall be paid by the party calling the witness.
- 5. No cost of any arbitration shall be awarded to or against either party.

SCHEDULE 'D'

DEPARTMENTS

- 1. Frood-Stobie Mine (includes Little Stobie)
- 2. Creighton Mine
- 3. Garson Mine
- 4. Levack Mine (includes McCreedy West)
- 5. Lower Coleman Mine (includes McCreedy East)
- 6. Crean Hill Mine
- 7. Copper Cliff North Mine
- 8. Copper Cliff South Mine
- 9. Exploration Diamond Drilling
- 10. Concentrators, Mills and Crushing
- 11. Copper Cliff Smelter (formerly Furnaces, Converters, Matte Processing including Acid and SO₂ Plants)
- 12. Nickel Refinery (Nickel Products)
- 13. Canadian Alloys/Oxide
- 14. Copper Refinery
- 15. Transportation
- 16. Electrical
- 17. Maintenance
- 18. Purchasing and Warehousing
- 19. Mines Research
- 20. All Mines Development
- 21. All Mines Construction

SCHEDULE 'E'

CLASSIFICATIONS ENTITLED TO PREFERENTIAL RECALL

Carpenters

Combustion Serviceman

Heavy Duty Equipment Technicians

Instrumentmen

Power Line Technician

Power Utilityman

Machinists

Maintenance Electricians

Masons

Plateworker

Power Protection & Control Servicemen

Stationary Engineers

Welder Fitters

Winders

SCHEDULE 'F' WAGE RATES

Step	July 8, 2010 \$/HR	June 1, 2011 \$/HR	June 1, 2012 \$/HR	June 1, 2013 \$/HR	June 1, 2014 \$/HR
1	\$25.66	\$25.76	\$25.91	\$26.16	\$26.46
2	\$25.88	\$25.98	\$26.13	\$26.38	\$26.68
3	\$26.10	\$26.20	\$26.35	\$26.60	\$26.90
4	\$26.32	\$26.42	\$26.57	\$26.82	\$27.12
5	\$26.54	\$26.64	\$26.79	\$27.04	\$27.34
6	\$26.76	\$26.86	\$27.01	\$27.26	\$27.56
7	\$26.98	\$27.08	\$27.23	\$27.48	\$27.78
8	\$27.20	\$27.30	\$27.45	\$27.70	\$28.00
9	\$27.42	\$27.52	\$27.67	\$27.92	\$28.22
10	\$27.64	\$27.74	\$27.89	\$28.14	\$28.44
11	\$27.86	\$27.96	\$28.11	\$28.36	\$28.66
12	\$28.08	\$28.18	\$28.33	\$28.58	\$28.88
13	\$28.30	\$28.40	\$28.55	\$28.80	\$29.10
14	\$28.52	\$28.62	\$28.77	\$29.02	\$29.32
15	\$28.74	\$28.84	\$28.99	\$29.24	\$29.54
16	\$28.96	\$29.06	\$29.21	\$29.46	\$29.76
17	\$29.18	\$29.28	\$29.43	\$29.68	\$29.98
18	\$29.40	\$29.50	\$29.65	\$29.90	\$30.20
19	\$29.62	\$29.72	\$29.87	\$30.12	\$30.42
20	\$29.84	\$29.94	\$30.09	\$30.34	\$30.64
21	\$30.06	\$30.16	\$30.31	\$30.56	\$30.86
22	\$30.28	\$30.38	\$30.53	\$30.78	\$31.08
23	\$30.50	\$30.60	\$30.75	\$31.00	\$31.30
24	\$30.72	\$30.82	\$30.97	\$31.22	\$31.52
25	\$30.94	\$31.04	\$31.19	\$31.44	\$31.74
26	\$31.16	\$31.26	\$31.41	\$31.66	\$31.96
27	\$31.38	\$31.48	\$31.63	\$31.88	\$32.18
28	\$31.60	\$31.70	\$31.85	\$32.10	\$32.40
29	\$31.82	\$31.92	\$32.07	\$32.32	\$32.62
30	\$32.04	\$32.14	\$32.29	\$32.54	\$32.84
31	\$32.26	\$32.36	\$32.51	\$32.76	\$33.06
32	\$32.48	\$32.58	\$32.73	\$32.98	\$33.28
33	\$32.70	\$32.80	\$32.95	\$33.20	\$33.50
34	\$32.92	\$33.02	\$33.17	\$33.42	\$33.72
35	\$33.14	\$33.24	\$33.39	\$33.64	\$33.94
36	\$33.36	\$33.46	\$33.61	\$33.86	\$34.16
37	\$33.58	\$33.68	\$33.83	\$34.08	\$34.38

^{*}Wage rates effective July 8, 2010, June 1, 2011, June 1, 2012, June 1, 2013 and June 1, 2014 will be increased by any Cost of Living Allowance in effect on those Dates and the Cost of Living Allowance will be reduced by the same amount.

SCHEDULE 'G'

SCHEDULE OF APPRENTICESHIP TRAINING (1040 HOURS TRAINING PERIOD) CWS WAGE CLASS PAYABLE FOR APPRENTICESHIP PERIODS

4-Year Apprenticeship

1 st	2 nd	3 rd	4 th	5 th	6 th	7 th	8 th	Trade Job Class
3	5	7	9	12	15	17	19	21
3	5	7	9	11	14	16	18	20
3	5	7	9	11	13	15	17	19
3	5	7	8	10	12	14	16	18
3	5	7	8	9	11	13	15	17
3	5	7	8	9	10	12	14	16
3	5	7	8	9	10	11	13	15

3-year Apprenticeship

1 st	2 nd	3 rd	4 th	5 th	6 th	Trade Job Class
3	6	9	14	17	19	21
3	6	9	13	16	18	20
3	6	9	12	15	17	19
3	6	9	12	14	16	18
3	6	8	11	13	15	17
3	5	7	10	12	14	16
3	5	7	9	11	13	15
3	5	6	8	10	12	14
3	5	6	7	9	11	13
3	5	6	7	8	10	12

2-Year Apprenticeship

1 st	2 nd	3 rd	4 th	Trade Job Class
3	8	14	19	21
3	8	13	18	20
3	8	13	17	19
3	8	12	16	18
3	7	11	15	17
3	7	11	14	16

SCHEDULE 'H'

The '	working areas lis	sted b	elow	Number of Plant	
will b	e considered Pla	Committeemen			
purp	ose of the Collec	ctive A	greement	at each Plant	
				0	
1.	Garson			2	
2.	Frood			2	
3.	Stobie			3	
4.	Creighton Con	nplex		5	
5.	Lower Colema	an (ind	cludes McCreedy East)	3	
6.	C.C. North Mir	ne - ir			
	Open Pit & Re		3		
7.	Nairn-High Fa		1		
8.	C.C. South Mi		3		
9.	Nickel Refiner	y & S	4		
10.	Copper Refine	•	•		
11.	Central Mills	-	Clarabelle Mill	3	
		,	Copper Cliff Mill	3 3 2 3 3 2 2	
12.	Smelter		Furnaces	3	
		b)	Converters	3	
13.	Matte Process	,		2	
14.	Lower Yard	a)	Transportation	2	
		b)	Shops	2	
		c)	Purchasing & Warehousing	1	
15.	Divisional Sho	,		1	
16.	Construction			1	

If additional working areas are opened or present working areas closed or the number of employees in a working area substantially changes, the Company and the Local Union shall meet to determine the number of Committeemen required.

SCHEDULE 'J'

The following will be considered work locations for Worker Safety Representatives	The following will be considered Operations for the Purpose of Article 17, Health, Safety and Environment		
1. Frood –Stobie Complex, Garson,	Frood – Stobie Complex		
Coleman	2. Garson		
Three (3) Representatives	Coleman (including McCreedy East)		
2. Creighton	Creighton Mine		
One (1) Representative			
3. C.C. North Mine, <i>C.C. South Mine</i>	 Copper Cliff North Mine, C.C. South Mine 		
One (1) Representative	 Diamond Drilling and Raise Boring Reconditioning Shop 		
4. Mills Tailings	1. <i>Mills Tailings</i> (Clarabelle Mill,		
One (1) Representative	Tailings, Booster Station, <i>Filter Plant</i>)		
5. Matte Processing, Utilities and Power One (1) Representative	1. Matte Processing 2. Utilities (Acid, SO2, Utilities, Water Plants), and Power		
6. Copper Cliff Smelter Complex	1. Furnaces, Field Services, Copper,		
One (1) Representative	Convertors, Casting, Cottrells		
7. Nickel Refinery Complex and Electrowinning One (1) Representative	1. N.R.C., I.P.C. and Electrowinning		
8. Transportation, Warehousing and	1. Transportation		
Central Laboratory	2. Warehousing		
One (1) Representative	3. Central Laboratory, Tolls & Reverts		
9. Central Maintenance, Divisional	Central Maintenance, Divisional		
Shops, Maintenance Training,	Shops, Maintenance Training,		
Maintenance Services	Maintenance Services		
One (1) Representative			

- If additional working areas are opened or present working areas closed, or the number of employees in a working area substantially changes, the Company and the Local Union shall meet to determine the appropriate committees required and any revisions in the number of Worker Safety Representatives.
- The primary responsibility of Worker Safety Representatives is to inspect, audit and address workplace conditions and work practices in relation to health and safety of personnel in mines and plants. They also assist in the promotion and development of health, safety and environment practices and procedures.

- 3. The duties and responsibilities of Worker Safety Representatives include the following:
 - a) Conduct health and safety inspections and investigations, monitor conditions and practices and make recommendations to appropriate plant supervision.
 - b) Participate in the preparation of training materials and programs and in the promotion of those programs.
 - c) Carry out the responsibilities of the Certified Member of the **OHSE** Committees assigned to his Work Location under Schedule 'J'.
 - d) Serve as **Co-Chair** of the **OHSE** Committees assigned to his work location under Schedule 'J'.

SCHEDULE 'N'

EMPLOYEE BONUS PLAN

This will confirm our agreement with respect to the payment of a new Employee Bonus Plan ("EBP") to employees. Employees will be eligible to receive, in each year, EBP payments totalling up to twenty-five percent (25%) of the employee's "AIP period earnings" (as defined below). The Employee Bonus Plan has two (2) components: the "Quarterly Bonus Component"; and the "Annual Bonus Component" as set out below:

1) Quarterly Bonus Component

The Quarterly Bonus Component represents up to seventy-five percent (75%) of the bonus in each year. In the second pay period following the reporting of the quarterly results for the periods ending March 31st, June 30th, September 30th and December 31st, employees will be paid the greater of:

- a) the Nickel Price Bonus (NPB) (as described below); or
- b) the Earnings Based Compensation (EBC) (as described below);

to a maximum of seventy-five percent (75 %) of twenty-five percent (25 %) of the employee's applicable hourly rate on the last day of the quarter multiplied by the regular non-overtime hours worked by the employee and vacation hours in the quarter.

NPB

For the purpose of the foregoing, the NPB attributable to an employee shall be calculated in the following manner:

- a) The NPB will be based on the average realized price per pound of nickel, in U.S. dollars, including intermediates, published by the Company in its Quarterly Reports for periods ending March 31st, June 30th, September 30th and December 31st (hereinafter called the 'Average Realized Price').
- b) If the Average Realized Price for any Quarter in which the Company has net earnings is equal to the Trigger Price, the NPB for each employee will be equal to the number of hours worked by him in that Quarter multiplied by ten cents (10¢) Canadian. If the Average Realized Price exceeds the Trigger Price, the multiplier used to calculate the NPB will be increased by one cent (1¢) Canadian for

- each one cent (1ϕ) U.S. by which the Average Realized Price exceeds the Trigger Price.
- c) The Trigger Price as of July 8, 2010 is three dollars and seventy-five cents (\$US 3.75).

EBC

For the purpose of the foregoing, EBC attributable to an employee shall be calculated in the following manner:

- a) EBC will be based upon the Operating Earnings of the Ontario Operations of the Company published/reported in U.S. dollars by the Company quarterly (the "Operating Earnings") for the periods ending in March 31, June 30, September 30 and December 31 in each year.
- b) If the Ontario Operations of the Company has Net Pre-Tax Operating Earnings for the quarter that are positive and the payment of the Quarterly Bonus Component for the quarter will not put Net Pre-Tax Operating Earnings at zero or below zero, the EBC attributable to an employee is: for every \$10 million in Operating Earnings published/reported in a quarter, 25 cents for each hour worked by the employee, including regular vacation, in that quarter (the Ratio). If the quarterly Operating Earnings are greater or lesser than \$10 million, the amount attributable to the employee will be adjusted using the Ratio. By way of example, if the Operating Earnings for the quarter are \$53.5 million and the employee works 475 hours in the quarter, the EBC attributable to the employee for that quarter will be 475 x 5.35 x \$.25 = \$635.31 Canadian.

2) Annual Bonus Component

The annual bonus component represents up to twenty-five percent (25 %) of the Employee Bonus in each year and is the part of the Vale S.A. Annual Incentive Plan ("AIP"), in effect from time to time, that is based upon the Corporate Performance Factor.

- a) The Corporate Performance Factor is the rating announced by Vale S.A. that represents how it has performed during the AIP period.
- b) The Annual Incentive Plan period (AIP period) is January 1 to December 31.

- c) Regular full-time employees who have completed at least one full calendar month of employment in the AIP period are eligible to participate in the Annual Bonus Component.
- d) In the month following the publication by Vale S.A. of its annual Corporate Performance Factor, all eligible employees will receive their Annual Bonus Component calculated as follows:
 - i) the Corporate Performance Factor will be multiplied by twenty-five percent (25 %) of the employee's AIP period earnings as defined below. This number will in turn be multiplied by twenty-five percent (25 %), which represents the maximum Employee Bonus payable in any year under the Annual Bonus Component.
 - ii) the "AIP period earnings" means the employee's applicable hourly rate on the last day of the AIP period multiplied by the regular non-overtime hours worked by the employee and his/her vacation hours in the AIP period.

Subject to the approval of Vale Inco General Counsel, the Company will agree to the following addition:

3) Verification Process

The Company will meet with three (3) representatives of the Union within sixty (60) days of the completion of each calendar quarter and make a report on the financial results that are relevant to the calculation of bonus payments for the last completed quarter. For this purpose, the Union will sign a confidentiality agreement that is satisfactory to the Company.

SCHEDULE '0'

SPECIAL SHORT TERM SHUTDOWN PROTECTION PLAN

An employee who is covered by this Special Short Term Shutdown Protection Plan will, notwithstanding anything to the contrary contained in the Collective Agreement, receive the following special protection during the period of the shutdown:

- 1. He will continue to accumulate continuous service with the Company;
- 2. He will be considered to have worked his regular hours in his occupational classification for the purpose of calculating his Years of Employment With Pay under the provisions of the Pension Plan for Hourly Paid Employees of the Company;
- 3. He will be considered to have worked his regular hours in his occupational classification and to have continued to receive incentive pay, where applicable, based on his Weekly Incentive Pay for the purpose of calculating his entitlement of vacation pay and Special Vacation pay under Articles 15 and 16 of the Collective Agreement.
- 4. He will continue to receive all the insurance benefits described in Article 24 of the Collective Agreement.
- 5. He will be entitled to the benefits of the Special Short Term Shutdown Protection section of the Supplemental Unemployment Benefit Plan.

SCHEDULE 'P'

The working areas listed below will be considered Plants for the purpose of Section 11.09 (2) of the Collective Agreement.

- 1. Frood
- 2. Stobie
- 3. Little Stobie
- 4. Creighton Complex
- 5. Garson Mine & Sand Plant
- 6. Levack & McCreedy West
- 7. Lower Coleman (includes McCreedy East)
- 8. Crean Hill Mine
- 9. Copper Cliff North Mine
- 10. Copper Cliff South Mine
- 11. Clarabelle Mill
- 12. Furnaces and Converters
- 13. Matte Processing
- 14. Copper Cliff Mill (including Sample House)
- 15. Divisional Shops
- 16. Nickel Refinery Complex
- 17. Copper Cliff Copper Refinery
- 18. Power and Construction
- 19. Central Utilities

Together with such additional Plants as the Company may from time to time establish.

SCHEDULE 'Q'

- For the purpose of and applicable to employees working on ten (10) or twelve (12) hour shifts, the Collective Bargaining Agreement will be amended as follows:
 - (a) Section 9.02 and 11.06: The words ' (not to exceed forty (40) hours per week)' shall be replaced with ' (not to exceed the standard number of hours which he otherwise would have worked at his applicable hourly rate)'.
 - (b) Section 13.01 (b): The words 'thirty (30) minutes per shift' shall be replaced with 'two (2) thirty (30) minute periods per twelve (12) hour shift', or 'one (1) thirty (30) minute period per ten (10) hour shift.
 - (c) Section 13.05: The words 'eight (8) hours' shall be replaced with 'his regularly scheduled hours'.
 - (d) Section 13.06 (a): The words 'forty (40) hours at his applicable hourly rate in his work week' shall be replaced with 'the number of regularly scheduled hours required in his work week at his applicable rate'.
 - (e) Section 13.06 (b): The words 'eight (8) hours' shall be replaced with 'twelve (12) hours' for 12 hour shift schedules or 'ten (10) hours' for ten hour shift schedules.
 - (f) Section 13.13: Delete the words "(excepting those employees for whom the recognized holiday is their regular scheduled day off)".
 - (g) Section 13.14: The words in the first sentence beginning with 'An employee' and ending with 'will not be so paid.' shall be replaced with 'An employee who does not work on a recognized holiday which occurs on his regular day off or during his vacation will be paid eight (8) hours at what would have been his applicable hourly rate had he worked on that day. If the recognized holiday occurs on what would otherwise have been a work day for the employee he will be paid eight (8) hours or the number of hours he was scheduled to work on that day, whichever is the greater, at what would have been his applicable hourly rate had he worked on that day. Provided that, in each case, he will not be so paid:"
 - (h) Section 13.19 shall be deleted and replaced with 'An employee shall be paid an averaged shift premium of eighty (80¢) cents per hour for time worked by him during his night shift hours' on twelve (12) hour shift schedules. The averaged shift premium for ten (10) hour shift schedules will be calculated based upon the working and off shift hours designated for each such schedule.
 - (i) Section 14.06: The words 'eight (8) hours' shall be replaced with 'twelve

- (12) hours' for 12 hour shift schedules or 'ten (10) hours' for ten hour shift schedules.
- (j) Section 11.20: The words 'twenty (20) working days' will be replaced with 'fifteen (15) working days'.
- 2. An employee regularly working on a twelve (12) hour or ten (10) hour shift schedule who is elected or appointed to the position of *Chairperson* of the General Safety, Health and Environment Committee or Chief Steward will be deemed to be in a job with a regular five (5) day, eight (8) hour per day schedule.
- 3. For purposes of Section 13.13 of the CBA, employees working on twelve (12) hour or ten (10) hour schedules will be considered to be in a separate working group.
- 4. The implementation of a twelve (12) hour or ten (10) hour shift is subject to the parties making a joint application for such twelve (12) hour or ten (10) hour shift to the Director of Employment Standards and receiving her approval.
- 5. Either the Company or the Union may cancel any particular twelve (12) hour or ten (10) hour shift upon at least thirty (30) days' written notice to the other.
- 6. An employee who is temporarily assigned to a twelve (12) hour or ten (10) hour shift schedule will be offered such additional hours of non-overtime work as are necessary to insure that the employee's scheduled work averages forty (40) hours per week over the period of the assignment.

SCHEDULE 'U'

- (1) The company will introduce in October, 1991, a new comprehensive procedure to be known as The Contracting Out Information and Review System. The System shall have the following components:
 - a) Major Projects (Projects which are in excess of one million dollars)

As soon as reasonably possible after receiving final approval for funding a Major Project, the Company will notify the Local Union in writing of the Major Project using the Notification Format. The Local Union executive and the Committee will meet with the Company within seven (7) days of receiving notification to review the work content of the Major Project and the Company shall give consideration to any comments or suggestions proposed at such meeting for the performance of any contract segment of the work by bargaining unit employees.

b) Mid-size Projects (Projects which are in excess of ten thousand but not more than one million dollars)

The Company will provide the Committee every month with a list of the Mid-size Projects for which either final funding approval has been received or which have been contracted out in the last one month reporting period. The list shall show for each project the nature of the work, its expected duration and the approximate number of contractor In the case of Mid-size Projects for which funding workers involved. approval has been received, the Company will endeavour to provide the Committee with notice of such Mid-size Project as soon as reasonably possible after receiving the final approval. At the next Contracting Out Committee meeting, the Company and the Committee will review the work content of the Mid-size Projects and the Company shall give consideration to any comments or suggestions proposed at such meeting for the performance of any contract segment of the work by bargaining unit employees. In cases where the work has already been contracted, such consideration shall relate, to the extent it is reasonable and possible, to future contracting out of similar work. The parties shall also use this information to develop trend analyses of the contracting of specific types of work.

c) Small Contracts (Work contracts which are up to ten thousand dollars)

The Company will, to the extent the information is reasonably available, provide the Committee every month with a list of the Small Contracts which have been contracted out in the last one month reporting period in

order that the Committee can review the work content of the Small Contracts and develop trend analyses of the contracting of specific types of work.

- d) The Notification Format for Major Projects will contain the following information.
 - a) The location of the work;
 - b) The type of work;
 - c) A description of the work;
 - d) An outline of the skills and equipment involved;
 - e) An estimate of the duration of the work;
 - f) An outline of any anticipated use of bargaining unit employees;
 - g) A description of the factors affecting the work schedule;

(2) Trend Analyses and Base Load

Trend analyses of the Small Contracts referred to above will be performed on a Division-wide basis by trade/skill taking into consideration a variety of factors including, without limiting the generality of the foregoing, the location of the work, its type, duration and frequency, the skills and equipment involved and any operating constraints associated with the work. A principal purpose of this trend analysis of Small Contracts is to assist the parties in agreeing on an appropriate definition or definitions of Base Load in the type of work normally performed by bargaining unit employees. Once agreement is reached and a base load is identified, the Company shall ensure an adjustment in the amount of such work performed by bargaining unit employees.

(3) Mobile Equipment

In order to minimize the use of contractor operated mobile equipment in conjunction with work performed by employees, the Company will revise the method of contracting for such equipment by introducing in October, 1991 a new system whereby, except in cases of emergency, contracts for operated mobile equipment will be arranged for by the Transportation Department.

(4) Annual Review

On or before December 15th of each year, the Company shall meet with the Committee and review all work or types of work which it anticipates will be performed by contractors in the next twelve (12) months.

The Company will consider any comments or suggestions proposed at such

meeting to have any contract segment of the work performed by bargaining unit employees, including specific types of work, taking into consideration the trend analyses developed during the previous year.

The Union may appoint a full time co-ordinator to work in connection with the Contracting Out Information and Review System, including the development of trend analyses. The Company will pay the co-ordinator at his applicable hourly rate plus applicable Cost of Living Allowance and **any applicable Schedule 'N' Bonus** for up to forty (40) hours in each week, other than vacation weeks.

The Company will appoint, on a trial basis, a Management Co-ordinator who will work with the Union Contracting Out Co-ordinator in connection with the Contracting Out Information and Review System as described in this Schedule 'U', including the development of trend analyses of the Small Contracts to define Base Load in the type of work normally performed by bargaining unit employees.

The Management Co-ordinator will report to the Vice-President of Production and be a member of the Division Contracting Out Committee and Chair of the Contracting Out Cost Reduction Committee. He will be responsible for preparing and distributing Division statistics and reports regarding contracting out activities. The Management Co-ordinator will also facilitate and promote awareness of matters concerning contracting out, including other available options and best practices, in an effort to reduce contracting out.

In the event a production shutdown is announced, the Management Contracting Out Co-ordinator will chair a special meeting with the appropriate managers, the Union Contracting Out Co-ordinator and appropriate Local Union executives. The purpose of this meeting will be to discuss work that could be performed by unit employees during such period.

EMPLOYEE BENEFIT PLANS

The following descriptions of Employee Benefit plans are presented here for the convenience of employees. They describe the principal features in non-technical language. The insurance contracts between the insurance carrier and the Company will govern in all cases.

GROUP TERM LIFE INSURANCE PLAN

This plan provides for -

Amount of	Amount of	Total Amount
Basic Insurance	Additional	of Insurance
	Insurance	
\$2,500		\$55,000
	<i>\$52,500</i>	

The additional **\$52,500** of Life Insurance will automatically terminate on retirement except as described below for employees who become totally and permanently disabled and retire on disability retirement with 3 or more years of continuous service.

Life Insurance

Benefits will be paid in a lump sum or in installments to the beneficiary designated by the employee. The beneficiary may be changed at any time in accordance with the applicable provincial laws.

Permanent Total Disability

If the employee with 3 or more years of continuous service becomes totally and permanently disabled while insured, \$5,000 of his Life Insurance will be paid to him either in a lump sum or in installments instead of to his beneficiary at death. The remaining portion of life insurance coverage will be cancelled. The entire and irrevocable loss of sight of both eyes, or of the use of both hands or both feet, or of one hand and one foot will in itself be considered permanent and total disability.

Eligibility and Effective Date of Insurance

All employees are eligible for insurance upon completion of six months of continuous service. Insurance coverage will automatically become effective, without medical examination, upon the completion of six months' continuous service provided the employee is then actively at work.

Premiums

The Company will pay one hundred percent (100%) of the premiums payable under this Plan.

Benefits (Employees Out Sick) -

Employees off for less than two years on account of disability resulting from accident or bodily disease will be considered as continuing in the service of the Company in order that their Group Life Insurance may be continued during their unavoidable absence from this cause.

If an employee is off on account of accident or bodily disease for more than two years his insurance may be continued only upon written application to and approval by the Company. The complete terms of this Group Insurance coverage are set forth in the Group Insurance Policy issued by the Carrier.

Termination of Employment and Conversion of Life Insurance

The Life Insurance of an employee will cease upon termination of employment except with respect to the life insurance continued for retired employees as described elsewhere. However, if death should occur within thirty-one days after termination of employment the full death benefit will be payable. Within this thirty-one day period an employee may convert his Group Life Insurance coverage to an individual Whole Life or Endowment Plan, by making application to *the Carrier*. This individual policy will be issued without medical examination and at the Insurance Company's regular rates.

GROUP NON-OCCUPATIONAL ACCIDENTAL DEATH AND DISMEMBERMENT INSURANCE PLAN

This Plan provides benefits in the event of an employee's loss of life, limbs or the entire and irrevocable loss of sight, excluding such losses resulting from occupational accidents. Benefits are payable only if the loss results directly from bodily injuries caused by an accident and occurs within 90 days after date of the accident causing the loss.

Benefits

\$40,000 will be paid for the loss of life, both hands, both feet, one hand and one foot, one hand and the sight of one eye, one foot and the sight of one eye, sight of both eyes, **\$20,000** will be paid for the loss of one hand, one foot, the sight of one eye. In no case will more than **\$40,000** be paid for all losses resulting from one accident.

Exclusions

No benefits are payable due to a loss contributed to or caused by bodily or mental infirmity, ptomaines, bacterial infections, disease, medical or surgical treatment not made necessary by injuries covered under the Plan, war, suicide, or the commission by the insured or any attempt by the insured to commit an assault or an offence under the Criminal Code of Canada.

Eligibility

All employees will be automatically covered by the Group Non-Occupational Accidental Death and Dismemberment Insurance Plan on the date of hire.

Premiums

The Company will pay one hundred per cent (100%) of the premiums payable under this Plan.

The complete terms of this Group Insurance coverage are set forth in the Group Insurance Policy issued by the Carrier

Termination of Employment

All coverage under this Plan will cease on the date of termination of employment, including Retirement.

PRESCRIPTION DRUG PLAN

(35¢ Deductible) (Formulary - 2)

Benefits

The Group Plan for Prescription Drugs (35¢ deductible) (Formulary - 2) covers drugs, serum, injectibles, insulin and diabetic supplies purchased on the prescription of a medical doctor for the use of the employee or a dependent under the Plan.

The Plan does not cover patent medicines, vitamins, unless injected, or drugs paid for by any other agency.

All eligible drug bills will be paid under this Plan, less a deductible of 35¢ for each prescription which will be paid by the employee.

The Plan provides full benefits for the subscriber, wife or husband and all unmarried, unemployed children up to the age of 21 or up to 23 if an unmarried full time student. Newborn children are covered automatically by the Plan.

Eligibility

Employees are eligible on the first day of the month following the month in which the employee was hired.

Premiums

The Company will pay one hundred per cent (100%) of the premiums payable under this Plan.

Certificates

Each employee will receive a personal Certificate showing his name, group number, amount of deductible and the date on which coverage begins.

Cancellation

Insurance will be cancelled on the date employment is terminated.

NON-OCCUPATIONAL SICKNESS AND ACCIDENT INSURANCE PLAN

A group insurance plan which provides weekly sickness and accident insurance.

Benefits

Effective June 1, 2006-Weekly Benefit - \$625

Eligibility

You will automatically become insured on the day after you complete three months of continuous service without medical examination, if you are then actively at work; otherwise, on the day you return to active work.

Premiums

The Company will pay one hundred per cent (100%) of the premiums payable under this Plan.

Benefits Paid

Benefits commence with the first day you are either totally disabled by a non-occupational accident, admitted to and confined to a hospital, or on the fourth day of total disability due to non-occupational sickness. You must be under the care of a physician licensed to practice medicine.

Benefits will continue as stated above during any one continuous period of disability due to one or more causes. Successive periods of disability will be considered as one continuous period of disability unless the disability is due to a different and unrelated cause and commences after you have returned to work.

Benefit Duration

The Plan will provide up to 52 weeks of benefits while disabled for employees with less than one year of service prior to the start of the disability and up to 104 weeks for employees with one or more years of service prior to the start of the disability.

Cancellation

Insurance will be cancelled on the date employment is terminated or when you retire.

DENTAL PLAN

Benefits

The Plan provides for basic dental care and covers examinations, consultations, x-rays, scaling of teeth, fillings, surgical removal of teeth and endodontics, periodontics and orthodontics (50% of fees, maximum life time benefit of \$1,500). The Plan does not cover treatment paid through any other source.

The Plan provides covered benefits in accordance with the **2009** Ontario Dental Association Schedule of Fees for the subscriber, wife or husband and all unmarried, unemployed children up to the age of 21 or up to 23 if an unmarried full time student. Effective **January 1**, **2011**, the applicable schedule will be the **2010** Ontario Dental Association Schedule of Fees, effective **January 1**, **2012**, the applicable schedule will be the **2011** Ontario Dental Association Schedule of Fees, effective **January 1**, **2013**, the applicable schedule will be the **2012** Ontario Dental Association Schedule of Fees, effective **January 1**, **2014**, the applicable schedule will be the **2013** Ontario Dental Association Schedule of Fees. Effective **January 1**, **2015**, **through to May 31**, **2015**, the applicable fee schedule will be the **2014** Ontario Dental Association Schedule of Fees, for Dental Services provided by General Practitioners.

Eligibility

Employees are eligible for coverage on the first day of the month following the month in which the employee has completed three months of continuous service.

Premiums

The Company will pay one hundred per cent (100%) of the premiums payable under this plan.

Certificates

You will receive a personal Certificate showing your name, group number and the date on which the coverage begins.

Cancellation

Insurance will be cancelled on the date employment is terminated.

GROUP OPTICAL INSURANCE PLAN

Benefits

The Plan provides for payment of up to \$250 for prescription glasses in each three year period for each employee, the spouse of the employee and all unmarried, unemployed children up to the age of 21 or up to 23 if an unmarried full time student. The plan will also provide unlimited lens replacement with prescription

change over the three year period.

Eligibility

Employees are eligible on the first day of the month following the month in which the employees was hired.

Premiums

The Company will pay one hundred per cent (100%) of the premiums payable under this Plan.

Cancellation

Insurance will be cancelled on the date employment is terminated.

LETTERS OF AGREEMENT

The following are the letters of Agreement between the Company and the Union, and are intended for the purpose of information. They do not form part of the Collective Bargaining Agreement.

CONTRACTING OUT LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Mining Skills & Contracting Out:

Dear Mr. Fera:

Given current economic circumstances and a prolonged hiring freeze, the Mines Training Centre will be suspended until further notice. The Mines Training Centre may be re-opened provided economic conditions have changed and a business case supports its re-opening.

However, the All Mines Development Department will remain active with at least 32 employees if during the life of this agreement contractors are performing lateral mine development in any of the active mines.

Mileage will be paid to the All Mines Development Department employees in accordance with the letter of agreement dealing with travel compensation for the construction group employees.

In all cases, the All Mines Development Department will be expected to meet and sustain appropriate performance standards.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Advance Notification of Small Contracts

Dear Mr. Fera:

This will confirm our understanding with respect to advance notification of Small Contracts.

The Company will, except in cases of emergency, endeavour to provide the Local Union with notice of Small Contracts in advance of the commencement of the contract work.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

All Mines Construction Group

Dear Mr. Fera:

This will confirm our understanding with respect to a new All Mines Construction Group.

In recognition of the intended positive impact on the reduction of contracting out and business results, the Company agrees to establish and maintain during the term of this Agreement an all mines construction group of at least twelve (12) employees to perform such work as is required in the Sudbury District. This work will include, but is not limited to, chutes, grizzlies, bins, and related work.

The work performed by this group will primarily target opportunities that can reduce or eliminate normally scheduled work performed by contractors in the Sudbury District. These opportunities will also reduce costs, improve quality and reduce rework. On a periodic basis this initiative will be evaluated for business competitiveness and potential workgroup expansion.

With this in mind, we wish to confirm our agreement to the following:

- 1. For the purpose of Article 17 and Schedule 'J' of the Collective Bargaining Agreement, All Mines Construction Group employees working in plants will use the assigned Safety Representatives of the plant.
- 2. In order to provide for addressing All Mines Construction Group employees' concerns and complaints, Stage 1 grievances may be submitted in the employee's work group or in the plant that the employee is working. If not resolved at the plant, these grievances will be forwarded to and handled by management where the grievance originated representatives at any further stages of the grievance procedure.

- 3. The Letter of Agreement in conjunction with the travel compensation for Construction Group Employees will apply to these employees.
- 4. The Local Union and the Company will review training arrangements associated with the assignment of new employees to this group.
- 5. The Contracting Out Steering Committee will meet to review issues relating to the All Mines Construction Group, which may arise during the life of the collective agreement.

Planned workforce levels for the All Mines Construction Group will be reviewed at the regularly scheduled meetings of the Contracting Out Steering Committee.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Contracting Out Committee

Dear Mr. Fera:

This will confirm our understanding with respect to various contracting out matters and the role of the Contracting Out Committee.

- The Contracting Out Committee will meet to review and develop recommendations in connection with the use of overtime in the Maintenance and Electrical Departments in order to reduce the amount of contracting out in those Departments.
- Any alleged use of company unfixed equipment or machinery by a contractor will be referred to the Contracting Out Committee for consideration and recommended resolution.
- 3. Should there be any significant change in the scope of a contractor's contract for a Mid-size Project, the change will be brought to the attention of and reviewed by the Contracting Out Committee.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Contracting Out Steering Committee

Dear Mr. Fera:

Vale and the United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers) are committed to reducing the current amount of contracting out in the Ontario Division and, as a result, improving the business competitiveness of the Division.

This will confirm that Inco agrees to launch this initiative at Mines and Plants to reduce contracting out and to substantially cut ongoing costs by the reduction of contracting out, with the following objectives:

- 1. To reduce significantly the current amount of mining development, production and support work performed by contractors;
- 2. To achieve and maintain an overall reduction in the current level of contracting out of work normally performed by bargaining unit employees in other areas of 40% during the term of the 2003 Collective Bargaining Agreement.

The Company and the Union agree to establish a Contracting Out Steering Committee consisting of the Local Union President or his designate, the Local Union Contracting Out Co-ordinator, a Union representative, the Ontario Division Vice-President, the Company Contracting Out Co-ordinator and the Manager of Human Resources.

We all recognize that the scope of and the issues involved in contracting out vary from area to area and agree that the jointly developed Contracting Out Policy and Procedure Handbook will guide us in our task. Each individual area will analyse its contracting out activities as it relates to work normally performed by bargaining unit employees and develop with the support of the Steering Committee a contracting out/cost reduction plan by October 1, 2000, bearing in mind the purposes, objectives and targets of this initiative. It is understood that

work that is part of a major capital project (\$1 million or more) may not be covered by this letter.

The Steering Committee will meet as frequently as required and its mandate will include the following:

- (a) the co-ordination and support of the efforts of the Mines and Plants developing their plans;
- (b) the review of the area plans and recommending changes to the plans consistent with achieving the purposes, objectives and targets of this initiative;
- (c) the monitoring of the performance of the Mines and Plants in implementing their contracting out/cost reduction plans during the term of this Agreement; and
- (d) communicating the progress of and results from this initiative to all employees of the Ontario Division on a quarterly basis.

If for any reason, the purposes, objectives and targets of this initiative are not realized and/or maintained, the Steering Committee will investigate the reasons for this and facilitate solutions. Only the Steering Committee can deal with issues in connection with this letter.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202
Sudbury, Ontario

Construction Group

Dear Mr. Fera:

The Company will re-establish the Construction Group within the Power and Construction Plant as per the CBA.

During the course of negotiations, the Union and the Company reviewed the role of the Construction Group and the positive impact on the reduction of contracting out and business results. In recognition of this, the company will hire new employees to increase the Construction Group to 15 employees after replacement hiring for employees retiring in June to August of 2006.

This workgroup of 15 will be maintained for the life of the **2010** Collective Bargaining Agreement and this initiative will be evaluated for business competitiveness on a periodic basis.

The Construction Group is expected to target opportunities that can reduce or eliminate regularly scheduled contracted-out work.

With this in mind, we wish to confirm our agreement to the following:

- 1. For the purpose of Article 17 and Schedule 'J' of the Collective Bargaining Agreement, Construction Group employees working in plants will use the assigned Safety Representatives of the plant.
- 2. In order to provide for addressing Construction Group employees' concerns and complaints, Stage 1 grievances may be submitted in the employee's work group or in the plant that the employee is working. If not resolved at the plant, these grievances will be forwarded to and handled by the Power and Construction Department management representatives at any further stages of the grievance procedure.
- 3. The Letter of Agreement in conjunction with the travel compensation for Construction Group Employees will apply to these employees.

- 4. The Local Union and the Company will review training arrangements associated with the assignment of new employees to this group.
- 5. The Contracting Out Steering Committee will meet to review issues relating to the Construction Group, which may arise during the life of the collective agreement.
- 6. Planned workforce levels for the Construction Group will be reviewed at the regularly scheduled meetings of the Contracting Out Steering Committee.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Central Field Services Workgroup

Dear Mr. Fera:

The Company will expand the current Smelter Field Services workgroup from 12 to at least 18 employees and change the name of this group to the Central Field Services Workgroup. The work performed by this group will primarily target opportunities that can reduce or eliminate normally scheduled maintenance performed by contractors on surface in the Sudbury District. These opportunities will also reduce costs, improve quality and reduce rework. On a periodic basis this initiative will be evaluated for business competitiveness and potential workgroup expansion.

This work group of at least 18 employees will be maintained for the life of the 2010 Collective Bargaining Agreement.

With this in mind, we wish to confirm our agreement to the following:

- 1. For the purpose of Article 17 and Schedule 'J' of the Collective Bargaining Agreement, Central Field Services Workgroup employees working in plants will use the assigned Safety Representatives of the plant.
- 2. In order to provide for addressing Central Field Services Workgroup employees' concerns and complaints, Stage 1 grievances may be submitted in the employee's work group or in the plant that the employee is working. If not resolved at the plant, these grievances will be forwarded to and handled by the Smelter management representatives at any further stages of the grievance procedure.
- 3. The Letter of Agreement in conjunction with the travel compensation for Construction Group Employees will apply to these employees.
- 4. The Local Union and the Company will review training arrangements associated with the assignment of new employees to this group.

- 5. The Contracting Out Steering Committee will meet to review issues relating to the Central Field Services Workgroup, which may arise during the life of the collective agreement.
- 6. Planned workforce levels for the Central Field Services Workgroup will be reviewed at the regularly scheduled meetings of the Contracting Out Steering Committee.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Contractors and Mobile Equipment

Dear Mr. Fera:

There are significant savings to be achieved when Vale can use its purchasing power to get a lower price on mobile equipment and avoid the additional cost of dismantling and removing equipment.

This will confirm our understanding that notwithstanding Section 2.04 (d) in the collective agreement, the Company can purchase mobile equipment from a third party for use underground by a contractor performing work on a specific major project as defined in Schedule 'U'. Upon completion of the project, the purchased equipment may be retained by the Company for use by its employees and, thereafter, section 2.04 (d) shall apply to the equipment.

Where the Company intends to purchase equipment for use by a contractor pursuant to this letter, it will advise the Union in advance during a Schedule 'U' meeting and inform them of the type of equipment being purchased.

All provisions of Schedule 'U' will apply to this letter as may be amended from time to time, in accordance with the "Contracting Out Initiative Review" letter of agreement previously agreed to by the Parties.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Contracting Out Initiative Review

Dear Mr. Fera:

During the life of this agreement the parties agree to jointly review our Contracting Out initiative for the Mines and Plants to further reduce contracting out and substantially improve business results. There are at least three parts of the current collective agreement that would be updated as a result of the review:

- Schedule 'U'
- Contracting Out Committee letter, page 108 CBA
- Contracting Out Steering Committee letter, page 109 CBA

Areas of the joint review will include the following:

- 1. Targets to be achieved
- 2. Metrics that measure performance against targets
- 3. System to collect and monitor metrics
- 4. Performance Reporting Mechanisms
- 5. Contracting Out Plant Committee purpose and accountabilities
- 6. Contracting Out Steering Team purpose and accountabilities
- 7. A review of the standard agenda for quarterly meetings that includes presentations from all contracting out initiatives

This review will be done by the current Contracting Out Steering Committee.

Yours truly,

DISABILITY MANAGEMENT LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers) 128 Pine Street, Suite 202 Sudbury, Ontario

Pension (YEWPs & Entitlement to Non-Occupational S&A Payments)

Dear Mr. Fera:

By way of clarification of the Collective Agreement effective July 8, 2010, between Vale Canada Limited and United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers) covering employees of the Company in the Sudbury District and of the Pension Plan for Hourly Paid Employees in the Ontario Division of the Company represented by United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), the following is agreed to by all parties.

A person who makes an application which is accepted by the Company Medical Examiner for any pension under the Pension Plan or a person who is retired under the terms of the Pension Plan ceases to be an employee and he shall not be entitled to continue to accumulate Years of Employment With Pay (YEWPs).

If such a person is in receipt of a benefit payable under the Group Non-Occupational Sickness and Accident Insurance Plan at the time he makes an accepted application or is retired by the Company, he shall receive the monthly pension benefit to which he is entitled under the Pension Plan. However, if the monthly amount of that pension benefit is less than the monthly amount he would have received if he had continued to be an employee and had received his Sickness and Accident benefit under the Collective Agreement, he shall be entitled to an additional amount equal to the difference between his pension benefit and the Sickness and Accident benefit he would otherwise have received for that month. This additional monthly amount shall continue until he has exhausted all of his Sickness and Accident benefits in the same manner as if he was still an employee.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202
Sudbury, Ontario

Increases in Non-Occupational S&A Payments

Dear Mr. Fera:

We confirm the advice given to you during negotiations that with respect to benefits to which an employee may be entitled under the Group Non-Occupational Sickness and Accident Insurance Plan, including the Long Term Disability Protection Plan outlined in Section 24.07, any increase which may be provided will take effect as of the date of the increase even though an employee may have been receiving a lower rate at the commencement of his disability.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Placement Committee

Dear Mr. Fera:

This will confirm our understanding in connection with permanent partially disabled employees. In order to assist in the placement of such employees, a Committee will be formed consisting of three (3) employees appointed by the Local Union and three (3) persons appointed by the Company.

This Committee will meet as required. It will be notified of and will review all placements of permanent partially disabled employees under Section 11.23 of the Collective Bargaining Agreement and may make recommendations to the Company regarding new and alternate placements. Each placement will be reviewed by the Committee one (1) year after it is made. Finally, the Committee will advise in connection with the placement of those employees in training and rehabilitation programs sponsored by the Workers' Compensation Board.

The Company will pay one of the employee members of the Committee eight (8) hours per week at his applicable hourly rate plus any applicable shift premium and any applicable Cost of Living Allowance and **any applicable Schedule 'N' Bonus** for attending meetings of the Committee, the Occupational **Health and** Medicine Department Mini-Tribunal, the Vocational Rehabilitation Committee, and for assisting in the administration of the programs and procedures applicable to Section 11.23 of the Collective Bargaining Agreement.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Absent Employees Subject to Layoff

Dear Mr. Fera:

This will confirm our understanding with respect to Section 11.02 of the Collective Bargaining Agreement. Employees who are absent from work for any reason, including leave of absence and illness or injury (whether in receipt of any kind of benefits from the Company or the government) are subject to layoff in accordance with their seniority ranking pursuant to Section 11.02 of the Collective Agreement in the same manner as any employee at work.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Disability Retirement Application for Individuals with Recall Rights

Dear Mr. Fera:

This will confirm our understanding with respect to the entitlement of individuals on layoff with recall rights to apply for a Disability Retirement under the Pension Plan for Hourly Paid Employees in the Ontario Division of the Company.

In order to be eligible for a Disability Retirement, the individual must be totally and permanently disabled within the meaning of the Pension Plan and his disability must have resulted from injury or disease suffered or contracted by him while in the employment of the Company and not while on layoff.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Recovered Disability Pensioners Eligible to Apply for Full Time Employment

Dear Mr. Fera:

This will confirm our understanding that if an individual in receipt of a Company disability pension ceases to be totally and permanently disabled within the meaning of Section III (f) of the *Ontario* Pension Plan for Hourly Paid Employees of *Vale Canada Limited*, Ontario Division, he may make application to return to active full time employment with the Company.

The individual will be re-employed subject to manpower requirements and his ability to meet the Company's medical standards. He will not be given preference over individuals with recall rights.

Upon rehire, the individual will be credited with the continuous service he had prior to being placed on disability pension and all disability benefits provided under the terms of the *Ontario* Pension Plan will be forfeited. Should the individual requalify for a disability pension at some later date, he will not be eligible to receive an additional \$5,000 benefit payable under the Group Term Life Insurance Plan and will have the \$5,000 previously paid deducted from his Life Insurance protection under Section 24.05 of the Collective Bargaining Agreement.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Vocational Rehabilitation Committee

Dear Mr. Fera:

This will confirm our understanding with respect to the creation of the Vocational Rehabilitation Committee to assist in the placement of permanently partially disabled employees (PPD's) into value-added work and to the initial placement into the Rehabilitation Group.

The Committee will consist of a representative of Local 6500, a member of the Occupational Health **and Medicine** Department and a person from the Claims Administration Group.

The Committee will meet for the purpose of making recommendations to the Company regarding the placing of PPD's into value-added vacant jobs using techniques such as functional abilities evaluations and physical demands analyses.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Rehabilitation Group

Dear Mr. Fera:

This will confirm our agreement with respect to employees who are either temporarily or permanently unable to perform their regular work due to age, disease or injury (PPD's).

The company will establish the Rehabilitation Group, consisting of the Safety and Miscellaneous Department, the Modified Work Centre and the Reconditioning Shop. This Group will provide temporary rehabilitation and work hardening employment to PPD's for the purpose of returning them to value-added work. Effective June 1, 1997, all PPD's in the Safety and Miscellaneous Department, the Modified Work Center and the Reconditioning Shop will be transferred to the rehabilitation group.

It is agreed that all transfers to the Rehabilitation Group are temporary. Any PPD may, on the basis of his ability, knowledge, training, skill, medical condition and seniority within the Group, be assigned to a vacant position in any Department, notwithstanding any other provision of the Collective Bargaining Agreement. However, once the PPD is so assigned, he must compete for positions in the Departments on the same basis as any other employee. If the PPD is subsequently transferred to the Rehabilitation Group, he may again be so assigned to a vacant position in any Department.

All assignments from the Rehabilitation Group to a Department will be made with the assistance of the Vocational Rehabilitation Committee.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

No Fault Auto

Dear Mr. Fera

This will confirm our understanding with respect to employees who are off work and in receipt of income replacement benefits under the statutory accident benefits provisions of the Insurance Act.

Such employees will continue to accumulate continuous service and Years of Employment With Pay (YEWPs) as per the Pension Plan. Any such credited leaves will not exceed those outlined in **Section** 11.24 of the Collective Bargaining Agreement.

Yours truly,

DISPUTE RESOLUTION LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Grievance Commissioners

Dear Mr. Fera:

The parties agree that the Grievance Commissioners under the **2010** Collective Agreement will be:

Professor C.G. Simmons Professor W.B. Rayner

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Grievance Process

Dear Mr. Fera:

During the course of negotiations for a new collective agreement the parties discussed a number of concerns regarding the grievance process and agree to the following:

- 1. The Company and the Union will establish a Joint Grievance Committee consisting of the Vice President of the Local Union, the Chief Stewards, the Manager Human Resources and two (2) persons appointed by the Company.
- 2. The Joint Grievance Committee will meet once every three months, if necessary, to establish appropriate measures and definitions for the determination of backlogs and share information which will promote the effective hearing of grievances. Meetings will be scheduled in plant locations where appropriate, to discuss grievance backlogs and ways to resolve grievance process issues.
- 3. Where it is agreed that there is a backlog of sixty (60) or more Stage Two grievances in a Plant, the parties will, as soon as practicable, but no later than twenty one (21) days, schedule an accelerated round of three (3) Stage Two grievance meetings per week until this backlog is heard.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Education Regarding the Grievance Process

Dear Mr. Fera:

The Company and the Union recognize that a large number of grievances were filed during the life of the previous CBA and that improving the effectiveness of the process is desired. A fundamental cornerstone to improving effectiveness is ensuring that all parties are fully educated and understand the process.

The parties will establish a six (6) person Dispute Resolution Education Committee consisting of the three Chief Stewards and three persons appointed by the Company to develop and have delivered a dispute resolution education/training package for the Supervisors and the Union Stewards in each plant by December 31, 2011. This education/training package will include the following:

- (i) An overview of the CBA
- (ii) A review of the grievance process
- (iii) A review of the primary sources of grievances

In addition, as a part of our commitment to assess the use of web-enabled technologies to improve overtime canvassing (as referenced in the "Distribution of Overtime" Letter of Agreement) the Company and Local Union will further consider ways in which web-enabled technologies may also be used to improve the administration of grievances.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Mediated Arbitration

Dear Mr. Fera:

This will confirm our agreement to introduce the following new process to try and resolve discharge grievances.

Mediation Process For Discharge Grievances

If a discharge grievance is referred to arbitration the party referring the grievance may, within fifteen (15) days of the referral, select an arbitrator from the panel to act as a special mediator.

The special mediator will meet with the parties and the grievor within thirty (30) days and make an effort to resolve the grievance. If the special mediator is unable to meet in thirty (30) days, the referring party shall select another arbitrator from the panel.

The special mediator will, in attempting to resolve the grievance, hear representations from the parties without formal evidence (I.e.: no reference to jurisprudence or use of sworn witnesses) being called and will provide them with his/her non-binding recommendation at the end of the day. The representations made on behalf of the parties will be without use of legal counsel.

The representations made by the parties and the recommended settlement of the special mediator will be without prejudice, or precedent. However, the parties may jointly request that the special mediator issue any settlement arrived at as an arbitrator's decision.

Failing resolution, the referring party may forward the grievance to arbitration and the parties may agree to have the special mediator sit as the arbitrator or select a new arbitrator from the panel.

Yours truly,

SAFETY SYSTEM LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202
Sudbury, Ontario

Joint Occupational Health Study Program

Dear Mr. Fera:

This will confirm the agreement between the Union and the Company to establish a Joint Occupational Health Study Program for both the Sudbury and Port Colborne operations of the Company. This program will be for the purpose of developing a better understanding of the working environment at these locations through independent industrial health surveys and research in connection with potential occupational illness and disease.

To this end a Joint Occupational Health Committee (hereinafter called the "Committee") will be created consisting of eleven (11) members, as follows:

For the Union

The President of Local 6500 or Designate
The President of Local 6200 or Designate
Local 6200 Worker Safety Representative
The *Chairperson* of the General Health, Safety and
Environment Committee (Sudbury)
A senior staff representative of the Union
A representative of Local 2020-005

For the Company

Regional Medical Director: Ontario
General Manager, Sustainability
Director of Health and Safety
Manager, Occupational Health and Medicine
Superintendent, Occupational Health

Occupational health research study groups (hereinafter called "Study Groups") may be established at Canadian Universities or Canadian based organizations for the purpose of undertaking such occupational health studies as are referred to it by the unanimous agreement of the Joint Occupational Health Committee. The university or other organization shall be selected by the unanimous agreement of all members of the Committee. The Study Groups shall consist of a research lead and such support staff and experts as deemed necessary by the research lead, for the purpose of carrying out the functions of the Study Group.

The Joint Committee may by unanimous agreement assign a Committee member or a non-Committee member to perform work that is required by the Committee for the purpose of aiding any study group.

The Committee shall have the following functions:

- 1. To refer occupational health matters to Study *Groups led by appropriate* experts in the field.
- 2. To make recommendations to the Company regarding the implementation and/or dissemination of the findings of the Study *Groups*.
- 3. To recommend methods of obtaining such statistical data and information as is necessary for the Study *Groups* to perform *their* functions.
- 4. To recommend procedures for identification and detection of potentially hazardous or toxic materials.
- 5. To review major occupational health matters and concerns arising out of the operations at Sudbury or Port Colborne.

The functions of the Study *Groups* shall be primarily to conduct studies of occupational health matters referred to it by *unanimous agreement of* the Committee and to formulate recommendations based upon its research. *The Study Groups will be required to report annually to the Committee, summarizing the work completed during the previous year and the work in progress at the time of the report.* Upon its completion of any major occupational health study, the Study Group shall provide the Committee with a written report outlining the nature of the research undertaken and the results.

The Committee shall meet semi-annually under the Joint *Chairs* of one member appointed by the Union and one member appointed by the Company, or more frequently if agreed to between the Joint *Chairs*.

In order to assist the Study *Groups* in the performance of *their* duties, the Company shall, to the extent its legal and contractual obligations permit, make

available to the Study Group for its sole and exclusive use, such statistical information in the possession of the Company as is needed by the Study Group. Any information so supplied shall be treated as private and confidential and is not to be released to any person or party including the Committee, without the prior written approval of the Company.

The Study *Groups* shall have the right to make independent studies on the premises of the Company at Sudbury and Port Colborne with the understanding that any such studies shall be conducted whenever possible without interfering with the requirements of operations.

Reports, recommendations and/or findings of the Study *Groups* shall be made on a confidential basis to the Company and to the Committee and shall not be released to anyone else without the prior written approval of the Company and the Committee.

The Company shall pay all necessary expenses of the Study **Groups** to a maximum of \$750,000 during the term of this Agreement, July8, 2010 to May 31, 2015. Unused amounts are not carried over into subsequent years.

The expenses of any member of the Committee shall be paid by the party appointing such member.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Training in Use of Monitoring Equipment

Dear Mr. Fera:

This will confirm our understanding with respect to monitoring the working environment and in connection with published material pertaining to hazardous substances.

As soon as is reasonably possible, each Operation Health, Safety and Environment Committee shall discuss at its meetings appropriate arrangements for the training of up to five (5) Local Union appointed members and five (5) alternate members of each such committee in the use of basic sampling, monitoring, and detection meters, in use by the company at that time, for use by such members on their regular monthly inspection tours provided for under the Collective Bargaining Agreement.

As soon as is reasonably possible, each Operation Health, Safety and Environment Committee shall discuss at its meetings the times at which and the places where dust samples should be taken at the locations within its Area where dust sampling is currently carried out.

In addition, each Committee shall consider the methods by which published material pertaining to potentially hazardous substances used in the Area can be obtained for the information of the Committee.

As soon as reasonably possible following the request of any Operation Committee, representatives of the Company will attend at a regular meeting of that Committee to review and explain the function of all monitoring equipment normally used by the Company in that Operation.

The Company will provide each Operation Committee with the results of the Occupational Exposure Monitoring Program pertaining to the work performance in that Operation.

Up to two (2) Committee members designated by the Union shall be given basic ergonomic training, to enable them to identify and address basic ergonomic issues. The Committee Co-*Chairs* will agree to the names of the members to be trained and the timing.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Prescription Safety Glasses

Dear Mr. Fera:

If it is established that an employee while doing assigned work and exercising due caution sustains damage to his prescription safety glasses the Company will reimburse the employee for the cost of necessary repairs or replacements.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Environmental Awareness Committee

Dear Messrs Fera & Sullivan:

The Company and the Union agree to establish a special senior level committee in connection with environmental matters relating to the operations of the Ontario Division.

This committee shall be known as the Environmental Awareness Committee and will consist of the following members:

For the Union

- the President of Local 6500
- the President of Local 6200
- the *Chairperson* of the General Health, Safety & Environment Committee
- the Director of District 6 of the Union

For the Company

- Three Senior Management representatives
 One of whom shall be a Vice President or General Manager
- The Director of Environmental and Health Sciences

The Committee will be co-chaired by the Director of District 6 of the Union and the **Director of Environmental and Health Sciences of Vale Canada Limited.**

The Committee shall focus on the Natural Environment (the air, land and water bordering real property in which Vale owns surface or mineral rights, while recognizing that our properties must be reclaimed to a natural state when mining and smelting activities cease) and have the following functions:

1. To examine the environmental laws and regulations pertaining to the Ontario Division.

- 2. To review and assess environmental matters that relate to Division operations.
- 3. To study and make recommendations with respect to the application of the Internal Responsibility System to these environmental considerations.
- 4. To develop and make recommendations to the Company relating to employee training for the purpose of increasing awareness of and responsibility for *natural environment* matters and the impact of operations on the external environment.
- 5. To study and make recommendations for enlarging the focus of the Safety, Health and Environment Committees so that these environment issues become part of their agenda.

The Committee shall meet semi-annually in Sudbury under Joint *Chairpersonship* as described above.

Reports and recommendations of the Committee shall be made on a confidential basis to the Company and the Union and shall not be released to anyone else without the prior written approval of both the Company and the Union.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Changes To **OHSE**

Dear Mr. Fera:

The company recognizes the concern of the union that changes in legislation have the potential to impact on practices in Health and Safety. To address those concerns the following is agreed upon:

- A. In the event that sections 41 or 42 of the Occupational Health and Safety Act currently in effect is repealed and not replaced, or replaced in a manner that diminishes the employees' right to know, the following shall come into effect:
 - The company will ensure that information pertaining to hazardous materials is made available for workers who may be exposed to the hazardous physical agent.
 - The company will also ensure that this information is available for the Joint Occupational Health, Safety and Environment Committee.
 - Notices identifying and warning of the potential hazardous physical agent shall be posted.
- B. In the event that section 43 of the Occupational Health and Safety Act currently in effect is repealed and not replaced, or replaced in a manner that diminishes the employees' right to refuse unsafe work, the following shall come into effect:
 - An employee may refuse to perform work where he/she has reasonable grounds to believe, and does believe, that the particular work is dangerous to his/her health or safety, or the health or safety of another employee.
 - Upon refusing to perform that particular work, he/she shall promptly inform his/her immediate supervisor of the refusal and the reasons therefore.
 - The supervisor shall promptly investigate the refusal with a Union member of the Joint Occupational Health, Safety and Environment Committee until the investigation is completed, the worker shall remain in a safe place near his/her work station.

- C. In the event that section 11 of the Occupational Health and Safety Act currently in effect is repealed and not replaced, or replaced in a manner that diminishes the Joint Occupational Health, Safety and Environment Committee's right to participate, the following shall come into effect:
 - The company shall consult with, and provide information to the worker safety representative at the plant with respect to proposed testing strategies for industrial hygiene at the workplace.
 - A worker member of the Joint Occupational Health, Safety and Environment Committee is entitled to be in attendance at the beginning of testing conducted with respect to industrial hygiene, if the worker cochair believes his/her attendance is necessary to ensure valid testing and results.

Yours truly,

TRADES LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Mileage Compensation – Construction Group Employees

Dear Mr. Fera:

This will confirm our understanding in connection with the travel compensation for Construction Group Employees.

Payments to such employee travelling to the job will be made on the following basis:

The distance travelled from his home to his job site less the distance from his home to the Copper Cliff Smelter Parking Lot equals the distance for which payments will be made at the mileage rate established by the Company for Staff Employees.

However, if such employee's job site is within a four (4) kilometre radius of the Copper Cliff Smelter Parking Lot, no payment will be made.

If, during the course of his working day, a Construction Group employee is instructed to and uses his own vehicle to travel from one job site to another, he shall be paid such rate for the distance travelled even if part or all of the distance is within the four (4) kilometre radius.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Trades Committee

Dear Mr. Fera:

We wish to confirm our discussions and understanding with respect to the existing trades classifications listed in Section 20.01 of the Collective Bargaining Agreement.

The need for improved productivity and competitiveness has resulted in changes in operating requirements. The increased manpower flexibility that was required to meet this need is a continuing one and, as a result, it may be necessary to create new trade classifications.

Therefore, the parties agree to continue the Trades Committee consisting of three (3) people appointed by the Company (a Manager of Mines, a Manager of Mills, Smelting and Refining and the Manager of Maintenance and Utilities) and three (3) persons appointed by the Union. The committee will meet at least once every three (3) months, and at such other times as may be agreed upon, for the purpose of exchanging and discussing information pertaining to existing and new trades and considering specific problems relating to the trades.

The Company shall pay employee members of the Committee at their applicable hourly rate plus any applicable Cost of Living Allowance and any applicable **Schedule 'N' Bonus** for any time lost during their regular working hours while attending Committee meetings.

The parties agree that existing *tradespersons* who wish to remain in their present trade classifications may do so, and they will neither be required to take any additional training in connection with any new trade classification nor be required to work in any new trade classification, unless they so elect.

In cases where there is an increase in the job class of a trade classification because of a change in job content, existing *tradespersons* in that trade classification will not be demoted if they cannot successfully complete any

additional required training.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Force Adjusted **Tradespersons**

Dear Mr. Fera:

This will confirm our understanding that an employee previously employed as a *Tradesperson* or Stationary Engineer, subsequently force-adjusted from that job, and currently employed as an apprentice within a trade listed in *Section* 20.09 of the Collective Agreement, will be allowed to bid on any trades or Stationary Engineer's job vacancies in which he was so previously employed.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Apprentice Posting Travel Compensation

Dear Mr. Fera:

This will confirm our understanding with respect to Section 20.10 of the Collective Bargaining Agreement.

If the successful applicant is working in a shift job, he will remain in that job until the apprentice has completed his final training period.

During that time, the successful applicant will receive travel compensation, if the location of the new job is closer to his home than his existing shift job location, in an amount equal to the difference between the distance from his home to his existing job location and his home to the new job location.

All travel compensation payments will be made at the rate per kilometre established by the Company for Staff employees.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Trades Apprenticeship Opportunities

Dear Mr. Fera:

This will confirm our understanding with respect to offering trades apprenticeship opportunities for existing employees.

The Company wishes to provide existing employees with an opportunity to advance their careers by entering into an apprenticeship program.

The Joint Trades Committee will meet to discuss possible new apprenticeship opportunities by November 1, 2010. The number of available apprenticeship opportunities and the trades in which these apprenticeship opportunities will be offered will be determined by the Company on the basis of future workforce requirements. An employee who is selected by the Company to participate will be paid in accordance with Section 20.12 and will be required to attend a recognized Community College or certified apprenticeship provider on a full-time basis as directed by the Ministry of Training, Colleges and Universities.

The Company further commits to convene the Joint Trades Committee on an annual basis to review the trades apprentice requirements for each upcoming year.

Yours truly,

WORKFORCE TRANSITION LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Timing Of Job Posting Procedure

Dear Mr. Fera:

This will confirm our agreement with respect to job posting as provided for pursuant to the provisions of the Collective Bargaining Agreement.

Should the Company introduce electronic job postings, it is understood that the references to "bulletin board or boards" as contained in Sections 11.12 (1), 11.18 and herein below, shall be read as permitting the use of electronic job postings in place of the physical postings on bulletin boards.

If the Company introduces electronic job postings, it will provide education and training to all employees on the new process.

All job postings will be made on Monday of each week (unless Monday is a recognized holiday under the Collective Bargaining Agreement, in which case such postings will be made on the next day, Tuesday) and shall be removed from the bulletin board or boards concerned on the next Monday.

The successful applicants for such postings will be posted no later than Thursday of the week in which such postings are so removed (unless a recognized holiday under the Collective Bargaining Agreement occurs during that week, in which case the successful applicants will be posted no later than the next day, Friday) and the Company will endeavour to assign the successful applicants to their new jobs on the next Monday. In the case of an unlimited job posting in the Maintenance or Electrical Department (other than one made to accommodate a new hire), the successful applicant will be moved to his new job within five (5) weeks of the posting of his name as the successful applicant.

In the case where a successful applicant for a trainee job is not so assigned to his

new job, he will be deemed qualified in the occupational classification of the trainee job for the purpose of any unlimited job posting of such occupational classification in his Department occurring after the date he would have otherwise completed his training.

In order to put this job posting procedure into effect, job posting applicants will be required to fully complete the job posting application form. No such application form will be accepted by the Company as a job posting application unless it is so completed.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202
Sudbury, Ontario

Trial Process for Awarding Unlimited Jobs

Dear Mr. Fera:

This will confirm our agreement to establish, on a trial basis for the term of the Collective Bargaining Agreement, a new process for awarding <u>unlimited</u> <u>job postings</u>. The senior applicant who meets the normal prerequisites for the job and is capable of being successfully trained, within the usual training period, to fulfill the normal requirements of the job shall fill the vacancy.

For this purpose, the following sections of the CBA will be suspended or changed during the trial period:

- 1. Section 11.14 (b) will only apply to limited jobs.
- 2. Sections 11.01 (d), 21.02, 21.03 and 21.04 will be suspended.

Notwithstanding anything to the contrary in the collective agreement, in the case of unlimited postings, should the successful applicant fail:

- a) to successfully complete the training; or
- b) to demonstrate the ability to fulfill the normal requirements of the job;

the employee will be returned to his prior unlimited job which will not have been filled on a permanent basis pending this determination.

The parties will meet regularly to review the operation and results of this new process.

Yours truly, Mitch Medina Manager Human Resources

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Reciprocal Rehire Rights

Dear Mr. Fera:

This will confirm our agreement with respect to the hiring of employees in the Sudbury District and Port Colborne Plant.

An individual whose name appears on the recall list for the Sudbury District shall be given preference, subject to meeting normal employment standards, when it is necessary to hire new employees at the Port Colborne Plant and vice versa.

When hired, the individual will be credited with continuous service accumulated at the hiring location.

Any individual hired either in the Sudbury District or at the Port Colborne Plant after his/her recall rights expired will have his/her anniversary of employment recalculated using his/her prior continuous service in either Port Colborne or the District of Sudbury. The recalculated anniversary date will be applicable only to the determination of regular vacation entitlement under Section 15.01 of the Collective Bargaining Agreement. The provisions of this letter shall be effective June 1, 2003.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Special Short Term Shutdown Protection Plan

Dear Mr. Fera:

This will confirm our understanding with respect to the Special Short Term Shutdown Protection Plan.

An individual who is off work from the shutdown Department or shutdown part of the Department, as the case may be, and in receipt of Group Non-Occupational Sickness and Accident insurance benefits will be covered by the provisions of the Plan when he is medically fit to return to his regular work and is unable to do so because of the special shutdown.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Travel Compensation

Dear Mr. Fera:

This will confirm our understanding that any employee who is temporarily transferred by the Company from one Department to another, or temporarily assigned from one Plant to another within a Multi-Plant Department, or is a successful applicant on a job posting and is not moved within thirty (30) days of the posting of his name as the successful applicant, is entitled to travel compensation on the following basis:

In the case of the temporary transfer or assignment, the distance from his home to his new job site less the distance from his home to his former job site equals the distance for which he will be compensated for each kilometre;

In the case of the job posting, the distance from his home to his former job site less the distance from his home to his new job site equals the distance for which he will be compensated for each kilometre.

All travel compensation payments will be made at the rate per kilometre established by the Company for Staff employees.

Yours truly,

OTHER LETTERS OF AGREEMENT:

July 8, 2010

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202
Sudbury, Ontario

'Will' and 'Shall'

Dear Mr. Fera:

The parties agree that the words 'will' and 'shall' bear the same meaning in the Collective Agreement.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Credit Union Deductions

Dear Mr. Fera:

The Company will continue to distribute deductions made by the Company on behalf of employees for deposit to Credit Unions or Caisse Populaire Branches. Such deductions will be made from the wages of employees who sign the necessary authorization.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Plant Committeeman Facilities

Dear Mr. Fera:

The Company recognizes the need for improving communications between the Union and employees and therefore supports the principle of the Plant Committeeman having in each Plant reasonable access to existing computer and telephone facilities and designated areas for meetings with employees or Grievance Stewards and for the storage of Union files and other papers for the processing of Grievance documents. The Company agrees that existing facilities for Plant Committeemen will be maintained during the life of this Agreement.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Plant Committeeman – Upper and Lower Yard

Dear Mr. Fera:

This will confirm our understanding with respect to Plant Committeemen and Schedule 'H'.

If at one of the three plants in the Lower Yard (i.e. Transportation, Shops or Purchasing and Warehousing) there is no Plant Committeeman reasonably available to perform the Plant Committeeman functions under Article 7 of the Collective Bargaining Agreement and an employee from that Plant requires the assistance of a Plant Committeeman for a Stage Two matter, such employee may use a Plant Committeeman from another Plant in the Lower Yard.

An employee from either Plant at the Smelter may similarly use a Plant Committeeman from the other Plant at the Smelter.

An employee from the Matte Processing Plant may similarly use a Plant Committeeman from the Copper Cliff Mill (including Sample House) and vice versa.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Union Elections or Votes

Dear Mr. Fera:

Upon application by the Union or the Local Union to the *Manager Human Resources*, the Company will meet with the Union or Local Union to agree upon space on the Company's premises in the Sudbury District for the holding of Union or Local Union elections or votes.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Vacations

Dear Mr. Fera:

This will confirm our understanding with respect to various matters in connection with vacations.

Although it is the intention of the Company to endeavour to operate during the period of July - August each year, if it is of the view that conditions do not warrant continuation of regular operations during such period, it may shut down for a period of up to five (5) weeks for vacation purposes subject to retaining such employees as the Company feels is necessary. If a shutdown is scheduled, the appropriate vacation lists will be posted in each Department by April 1st in each year of the current Collective Bargaining Agreement.

If the Company is of the view that conditions warrant continuation of regular operations during such period, it will notify employees and post the appropriate vacation lists in each Department not later than the first day of December of the preceding year.

The prime vacation period in any year shall be thirteen (13) consecutive weeks between June 1st and September 15th.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Lists of Overtime Worked

Dear Mr. Fera:

This will confirm our understanding with respect to information concerning overtime of employees in a working group within the meaning of Section 13.09 of the Collective Bargaining Agreement.

The Company will continue to post monthly, in appropriate locations, lists showing the cumulative amount of overtime hours worked by each employee within such a working group, commencing with the first month of such posting.

In addition, the lists will indicate the number of overtime hours each such employee has been unavailable or refused to work when overtime was offered.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Employee Involvement in Training

Dear Mr. Fera:

This will confirm our agreement that the current practice of involving employees in training will be continued and enlarged.

For this purpose, the Company will introduce surface trainer instructor classifications for operations job classifications training.

Jobs (limited or unlimited) will be posted as required in surface plants for the purpose of classroom and other training needs determined by the company. The provisions of Section 11.14 of the Collective Bargaining Agreement shall be applied to determine the successful applicants.

It is understood, however, that any training required may be performed by either employees or other persons.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Distribution of Overtime

Dear Mr. Fera:

The company recognizes the principle of distributing overtime as evenly as possible under Section 13.09 of the Collective Bargaining Agreement when it is practical to do so. The Union recognizes the difficulty involved in the administration of Section 13.09 as it pertains to call out overtime under Section 13.22 and confirms its commitment to the overtime policy dated March 6, 2001.

In accordance with this policy, and bearing in mind that overtime opportunities are not equalized within any time limit, it is also agreed that neither will an employee be equalized at the time he is no longer part of the working group or his classification changes.

The parties agree to establish an overtime review committee for the duration of this Agreement consisting of the Chief Stewards of the Local Union and two (2) persons appointed by the Company for the purpose of suggesting protocols for managing Emergency Overtime, reviewing ways and means to improve the communication of the Overtime Guidelines and identifying best practices which can be shared with stakeholders. The Committee will also review opportunities to use web-enabled technologies to improve the overtime canvassing reports and system. The Company and the Union will meet every three months at the request of either party.

The parties agree that an employee absent from work on either bereavement or jury duty leave of absence shall be deemed as unavailable, and accordingly not charged, for overtime opportunities including those that may occur on Recognized Holidays.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Orientation of New Employees

Dear Mr. Fera:

This will confirm our understanding with respect to Union participation in the orientation of new employees, excluding students. A member of the Local Union Executive or a staff representative of the Union will have the opportunity to address new employees for up to two hours during the course of their introduction to employment. It is understood that this opportunity will not be used to have new employees sign applications for union membership cards and that employees will be advised that membership in the union is voluntary.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Coleman and Lower Coleman

Dear Mr. Fera:

This will confirm our understanding with respect to Lower Coleman. For the purposes of this Agreement, Coleman and Lower Coleman shall have the same meaning.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Education, Humanities, SOAR & Community Funds

Dear Mr. Fera:

This will confirm our agreement with respect to the Steelworkers Education Fund and the Steelworkers Humanities Fund.

1. Steelworkers Education Fund

The Company will allocate by a deduction arrangement satisfactory to the parties two (2) cents per hour to the Steelworkers Education Fund to be administered by a committee consisting of a representative appointed by the Union and a representative appointed by the Local Union.

2. Steelworkers Humanities Fund

The Company will allocate by a deduction arrangement satisfactory to the parties one (1) cent per hour to the Steelworkers Humanities Fund.

3. SOAR Deduction

The Company will allocate by a deduction arrangement satisfactory to the parties five (5) dollars of regular earnings in one of the pay periods in January of each year to SOAR.

4. Community Project

The Company will allocate by a deduction arrangement satisfactory to the parties one (1) cent per hour to a community project. The current community project will be the Sudbury Regional Hospital Children's Treatment Centre.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Warehouse Travel Crew

Dear Mr. Fera:

This will confirm our understanding with respect to the 'Warehouse Travel Crew' as it relates to Schedule 'P' of the Collective Bargaining Agreement.

The Company will, for the duration of this Collective Bargaining Agreement, consider the 'Warehouse Travel Crew' as an individual Plant for the purpose of administering Schedule 'P' of the Collective Bargaining Agreement.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202
Sudbury, Ontario

Vacation Earnings For S&A and Pregnancy or Parental Leaves

Dear Mr. Fera:

This will confirm our understanding with respect to an employee who has been in receipt of benefits under the Group Non-Occupational Sickness and Accident Insurance Plan or has taken pregnancy or parental leave under Section 11.29 (b) of the Collective Bargaining Agreement in the immediately preceding calendar year.

Such employee will, provided he or she has returned to work, be considered to have had earnings in that preceding calendar year (for the purpose of determining the employee's vacation pay in the following vacation year) calculated on the basis of eight (8) hours per day at the applicable hourly rate of the employee's occupational classification at the time the employee either began receiving benefits or commenced pregnancy or parental leave for each day the employee was in receipt of such benefits or on such leave, provided that the total hours on which the employee's vacation pay is calculated do not exceed 2,080.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Whistle Mine

Dear Mr. Fera:

This will confirm our understanding that should Whistle Mine come into underground production, with ramp access, all development work and production mining will be done by bargaining unit employees. If shaft sinking is involved, bargaining unit employees will do all production mining and development work other than that associated with the shaft or immediately adjacent to the shaft.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Team Leader

Dear Mr. Fera:

This will confirm our policy in relation to support for any Team Leader/ Leader charged with an alleged breach of the Occupational Health and Safety Act. By way of clarification, we are referencing:

- a.) Team Leader positions as identified by the Company in which we have declared the incumbent a 'supervisor' in accordance with Article 25(2) (c) of the Occupational Health and Safety Act.
- b.) Any other Leader classification where the incumbent has not been declared a supervisor by the Company, who through the performance of his/ her duties is 'deemed' to be performing duties as a supervisor by the Ontario Ministry of Labour.

Further to its commitment to operate everywhere in a safe and efficient manner and conform with all laws and regulations, the Company requires Team Leader/ Leader to have a sound working knowledge of all work related laws, regulations and applicable Company practices and to take all steps a reasonable person would take in similar circumstances to comply with these.

The Company will fully support the Team Leader/ Leader in dealing with charges alleging breaches of such legislation, unless disregard for workplace legislation and/or Company policies are involved. Save for exceptional circumstances, to be determined by the Company at its sole discretion, this support is confined to the disposal of the initial charges.

A Team Leader/ Leader who is charged with a work related offense shall report the matter immediately to his/her Manager, who will direct the matter to the President of the Division. The Team Leader/Leader will participate in any related investigation providing full details of dates, times, who was involved, the outcome, witnesses, environmental factors and any other information that may be helpful in responding to these charges. Any reports submitted in the course of the incident investigation shall be addressed to Legal Counsel marked "privileged and confidential."

Subject to approval by the Company, the Team Leader/ Leader may retain legal counsel of his/her choice. The Company will reimburse the Team Leader/ Leader for reasonable legal expense.

The Company will decide in its sole discretion, the extent to which it will pay any fines and whether any disciplinary action will be taken.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Health Care Benefits Protection Program

Dear Mr. Fera:

We agree that the following will be included in the Program:

- 1. A generic rider will be added to the Group Plan for Prescription Drugs that specifies that if a brand medication is prescribed, the plan will pay for the amount equal to the lowest cost generic equivalent unless the physician specifies "no substitution", in which case the Plan will cover the cost of the prescribed drug. The Joint Health Care Committee will develop and implement a "stepped prescribing" approach.
- 2. A trial prescription program will be developed and introduced. This program will identify a list of medications with a potential for side-effects and providing that if any of the medications is prescribed for the first time, the recipient will first receive a one to two week supply, depending on the medication:
- 3. A 100 day supply rule for maintenance drugs prescribed by a physician will be introduced.
- 4. The recall period for preventative examinations and cleaning under the Group Dental Plan will be once every nine months and the plan will be amended appropriately.
- 5. The Joint Health Care Committee will make recommendations to the Company from time to time with respect to the process for positive enrolment for all active employees, retirees and dependents.
- 6. The Committee will also define and implement a process to keep enrolment data current so as to maximize coordination of benefits with other insurance plans by June 1, 2011.

7.	The	Committee	will	develop	measures	to	encourage	increased
utilization of the Preferred Pharmacy Network.								

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)
128 Pine Street, Suite 202
Sudbury, Ontario

Summer Student Category

Dear Mr. Fera:

This will confirm our understanding with respect to compensation for summer students.

Summer students are hired for a fixed term to perform work during the period of April 1 to September 15 in any year and are assigned to a summer student occupational classification which will be paid at Wage Step 1 for all time worked.

Summer students will not be eligible to receive any bonus compensation under Schedule 'N'.

Yours truly,

United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers), and Local 6500, United Steel, Paper and Forestry, Rubber, Manufacturing, Energy, Allied Industrial and Service Workers International Union (United Steelworkers)

128 Pine Street, Suite 202

Sudbury, Ontario

Return to Production Bonus

Dear Mr. Fera:

In order to provide employees with an incentive to focus on an efficient, safe and successful return to production, the Company agrees to make two (2) bonus payments as described below:

- 1. The Company will pay one thousand dollars (\$1,000), less any required deductions, to each employee provided:
 - (a) The bessemer matte milled over a rolling forty-two (42) day period equals or exceeds 28,253 tons (70 % of historic output);
 - (b) The employee was employed throughout the period from the second month after ratification and the date that this amount becomes payable under paragraph (a), above.
- 2. The Company will pay one thousand dollars (\$1,000), less any required deductions, to each employee provided:
 - (a) The bessemer matte milled over a rolling forty-two (42) day period equals or exceeds 38,344 tons (95 % of historic output);
 - (b) The employee was employed throughout the period from the second month after ratification and the date that this amount becomes payable under paragraph (a), above.

The payments provided for herein will be made only if the date on which the targets set in paragraph 1(a) and 2(a) are achieved is no later than six (6) months after the date of ratification.

Yours truly,